# Democracy and Growth in the 21st Century

Yusuke Narita\*

April 28, 2025

#### Abstract

We find that, in the 21st century, democracy has persistent *negative* impacts on growth in GDP and night-time light intensity. This finding emerges from five different instrumental variable strategies that account for potential invalidity in some of the instruments. Our analysis also suggests a key mechanism: In this century, many electoral democracies shift toward populism and protectionism. These political changes weaken trade and investment, collectively dampening economic growth. Democracies also experienced lower growth in subjective life satisfaction among citizens. However, democracy causes slower growth in CO2 emissions and energy use, suggesting a trade-off between economic growth and socio-environmental performance.

*Keywords:* Political Institution, Economic Growth, Inequality, Happiness, Environment, Populism, Protectionism, Causality

<sup>\*</sup>Narita: Department of Economics and Cowles Foundation, Yale University, email: yusuke.narita@yale.edu. I am grateful to Raj Chetty, Suresh Naidu, Rohini Pande, and many seminar participants for their feedback. I also thank exceptional research assistance by Ayumi Sudo, Richard Gong, Tra Nguyen, and Leonardo Fancello.

### I Introduction

Is democracy good for economic growth? This long-standing question is becoming increasingly debatable. Over the past few decades, the remarkable economic rise of China, the setbacks of the Arab Spring, and rising populism in many democracies have fueled new doubts about the advantages of democracy. This concern is well expressed by recent bestseller titles such as *How Democracies Die* and *How Democracy Ends*, and is reinforced by democratic backsliding (Haggard and Kaufman, 2021), where democratic institutions erode or disappear in many countries.

This paper shows that democracy negatively impacts economic growth in the 21st century, a departure from previous findings using 20th-century data. We start by looking at the cross-country correlation between national outcomes and a widely used electoral democracy index. The index quantifies the extent to which the ideal of electoral democracy is achieved by aggregating freedom of association and expression, clean elections, and suffrage. As reported in Figure 1, democracy is associated with lower growth in 2001-2022. This contrasts the 1980s and 1990s, for which periods we and the prior literature find no such negative correlation between democracy and economic growth.

We aim to investigate whether this emerging association of democracy with lower growth has any causality. To identify the causal effect of democracy, we adopt the most influential IVs for current political institutions.

- Mortality of European colonial settlers (Acemoglu et al., 2001), which influenced the types of political institutions Europeans introduced to former colonies
- Population density in the 1500s (Acemoglu et al., 2002), which also influenced the types of political institutions former European colonies adopted
- Legal origin, based on the impact of a European colonizer's legal structure on the colony's eventual political regime (LaPorta et al., 1998)
- Availability of crops and minerals, which reflects historical agricultural endowments and influences political organization through heterogeneous demand for slave labor (Easterly and Levine, 2003)
- Fraction of the population speaking English and a Western European language (Hall and Jones, 1999)

These IVs help identify the effects of political institutions by tracing their origins back to geographical and historical determinants. These determinants of today's democracy level capture the feasibility and incentives of colonial powers to invest in local institution-building and each country's cultural and industrial affinities with Western culture. Indeed, first-stage regressions show that these IVs are significant drivers of the cross-country variation in today's democracy levels. These IVs also pass the J test of overidentification, providing no evidence against the validity of the instruments.

All of these IVs produce similar estimates of democracy's impact, based on Two-Stage Least Squares (2SLS) and two-step efficient Generalized Method of Moment (GMM). These estimates indicate that democracy has a

persistent negative impact on economic growth in this century. The median estimate among our five IV strategies is that a standard deviation increase in the democracy level causes a 2.2 percentage point GDP decrease per year in 2001-2019 (57.9% of the outcome mean) and a 0.9 percentage point GDP decrease per year in 2020-2022 (47.3% of the outcome mean). To facilitate interpretation of the findings, the political-regime difference between China and the US is equivalent to a 3 standard deviation difference in the democracy index in 2019. Put differently, a standard deviation change in the democracy index is equivalent to the political-regime difference between Mozambique and Mexico, or Mexico and Denmark. We also apply recent methodological advances to address the possibility that some of the instruments may be invalid. Allowing for potential IV invalidity does not substantially change our results.

This finding suggests that the way democracy influences growth has fundamentally shifted in the 21st century. To allay concerns that the GDP data may be manipulated, especially by authoritarian governments, we also use night-time light intensity growth as a nonmanipulable proxy for growth. The results remain similar: a one standard deviation increase in the democracy level leads to a 2.3 percentage point decrease in night-time light intensity growth in 2001-13 (33.3% of the outcome mean). Therefore, the adverse effect of democracy on growth is not due to misreporting or mismeasurement by authoritarian governments.

Our finding is robust to various alternative specifications and measurement choices. Controlling for baseline total or per capita GDP has little effect on the estimates. Controlling for latitude, temperature, precipitation, population density, median age, continent dummies, and diabetes prevalence does not change the results. The results change little with alternative indices for democracy or alternative weighting of countries. Moreover, the adverse effect of democracy is robust to excluding outlier nations from the sample. The results are not driven by the US and China alone, nor do the G7 nations drive them. The negative effect of democracy persists across different time periods before, during, and after the Great Recession.<sup>1</sup>

We explore many potential mechanisms that underlie the perverse effect of democracy. One critical factor appears to be the rise in xenophobia and myopia, manifested through both political and economic channels. Consistent with the widespread concern, we find that democracies experienced larger increases in threats to the quality of democracy, especially protectionism, political polarization, hate speech, and populism. These trends can be interpreted as political expressions of xenophobic and myopic tendencies. These political shifts may discourage long-term economic planning. For instance, democracies experienced stagnation in investment in the future (tangible capital formation, intangible R&D and education) and trade with foreign countries (both imports and exports). Democracy also depresses TFP growth. As a result of these factors, 2SLS estimates suggest that democracy slows value-added growth especially in manufacturing and services.<sup>2</sup> Together, these results suggest that since the turn of the twenty-first century, democracy may be increasingly influenced by short-termism and nationalist sentiment,

<sup>&</sup>lt;sup>1</sup>We also analyze a complementary question of the effect of *becoming* more democratic (in contrast to the effect of *being* more democratic at the baseline). Consistent with the above main results, we find that positive changes in democracy level are negatively associated with GDP growth in 2001-2019 and night-time light intensity growth in 2001-2013. However, we cannot conclude a causal relationship due to lack of power.

<sup>&</sup>lt;sup>2</sup>In contrast, other potentially important channels such as taxes, child mortality, domestic conflict, and the number of new businesses appear to play a less significant role in explaining democracy's adverse effect.

undermining key drivers of long-term growth.

A possible interpretation of the negative effect of democracy is that democracies may try to achieve other objectives than average economic growth. We investigate this possibility by examining the impacts of democracy on additional outcomes. We find that democracy negatively affects growth in subjective life satisfaction between 2001 and 2022, which is consistent with our main results. At the same time, we also find that democracy alleviates income inequality between 2001 and 2022. Democracy also slows down the growth in CO2 emissions and energy consumption per capita. This result suggests that democracies may achieve more equality and environmental sustainability at the expense of average growth.

Finally, we provide a supplementary analysis of the Covid-19 pandemic. From 2020 to 2022, the US and other major democracies faced historic recessions and death tolls due to the pandemic. During this period, democracy is associated with bigger shocks to GDP and more excess deaths. These associations turn out to be causal. Our median estimate among our five IV strategies is that a standard deviation increase in a country's democracy index results in 22.4 (s.e. = 8) more excess deaths per 100,000 people (25.5% of the outcome mean). We also provide evidence that a significant channel for democracy's adverse effect appears to be weaker and narrower containment policies at the beginning of the pandemic, rather than the speed of policy implementation.

**Related Literature.** Any cause of national outcomes is difficult to identify due to omitted variable biases, measurement errors, and limited data size. Classic cross-country regression studies claim that democracy's cumulative effect on economic growth may be negligible (Barro, 1997; Przeworski and Limongi, 1993; Przeworski et al., 2000). With more quasi-experimental research designs, however, later studies show that democracies experience more stable, long-term growth than non-democracies (Acemoglu et al., 2019; Aghion et al., 2007; Papaioannou and Siourounis, 2008; Persson and Tabellini, 2006; Quinn and Woolley, 2001; Rodrik and Wacziarg, 2005). Similar findings exist for democracy's positive effects on health, such as child mortality (Besley and Kudamatsu, 2006; Gerring et al., 2012). More broadly defined Western social institutions are also shown to positively affect economic growth (Acemoglu et al., 2001, 2002; Easterly and Levine, 2003; Hall and Jones, 1999).<sup>3</sup>

The prior work chiefly focuses on the 20th and earlier centuries, while we analyze the 21st century. We deliberately use the same quasi-experimental research designs as prior work, so as to see the difference between the 21st and ealier centuries. Our results suggest that the role of democracy in economic growth may differ between this and previous centuries. This finding echoes a growing set of recent facts that challenge the conventional wisdom about economic growth. For example, as opposed to studies from the 1990s, Kremer et al. (2021) and references therein note a trend towards convergence (poor countries catch up with rich) since 2000. See also Acemoglu and Molina (2021) for the causal interpretation of Kremer et al. (2021)'s descriptive finding. Easterly (2019) reports that policy outcomes in inflation, black market premiums, currency overvaluation, real interest rates, and trade shares to GDP started improving in developing countries since the late 1990s. Song et al. (2011)

<sup>&</sup>lt;sup>3</sup>Other studies inspect the micro mechanisms behind democracy's effects. Some studies use regional differences in democratic representation to find that higher representation leads to greater investments in education and public health (Baum and Lake, 2003; Doucouliagos and Ulubaşoğlu, 2008; Tavares and Wacziarg, 2001). Studies such as Besley and Case (2003) and Burgess et al. (2015) focus on how different political processes within countries lead to different income redistributions and provisions of public goods.

document a series of facts about China's unprecedented economic transition and present a new growth model to explain the facts. Autor et al. (2016) and references therein point out that American labor-market adjustments to China's trade shocks challenge much of the received empirical wisdom about economic growth.

Our analysis of 2020-2022 also contributes to the literature on the economics of pandemics. Studies show that obedience to travel restrictions or social distancing differs by government communication and political systems (Allcott et al., 2020; Frey et al., 2020; Grossman et al., 2020; Schmelz, 2021). In contrast to the prior work, which is mostly correlational, this paper sheds light on democracy as a root cause of Covid-19-related outcomes.

We integrate these strands of the literature to find that democracy causes worse economic and social outcomes since the beginning of the 21st century. To our knowledge, this paper seems to be the only study that shows any substantially adverse effect of democracy on any crucial national outcome.

We organize this paper as follows. Section II analyzes the correlation between democracy and national outcomes. Section III presents our 2SLS estimates of the causal effect of democracy. Section IV explores the channels behind democracy's effect. Section VI discusses alternative specifications and measurement choices, nonlinear relationships, and placebo tests using 1980-2000. This section also studies the effect of becoming more democratic. Section VII concludes.

### **II** Democracy is Associated with Lower Growth

Before turning to causal analysis, we first examine whether democracy is correlated with growth in 2001-2022, by using the following types of data. Table A2 provides descriptive statistics for our main variables.<sup>4</sup>

**Outcomes.** We examine two complementary measures of economic growth: GDP growth and night-time light intensity growth. GDP growth rates come from the World Bank. For our sample of 159 countries, the mean annual GDP growth rate between 2001 and 2019 is 3.8% with a standard deviation of 1.8% (Table A2 row 1). We also look at the mean GDP growth rate between 2020 and 2022. Although 2020 was a disastrous year, countries' economies bounced back in 2021 and 2022, yielding an average growth rate of 1.8% during this period.

To address the widespread concern that the GDP growth data may be manipulated and exaggerated, especially in authoritarian countries, we also use mean night-time light intensity growth rate in 2001-2013, sourced from Martinez (2022). Nighttime light intensity, derived from satellite imagery capturing the Earth's surface after dark, serves as an alternative proxy for economic growth by measuring the luminosity associated with human activity. Nighttime light intensity is less prone to manipulation than GDP. For our sample of 155 countries, the mean is 6.8% with a standard deviation of 6.0% (Table A2 row 3), which is higher and more volatile than GDP growth.

**Democracy indices.** Measuring the extent of democracy is tricky. Our baseline measure is the electoral democracy index from the *Varieties of Democracy* (V-Dem) Project. It considers multiple facets of democracy, such as the freedom of association and expression, and clean elections. It is increasingly accepted in the economics

<sup>&</sup>lt;sup>4</sup>Table A1 provides details on data sources. Descriptive statistics for the remaining variables are in Table A3.

and political science literature as a measure of democracy. As shown in Table A2, the index captures our intuitive notion of democratic countries. According to the index, the most democratic countries are Sweden and Denmark, while the least democratic country is Saudi Arabia. As a further sanity check, Table A4 ranks 30 nations with the largest GDP by their democracy levels. As robustness checks, we also use the polity index by the Center for Systemic Peace, the freedom index by Freedom House, and the democracy index by the Economist Intelligence Unit.<sup>5</sup>

**Country characteristics.** To control for country characteristics, we collect country-level data about GDP, absolute latitude, mean temperature, mean precipitation, population density, median age, and diabetes prevalence. We source data from the United Nations, the World Bank, and the International Diabetes Federation.

With this data, we first examine the performance of democratic and authoritarian countries in the 21st century. Figure 1a shows that higher levels of democracy are associated with lower GDP growth rates from 2001 to 2019. For the period 2020–2022, Figure 1b indicates that more democratic nations experienced greater GDP losses due to the pandemic. A similar negative association with democracy is observed for nighttime light-intensity growth, as shown in Figure 1c. Additionally, OLS estimates in Table A6 confirm that democracy is strongly and significantly associated with worse performance in the 21st century. The baseline estimates in column 1 suggest that a one standard deviation increase in the democracy measure corresponds to a 1.6 percentage-point decrease in the annual GDP growth rate over 2001–2019 (s.e. = 0.4), a 1.6 percentage-point decrease in nighttime light-intensity growth rate between 2001 and 2013 (s.e. = 0.3), and, from 2020 onward, a 0.8 percentage-point decrease in the mean annual GDP growth rate (s.e. = 0.2). These patterns hold after controlling for absolute latitude, mean temperature, mean precipitation, population density, and median age, along with baseline GDP per capita and baseline total GDP. For the pandemic period 2020-2022, we additionally control for diabetes prevalence.

### III Causal Effects of Democracy in the 21st Century

#### **A** IVs for Political Regimes

We cannot interpret the above relationship as causal, however. Many omitted determinants of outcomes also correlate with the degree of democracy. To assess whether democracy causes slower growth, we use five established instruments for democracy that reflect historical and geographic determinants.

**European settler mortality.** European settler mortality is the mortality rate (annualized deaths per thousand mean strength) of European soldiers, bishops, and sailors stationed in the colonies between the seventeenth and nineteenth centuries. Europeans used mortality rates to decide where to settle (Curtin, 1989). In colonies with inhospitable germs, they did not settle and established extractive institutions that extracted local resources and

<sup>&</sup>lt;sup>5</sup>The polity index measures democratic and autocratic authority in governing institutions by evaluating executive recruitment, constraints on executive authority, and political competition. Meanwhile, the freedom index focuses more on citizens' political rights and civil liberties. The democracy index by the Economist Intelligence Unit rates democracy holistically by considering electoral processes, government functions, political participation, democratic culture, and civil liberties. Table A5 shows that the indices are highly correlated with each other.

lacked checks and balances against government expropriation. In colonies with hospitable disease environments, Europeans settled and established inclusive institutions that protected individual liberties. The effect of these institutions persists today. Consistent with this hypothesis by Acemoglu et al. (2001), Figure 2a shows that countries with higher European settler mortality have lower democracy levels today. Doubling European settler mortality leads to a 0.8 (s.e. = 0.2) standard deviation decrease in democracy levels (Table A7 column 1). This fact motivates us to use European settler mortality as an IV among ex-European colonies.

**Past population density.** Population density in the 1500s is the number of inhabitants per square kilometer in the 16th century. Sparse populations in the 16th century induced Europeans to settle and develop Western-style institutions, while denser populations made extractive institutions more profitable. As a result, population density at the beginning of the colonial age determined colonial institutions' inclusiveness. Acemoglu et al. (2002) use this IV to show that European institutions positively affect economic growth.<sup>6</sup> Figure 2b confirms that higher population density in the 16th century corresponds to lower democracy levels today. Doubling population density at the beginning of the colonial age is associated with a 0.5 (s.e. = 0.09) standard deviation decrease in democracy (Table A7 column 3). Similar to the European settler mortality IV, we use this IV for ex-European colonies.

**Legal origin.** This IV is a dummy variable for British legal origin that takes the value 1 if the country's legal origin is British (common law) and 0 if it is French, German, or Scandinavian (civil law). Many countries derive their legal systems from European colonization (LaPorta et al., 1998). Such legal origin determines how the law protects civil liberties and political rights. Indeed, first-stage regressions show that countries with a British legal origin are significantly more democratic today, exhibiting a 2.0 (s.e. = 0.6) standard deviation increase in democracy (Table A7, column 5).

**Fraction speaking English or European.** This variable is the fraction of the population speaking English or a major Western European language (French, German, Portuguese, and Spanish) as a mother tongue in 1992. As Hall and Jones (1999) argue, an essential feature of modern world history is the spread of Western European influence, which created an institutional and cultural background conducive to democracy. The language variable is a proxy for such influence. Indeed, the fraction of the population speaking a major European language positively correlates with democracy (Figure 2c).<sup>7</sup> For example, a 100% increase in the fraction of the population speaking a European language is associated with a 1.8 (s.e. = 0.6) standard deviation increase in democracy (line 5 in column 7 of Table A7). Like the original authors, we include all countries in the world in the sample definition. <sup>8</sup>

**Availability of crops and minerals.** Bananas, coffee, maize, millet, rice, rubber, sugarcane, and wheat are dummy variables coded 1 if a country produced the crop in 1990. Copper and silver are coded 1 if a country mined the mineral in 1990.<sup>9</sup> According to Sokoloff and Engerman (2000), certain commodities induced economies of

<sup>6</sup>They also use urbanization in the 1500s as an IV. Using this IV for our analysis produces similar estimates (available upon request).

<sup>&</sup>lt;sup>7</sup>The original specification also uses absolute latitude and the Frankel-Romer trade share as IVs. Our results remain similar with or without these variables as IVs (available upon request).

<sup>&</sup>lt;sup>8</sup>Missing data restricts the actual sample to 132 countries.

<sup>&</sup>lt;sup>9</sup>The binary availability of crops and minerals as of 1990 is a good proxy for historical agricultural endowments (Easterly and Levine, 2003). The reason is that although the quantity produced would endogenously respond to price incentives, institutions, and other country characteristics, whether any of the commodity is produced is likely to reflect exogenous characteristics like soil and climate, which are stable over time.

scale and incentivized slave labor, leading to weaker liberty and rights protection for the broad population. Meanwhile, other commodities encouraged production by middle-class farmers, which induced inclusive institutions. The historical agricultural endowments thus influenced political regimes. Consistent with this narrative by Easterly and Levine (2003), first-stage regressions confirm that several of these IVs are significant determinants of today's democracy levels (Table A7). For example, mining silver in 1990 corresponds to a 1.1 (s.e. = 0.4) standard deviation increase in democracy (column 9, line 13). We include all countries in the world in the base sample.<sup>10</sup>

**Evaluating the Validity of the IVs.** We are aware that these IVs are not ideal. Each IV is likely to be threatened by its own mix of potential measurement errors, omitted variables, and exclusion violations. At the very least, however, we provide suggestive evidence that the IVs satisfy the independence and monotonicity requirements.<sup>11</sup> The IVs also pass the overidentification test, as shown by J statistics in Table 1. Except for one specification with marginally significant rejection (column 7), all the specifications have J-test p-values above 0.05, providing no evidence against the validity of their instruments.

Our strategy is to use these five different IVs as robustness checks with each other. Indeed, Table A9 shows that the correlation among the IVs is limited, suggesting that the different IVs exploit different sources of variation to estimate democracy's effect. Importantly, we find no apparent reason to believe that potential exclusion violations by different IVs lead to biases of the same sign. For example, the European settler mortality IV may have excluded negative effects on growth since worse disease environments may directly hamper economic activities. On the other hand, the population density IV may have excluded *positive* effects on growth thanks to returns to scale and agglomeration effects. These two exclusion violations would result in biases of opposite signs. Table A10 summarizes the likely direction of potential bias for each IV. Different IVs have expected bias of the opposite signs, providing support for the idea of using the different IVs as mutual robustness checks. As a further robustness check, we also implement state-of-the-art methods to allow for potentially invalid instruments (Guo et al., 2018; Guo, 2023).

### **B** IV Estimation

This section presents our main results. With the above IVs, we estimate democracy's impact by the following 2SLS regressions:

$$Y_i = \alpha_2 + \beta_2 Democracy_i + \gamma_2 X_i + \varepsilon_{2i}$$
(1)

$$Democracy_i = \alpha_1 + \beta_1 Z_i + \gamma_1 X_i + \varepsilon_{1i}$$
<sup>(2)</sup>

<sup>&</sup>lt;sup>10</sup>Easterly and Levine's dataset only contains 71 countries. We extend their data-gathering process to cover 142 countries.

<sup>&</sup>lt;sup>11</sup>We test whether the first-stage relationship between the univariate IVs and democracy is monotonic, i.e., of the same sign for different countries. Table A8 evaluates this assumption by estimating the first stage for different groups of countries (created by randomly dividing continents into groups). The first-stage estimates are mostly of the same sign and never have opposite signs with statistical significance. This result suggests that the first-stage relationship satisfies monotonicity. The IVs also achieve covariate balance, i.e., are not significantly correlated with covariates such as the length of the country's name.

The coefficient  $\beta_2$  represents the effect of *Democracy<sub>i</sub>* on  $Y_i$ , the outcome variable, conditional on a varying vector of country characteristics  $X_i$ . Given that *Democracy<sub>i</sub>* is far from randomly assigned, we instrument for *Democracy<sub>i</sub>* by each vector of IVs,  $Z_i$ , in the first-stage equation (2).

Does democracy cause worse economic performance? Reduced-form figures using the IVs suggest so. Figures 2d, 2e, and 2f show that lower European settler mortality causes higher democracy levels, which cause slower economic growth in 2001-2019, bigger shocks to GDP in 2020-2022, and lower nighttime light intensity growth between 2001 and 2013. A higher log of population density in the 1500s is also associated with lower growth outcomes (Figures A1a, A1b, and A1c). Likewise, Figures A1d, A1e, and A1f reveal that the British Legal Origin instrumental variable is also negatively related to these same outcomes.

Table 1's Panel A reports the 2SLS estimates of the effect of democracy, using each of the five IV strategies. They all indicate significant adverse effects of democracy. Columns 1 and 2 show our estimates using log European settler mortality as an IV for our base sample of ex-colonies. The corresponding 2SLS regression estimates in Panel A's column 1 in Table 1 show that a standard deviation increase in the democracy measure causes a -2.5 (s.e. = 0.3) percentage-point decrease for the GDP growth rate in 2001-2019, and -2.8 (s.e. = 0.7) percentage-point decrease for the night-time light intensity growth rate in 2001-2013. Democracy's effect persists in the next decade. We estimate that a standard deviation increase in democracy causes a -0.9 (s.e. = 0.1 percentage-point decrease for the GDP growth rate in 2020-2022. Our confidence in the plausibility of the IV estimates is bolstered by the fact that controlling for various potential sources of omitted variable bias has little impact on our estimates. In column 2, we control for climate, population density, population aging, and diabetes prevalence. The coefficients remain similar.

To check whether the above results are sensitive to the choice of IVs, columns 3 and 4 use population density in the 1500s as an IV for a similar sample of ex-colonies. We continue to find a negative effect of democracy. The 2SLS estimates in Table 1 column 3 show that the effects of a standard deviation increase in the democracy measure are -2.2 (s.e. = 0.4) percentage points decrease for the GDP growth rate per year in 2001-2019, -1.8 (s.e. = 0.4) for the night-time light intensity growth rate in 2001-2013, and -0.9 (s.e. = 0.1) for the GDP growth rate in 2020-2022.

The overall pattern remains the same for the legal origin IV in columns 5 and 6. The corresponding 2SLS estimates in Table 1, Panel A, column 5 show that the effects of a standard deviation increase in the democracy measure are -1.8 (s.e. = 0.5) percentage-points decrease for the GDP growth rate in 2001-2019, -2.5 (s.e. = 0.6) for the night-time light intensity growth rate in 2001-2013, and -0.8 (s.e. = 0.2) for the GDP growth rate in 2020-2022. Adding controls in column 6 preserves the estimates.

Columns 7 and 8 use the fraction of the population speaking English or a European language as IVs. Unlike the previous three IVs, the base sample definition is not limited to former European colonies. Yet, the results remain similar to the previous estimates. Column 7's estimates in Table 1 (Panel A) show that the effects of a standard deviation increase in the democracy measure are -1.0 (s.e. = 0.9) for the GDP growth rate in 2001-2019, -1.6 (s.e. = 1.1) for the night-time light intensity growth rate in 2001-2013, and -0.7 (s.e. = 0.3) for the GDP growth rate in 2020-2022. Controlling for baseline covariates in column 8 barely changes the estimate.

Finally, we use dummies for the ability to grow crops and mine minerals as IVs, finding estimates consistent with our baseline results. The coefficients in Panel A's column 9 in Table 1 are -2.2 (s.e. = 0.5) percentage-point decrease for the average GDP growth rates per year in 2001-2019, -2.3 (s.e. = 0.5) for the night-time light intensity growth in 2001-2013, and -1.0 (s.e. = 0.2) for the GDP growth rate in 2020-2022. The regression with controls in column 10 produces similar results.

In summary, the several different sources of variation in democracy from the historical democratization process lead to similar estimates of the negative impact of democracy. The estimates are also similar to the OLS estimates. It is particularly reassuring that the different IV strategies, which use different sources of variation in democracy, nonetheless produce similar estimates. A majority of these estimates also pass Lee et al. (2022)'s 95% confidence level test, which explicitly allows for the presence of potentially weak IVs. We also perform two-step efficient GMM estimation in Table 1 Panel B and obtain almost the same estimates as in 2SLS.<sup>12</sup>

#### **B.1** Potentially Invalid Instruments

The IVs we use for democracy may not fully satisfy instrument relevance or exclusion restriction. To deal with the concern, in Table 1 Panel C, we implement two recent methods (Guo et al., 2018; Guo, 2023) for IV estimation with potential invalidity in some of the instruments.

The first method is two-stage hard thresholding with voting, which selects a set of valid instruments from a set of candidate instruments. This method uses a three step process to select valid instruments. The first hard thresholding keeps strong IVs based on their correlation with the treatment, and the second hard thresholding generates candidate sets of valid IVs based on satisfying the exclusion restriction and no unmeasured confounding. Majority or plurality voting rules then select the final set of valid IVs using the number of candidate sets each IV appears in as votes. This method selects the valid IVs with probability one as sample size grows large, so the estimator has the same asymptotic distribution as the oracle 2SLS estimator using only valid IVs (see Section 3.5 of Guo et al. (2018)).

The second method, searching-sampling, improves on a potential shortcoming of the first method: confidence intervals produced by two-stage hard thresholding with voting may not be robust to bias from locally invalid IVs, which may be hard to separate from valid IVs due to data limitations. The searching-sampling method constructs uniformly valid confidence intervals that are robust to mistakes in separating valid and invalid IVs. The method searches for a range of treatment effects that permit sufficiently many valid IVs and uses resampling to improve the precision of the confidence interval. Searching sampling produces confidence intervals instead of point estimates

<sup>&</sup>lt;sup>12</sup>We use the ivreg2 implementation of 2SLS and GMM. For an exactly identified model, under the assumptions of conditional homoskedasticity and independence, the efficient GMM estimator coincides with the traditional 2SLS estimator. The two estimators differ in more general cases (though they share the same estimand). In two-step efficient GMM, the efficient or optimal weighting matrix is the inverse of an estimate of the covariance matrix of orthogonality conditions. The efficiency gains of this estimator relative to the traditional 2SLS estimator derive from the use of the optimal weighting matrix, the overidentifying restrictions of the model, and the relaxation of the i.i.d. assumption. For further details, see Hayashi (2000), pp. 206-13 and 226-27.

with standard errors.<sup>13</sup>

Table 1 Panel C shows the results from applying both methods to the 2SLS specifications with multiple instruments. Both methods may choose an empty set of valid IVs when candidate IVs are too weak conditional on controls. We are unable to report estimates for such cases. Overall, we find results consistent with those in Panels A and B, although standard errors tend to be larger.

### IV Mechanisms Behind Democracy's Adverse Effect

We now turn to potential mechanisms. What explains democracy's negative impact on growth? Our analysis documents that the effect of democracy is reversed between the 21st and earlier centuries. This reversal suggests that the key mechanisms cannot be stable factors like property rights institutions, which remained similar between the 21st and late 20th centuries. We obtain data on other potential mechanisms that might experience key changes in the last few decades. Our data cover different mechanisms, including political, economic, demographic, and educational channels, from 2001 to 2019. We list and summarize these variables in Table A2.

For the mean annual growth rate of each potential mechanism  $M_i$  in 2001-2019, we estimate the following 2SLS equations:

$$M_i = \alpha_2 + \beta_2 Democracy_i + \gamma_2 X_i + \varepsilon_{2i}$$
(3)

First Stage: 
$$Democracy_i = \alpha_1 + \beta_1 Z_i + \gamma_1 X_i + \varepsilon_{1i}$$
. (4)

In the following, we report the coefficient  $\beta_2$ , which captures the percentage points increase in the mean annual growth rate of  $M_i$  in response to a standard deviation (s.d.) increase in the democracy index. This specification mirrors our main 2SLS specification except that the second-stage dependent variable  $M_i$  is a mechanism rather than an outcome. This approach is similar to Acemoglu et al. (2003)'s, which evaluates channels behind democratization's effects using a similar 2SLS. All of the following results are robust to controlling for the baseline level of the mechanism variable, absolute latitude, mean temperature, mean precipitation, population density, and median age (results are available upon request).

**Political Mechanisms.** We first find that in this century, democracies experience greater increases in threats to quality democracy. In Table 2 Panel 1 (columns 1–4), we consider four political mechanisms: Protectionism (measured by a formula incorporating tariffs and other regulatory trade barriers), populism (the extent to which representatives of a political party use populist rhetoric), hate speech (how often major political parties use hate speech in their rhetoric), and polarization (how significant the differences of opinion are on major political issues among major political parties). The overidentified 2SLS estimates in Panel G indicate that for 2001–2019, a standard deviation increase in democracy increases protectionism by 0.4 s.d. (s.e. = 0.1 s.d.), populism by 0.6 s.d.

<sup>&</sup>lt;sup>13</sup>In Table 1 Panel C, we report point estimates and standard errors derived from 95% confidence intervals. We assume searching sampling produces an equal bandwidth confidence interval around a normally distributed point estimate.

(s.e. = 0.2 s.d.), hate speech by 0.5 s.d. (s.e. = 0.1 s.d.), and political polarization by 0.6 s.d. (s.e. = 0.08 s.d.). This result is consistent with the widespread concern that democracies become more myopic and xenophobic. Other 2SLS estimates are similar, though they tend to be less precisely estimated than the overidentified estimates. In the remainder of this section, we report the overidentified estimates in Panel G.

**Economic Mechanisms.** These political changes come with analogous economic changes. Consistent with the rising populism in democracies, democracy decreases growth in tangible and intangible investment in the future. For example, column 5 in Table 2 shows that a standard deviation increase in democracy decreases the growth of capital stock formation by -0.6 percentage points (35.3% of the mean, s.e. = 0.2 percentage points). Capital stock formation consists of outlays on additions to the fixed assets of the economy plus net changes in the level of inventories. A similar effect is observed for R&D investments in column 6 of Table 2, where the mean annual growth rate decreases by 2.1 percentage points (36.8% of the mean, s.e. = 0.5 percentage point) per a standard deviation increase in democracy.

We observe similar negative impacts of democracy on education, which can also be interpreted as intangible social investment. As shown in columns 11 and 12 of Table 2, from 2001 to 2019, a standard deviation increase in democracy reduces the mean annual growth rate of primary school enrollment by 0.3 percentage points (60% of the mean, s.e. = 0.1 percentage points) and that of secondary school enrollment by 1 percentage point (58.8% of the mean, s.e. = 0.2 percentage points). Potentially as a result of stagnating investment, democracy also depresses TFP growth. As shown in Table 2 column 8, the mean annual growth rate of TFP experiences a reduction of 0.4 percentage points (s.e. = 0.1 percentage points, 50.0% of the mean) in response to a standard deviation increase in democracy.<sup>14</sup>

Consistent with increasing protectionism, democracies also experience stagnation in trade growth. Panel A column 9's overidentified estimate is -3.9 percentage points (s.e. = 0.4 percentage points, 40.6% of the mean) decrease in the mean annual growth rate of the import value in 2001-2019. Column 10 exhibits similar estimates for exports, with the overidentified estimate being a 3.4 percentage point decrease in the mean annual growth rate (s.e. = 0.5 percentage points, 34.7% of the mean) in response to a standard deviation democracy increase. <sup>15</sup>

As a result of these factors, democracy slows value-added growth in manufacturing and services. Value added is the net output of a sector after adding up all the output and subtracting intermediate input. In columns 1-3 of Table A11, we analyze the effect of democracy on the value added by sector. Column 1's overidentified estimate indicates that a standard deviation increase in democracy reduces the mean growth rate of the value added in agriculture by -1.3 percentage points (s.e. = 0.4 percentage points, 18.8% of the mean). Similarly, columns 2-3 show that democracy significantly dampens value-added growth in manufacturing and services. The overidentified

<sup>&</sup>lt;sup>14</sup>The impact of democracy on labor force growth is unclear. As shown in column 7, the overidentified estimate is that a standard deviation increase in democracy insignificantly causes the mean annual growth rate of the labor force (the number of people aged 15 and older who supply labor for the production of goods and services) to decrease.

<sup>&</sup>lt;sup>15</sup>A potential explanation for the dampening effect of democracy on trade is that electoral competition could lead to trade barriers (Anderson et al., 2013). We also check whether democratic nations are less likely to trade with China in Table A11. We run 2SLS regressions with the share of the total value of imports from China or exports to China in GDP as the outcome variable. We find negative effects of democracy on imports from and exports to China.

estimate in column 2 is that a standard deviation increase in democracy causes a -2 percentage points (s.e. = 0.7 percentage points, 23.8% of the mean) decrease in the mean annual growth rate of manufacturing value added in 2001-2019. For services, column 3's overidentified estimate is -3.7 percentage points (s.e. = 0.4 percentage points, 40.7% of the mean) decrease in the mean annual growth rate.

Note that value-added growth can be decomposed into changes in capital input, labor input, and productivity.<sup>16</sup> The above results suggest that the slower value-added growth in manufacturing and services is primarily caused by less capital investments and lower productivity, rather than less labor. Ultimately, our results suggest that democracy might have stopped improving building blocks (investment in the future and trade with other countries) for growth, along with the degradation of democracy in the form of populism, protectionism, and hate speech, as summarized in the picture.<sup>17</sup> We also confirm that these mechanisms are not active during prior decades in the past century (results available upon request).



**Other Mechanisms.** In contrast, demographic channels appear less important in explaining the adverse effect of democracy. As shown in columns 13-14 of Table 2, democracy causes only insignificant effects on population growth. The effect on median age is even negative. In Table A11, we also explore other potentially important channels, such as tax revenue growth, R&D researchers, the number of new registered businesses, foreign direct investment (FDI), domestic conflict, and child mortality. In column 11, we find that, at the overidentified estimate (Panel G), a standard deviation increase in democracy slows the mean annual growth rate of tax revenue by 1 percentage point (s.e. = 0.4 percentage points, 71.4% of the mean). Additionally, in column 11, we find that a

<sup>&</sup>lt;sup>16</sup>Change in value-added in industry *i* in period *t* can be expressed in logarithms as  $\Delta \ln V_{i,t} = \frac{\bar{v}_{K_{i,t}}}{\bar{v}_{V_{i,t}}} \Delta \ln K_{i,t} + \frac{\bar{v}_{L_{i,t}}}{\bar{v}_{V_{i,t}}} \Delta \ln L_{i,t} + \frac{1}{\bar{v}_{V_{i,t}}} \Delta \ln A_{i,t}$ . Here  $\Delta \ln K_{i,t} \Delta \ln L_{i,t}$  and  $\Delta \ln A_{i,t}$  denote the log change in capital investments, labor inputs, and productivity of industry *i* in period *t*, respectively.  $v_{K_{i,t}}$  and  $v_{L_{i,t}}$  are the shares of capital or labor compensation in the gross output of industry *i*.  $v_{V_{i,t}}$  is the share of industry value added in industry gross output. The bars denote averages for the current and previous periods, *t* and *t* - 1. See Jorgenson (2008) for details.

 $<sup>^{17}</sup>$ It is possible that the degradation of democracy was induced by the spread of the mobile web and the social media, though it is hard to empirically prove it due to the lack of data.

standard deviation increase in democracy impacts child mortality, with its mean annual growth rate increasing by 1.9 percentage points (s.e. = 0.2 percentage points, 55.8% of the mean) as the overidentified estimate (Panel G). For the other aforementioned mechanisms of Table A11, we do not find a strong causal effect of democracy.

Appendix A.2 provides a separate analysis of policy channels in the pandemic period (2020-22). A major channel for democracy's adverse effect in this period appears to be weaker and narrower containment policies at the beginning of the pandemic rather than the speed of containment policy implementation.

### V Life Satisfaction, Inequality, and the Environment

The negative effect of democracy on average economic growth may be a side effect of democracies' efforts to achieve other objectives. To study this possibility, we analyze the impacts of democracy on additional outcomes, especially life satisfaction, economic inequality, and environmental burden. Consistent with the negative effect of democracy on economic growth, we also find a negative impact of democracy on life satisfaction, especially during the pandemic period. Figures A2b and A2c report a negative association between democracy and life satisfaction in 2001–2019 and a more pronounced negative association during 2020–2022. Causally, the median estimate among our five IV strategies in Panel A in Table A12 is that a standard-deviation increase in democracy reduces the mean growth rate of life satisfaction by 1.6 percentage points (s.e. = 0.2) during 2020–2022 (third line of column 9) and by 0.1 percentage points (s.e. = 0.4) during 2010–2019 (second line of column 9).<sup>18</sup> Therefore, even subjective well-being lag under democratic regimes in this period.

More nuanced views emerge from results on inequality and environmental burden. In Figures A2d and A2e, democracy is slightly negatively associated with the growth of the top 1% income share in 2001–2019 but is positively associated with it in 2020–2022. Moreover, according to the median 2SLS estimates in Table A12, a standard-deviation increase in democracy reduces the mean annual growth rate of the top 1% income share by -0.4 percentage points (s.e. = 0.2) during 2001–2019 (fourth line of column 9) and increases it by 0.5 percentage points (s.e. = 0.1) during 2020–2022 (fifth line of column 9).

In Figures A2f, A2g, A2h, and A2i, we find a strongly negative association between democracy and environmental burden. Additionally, the median 2SLS estimates in Table A12 suggest that a standard deviation increase in democracy slows the mean annual growth rate of CO<sub>2</sub> emissions per capita by 2.7 percentage points (s.e. = 0.5) during 2001–2019 (sixth line of column 7) and by 1.6 percentage points (s.e. = 0.4) during 2020–2022 (seventh line of column 5). Similarly, a one-standard-deviation increase in democracy decreases per-capita energy consumption by 2.5 percentage points (s.e. = 0.4) during 2001–2019 (eighth line of column 5) and by 1.5 percentage points (s.e. = 0.3) during 2020–2022 (ninth line of column 7).

These results suggest that democracies' economic growth may slow as they pursue more complex objectives,

 $<sup>^{18}</sup>$ Democracy is also positively associated with excess deaths during the Covid-19 period, as shown in Figure A2a. Furthermore, in Table A12, Panel A, the median estimate (first line of column 3) suggests that a standard-deviation increase in democracy increases the average excess deaths per 100,000 people by 22.4 (s.e. = 8.1).

such as economic equality and environmental sustainability. This creates a trade-off that must be weighed by policymakers.

### VI Discussion

#### **A** Alternative Specifications and Measurements

Our analysis may be sensitive to measurement and modeling choices, such as how to measure economic performance, how to measure democracy, which countries and periods to analyze, whether to control for baseline GDP and other important characteristics, and how to weight countries. Extreme nations may also be driving our results. Below we check whether these concerns threaten our findings.

**GDP** per capita and light intensity per capita growth as the outcome. We check whether our results hold for the per-capita GDP and per-capita light intensity growth rates in Table A13. We continue to observe a negative democracy effect.

Alternative democracy indices. We adopt alternative democracy indices by Polity (the Center for Systemic Peace), Freedom House, and the Economist Intelligence Unit. These indices are modestly correlated with each other (Table A5). Importantly, Table A14 confirms that our results stay similar regardless of which democracy index is used. Our results are thus not dependent on the particular V-Dem index.

**Separating the Great Recession.** We check if the Great Recession drives our results. Table A15 conducts the same analysis separately for growth rates before, during, and after the recession period (2008-9). We find negative effects of democracy in every one of the periods.

Alternative sample definitions. To check if our results are driven by a few countries, such as the US and China, we show our results without the two countries in Table A16. We re-estimate our preferred specification without outlier countries with a standardized residual above 1.96 or below -1.96 in Table A17. Furthermore, we remove G7 countries from the sample in Table A18. In all cases, we continue to estimate democracy's adverse effects. Limiting the sample to G20 countries also produces similar results. Thus, the negative impact of democracy is a global phenomenon and not driven by a handful of countries.

Alternative weightings. Our 2SLS results so far weight countries by baseline GDP. We believe that baseline GDP weighting is reasonable, especially when the outcomes are GDP growth rates. Nonetheless, we compare our results with weighting by population or no weighting in Table A19. The qualitative pattern is the same among the three ways to weight countries.

**Control for baseline GDP.** We test whether our results are due to the mechanical reason that more developed countries tend to grow slower. Table A20 runs regressions with baseline total GDP or GDP per capita as controls. The resulting estimates continue to find democracy's negative effect, confirming that baseline GDP conditions do not drive our results.

Control for continents. We additionally control for dummy variables for each continent in Table A21. Al-

though the estimates are less precisely estimated, we continue to observe democracy's negative effect on economic growth and public health. This suggests that the democracy treatment is significant within each continent.

#### **B** 21st vs 20th Centuries.

It is natural to ask whether our finding is unique to the 21st century. Additional evidence suggests so. We apply the same descriptive and IV analyses to data from the 20th century. The resulting estimates show that the negative association between democracy and economic growth did not exist in 1981-1990 or 1991-2000 (Figure A3). More importantly, for the same period, we do not observe a significantly negative causal effect of democracy (Table A22). The reduced-form relationships between the IVs and economic growth in 1981-2000 are either insignificant or of the opposite sign to those in 2001-2020. These results suggest that democracy's growth effects worsened specifically in the 21st century. Thus, how democracy matters for economic growth might have changed around the turn of the 21st century.

Barro and Sala-i-Martin (2004) find an inverse U-shaped relationship between democracy (electoral rights) and economic growth using data from 1965–1994. That is, for the early period, democracy is positively associated with growth for non-democratic countries, but is negatively associated with growth for democratic countries. We also find the same inverse U-shaped relationship in 1981–2000. However, the relationship changes and becomes monotonically downwards sloping in 2001–2020, i.e., democracy becomes globally negatively associated with growth (Figures A4 and A5). This finding further supports our results and suggests that the way democracy matters for growth has changed around the turn of the 21st century. We also test whether the causal relationship between democracy and growth is nonlinear. We find no conclusive evidence, mainly due to the small data size. Results are available upon request.

#### C Effects of Becoming More Democratic

We have focused on studying the adverse effect of being more democratic at the baseline. A related problem is the effect of *becoming* more democratic (i.e., democratization) in this century. This question is especially relevant due to the recent concern about democratic backsliding. In the past decade, several countries have experienced changes towards less democratic institutions and cultures. The effect of democratization could be positive, especially because democracy change in 2001–2019 is slightly negatively associated with baseline democracy level in 2000 (Figure A6c). However, Figures A6a and A6b show that the correlation between economic growth and democracy change (the average annual change in the democracy index) is negative. Democracy change is associated with lower GDP growth in 2001–2019 and lower night-time light intensity growth in 2001–2013.<sup>19</sup> Table A23 panel B provides OLS estimates of the negative correlations. We also find similar negative correlations when using baseline democracy change between 1999 and 2000 instead. Table A23 presents 2SLS estimates of the effects of

<sup>&</sup>lt;sup>19</sup>Standard errors are large for the correlation between night-time light intensity growth and democracy change. The association between the variables could be neutral or positive.

democracy change on growth outcomes using the same set of instruments from before. Due to a weak first stage, however, we cannot detect a significant causal effect.

In Appendix Section A.2, we further explore the causal effect of democratization on growth under the dynamic panel-data framework of Acemoglu et al. (2019). We only find inconclusive evidence, yet the points estimates suggest that the effect of democratization may be weaker after 2000 than before that. The causal effects of democracy change on economic growth in the 21st century are therefore not conclusive but are largely consistent with our main results.

## VII Conclusion

Skepticism about the performance of democratic political regimes is as old as the invention of democracy:

"having them [the multitude of the citizens] take part in the greatest offices is not safe: through injustice and imprudence they would act unjustly in some respects and err in others." (Aristotle, Politics, 1281b25)

We bring data and five different IVs to find that democracy has a significant negative effect on economic growth in the 21st century — in sharp contrast to 20th-century findings. The channels likely behind the negative effect of democracy are rising myopic and xenophobic tendencies in democracies, which are associated with stagnant investment and trade growth. The negative impact of democracy remains strong during the pandemic, in which democracy causes not only worse GDP shocks, but also more excess deaths. Overall, political institutions still matter for economic growth, but how they matter might have changed between the prior and current centuries.

We do not argue that democracy is undesirable overall, or that authoritarianism is preferable, for at least two reasons. First, democracy has normative and procedural virtues, regardless of whether they result in good economic outcomes. Second, despite our findings on democracy's negative impacts in the 21st century, democracies may produce better outcomes in the long run or other aspects. Indeed, we find that democracy causes improvements in environmental burden for the same period, pointing to a trade-off between economic growth and social or environmental outcomes. Understanding how democracy can adapt to meet both economic and social goals is a central question for scholars and policymakers alike.

Figure 1: Correlation Between Democracy and Growth

(a) Mean GDP Growth Rate in 2001-2019



(b) Mean GDP Growth Rate in 2020-2022



(c) Mean Night-time Light Intensity Growth Rate in 2001-2013



*Notes:* This figure shows the correlation between democracy and growth outcomes. Panels (a) and (b) display the mean GDP growth rates in 2001-19 and 2020-22, respectively. Panel (c) examines the mean night-time light intensity growth rate in 2001-13. The size of each circle (country) is proportional to its baseline GDP. The colors represent the level of democracy, with warmer colors indicating higher levels of democracy. The line represents the OLS regression fitted line without controls, weighted by countries' baseline GDP. The shaded area corresponds to the 95% confidence interval. Variable definitions and data sources are in Appendix Table A1.

### (a) First-stage: Log European Settler Mortality IV



(b) First-stage: Log Population Density in 1500s IV



(c) First-stage: Fraction Speaking European IV



(e) Reduced form: Mean GDP Growth Rate in 2020-2022







*Notes:* Panels (a), (b), and (c) show the first-stage relationships between democracy in 2019 and three univariate IVs: the log European settler mortality IV, the log population density in 1500s IV, and the fraction speaking European IV. Panels (d), (e), and (f) show the reduced-form relationship between the log European settler mortality IV and three outcomes: the mean GDP growth rate in 2001-2019, the mean GDP growth rate in 2020-2022, and the mean night-time light intensity growth rate in 2001-2013. The Democracy Index (V-Dem) is normalized to have mean zero and standard deviation one in each year. The size of each circle (country) is proportional to its baseline GDP. The colors depend on the level of the democracy index (warmer colors for democracy and darker colors for autocracies). The line is the OLS regression fitted line without controls and weights countries by baseline GDP. The shaded area corresponds to the 95% confidence interval. Variable definitions and data sources are in Appendix Table A1.

(d) Reduced form: Mean GDP Growth Rate in 2001-2019

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
Panel A: Two-Stage Least Squa	res		De	pendent Var	iable is N	Mean GF	P Growt	h Rate i	n 2001-20	019		
Democracy Index (V-Dem, 2000)	-2.5	-4.2	-2.2	-3.1	-1.8	-1.8	-1.0	-1.1	-2.2	-1.6	-2.1	-2.6
Democracy findex (V-Dem, 2000)	(0.3)	(2.7)	(0.4)	(0.6)	(0.5)	(1.3)	(0.9)	(0.6)	(0.5)	(0.6)	(0.2)	(0.2)
	0.00	0.13	0.00	0.00	0.00	0.18	0.28	0.06	0.00	0.01	0.00	0.00
	0.00			ariable is M								0.00
Democracy Index (V-Dem, 2000)	-2.8	-5.3	-1.8	-1.3	-2.5	-3.2	-1.6	-3.0	-2.3	-2.7	-2.0	-2.0
•	(0.7)	(3.8)	(0.4)	(0.6)	(0.6)	(1.0)	(1.1)	(1.0)	(0.5)	(0.5)	(0.2)	(0.4)
	0.00	0.17	0.00	0.02	0.00	0.00	0.16	0.00	0.00	0.00	0.00	0.00
			De	pendent Var	iable is N	Mean GE	P Growt	h Rate i	n 2020-20	022		
Democracy Index (V-Dem, 2019)	-0.9	-0.8	-0.9	-1.0	-0.8	-0.7	-0.7	-0.6	-1.0	-0.8	-0.9	-1.0
	(0.1)	(0.2)	(0.1)	(0.1)	(0.2)	(0.1)	(0.3)	(0.2)	(0.2)	(0.2)	(0.1)	(0.1)
	0.00	0.00	0.00	0.00	0.00	0.00	0.02	0.02	0.00	0.00	0.00	0.00
IVs	settler	mortality	populati	on density	legal	origin	lang	uage	crops &	& minerals	all	IVs
Number of IVs	1	1	1	1	1	1	2	2	10	10	15	15
F-Statistic (First stage)	9.7	7.3	27.2	135.2	12.0	16.8	4.3	14.0	6.4	6.0	57.5	351.8
J-Statistic (p-value)		110	27.2	10012	12.0	10.0	0.03	0.22	0.48	0.12	0.45	0.31
N	81	81	86	86	90	90	132	132	138	138	71	71
Panel B: Two-Step Efficient GM		01	00		70	20	152	152	150	150	,1	/1
Taker D. Two-Step Enterent Giv	1171		De	pendent Var	iable is N	Mean GE	P Growt	th Rate i	n 2001-20	019		
Democracy Index (V-Dem, 2000)	-2.5	-4.2	-2.2	-3.1	-1.8	-1.8	-0.5	-1.2	-1.9	-1.6	-2.1	-2.6
• · · · ·	(0.3)	(2.7)	(0.4)	(0.6)	(0.5)	(1.3)	(0.9)	(0.6)	(0.3)	(0.5)	(0.1)	(0.2)
	0.00	0.13	0.00	0.00	0.00	0.18	0.54	0.04	0.00	0.00	0.00	0.00
		De	pendent V	ariable is M	ean Nigh	ttime Li	ght Inten	sity Gro	wth Rate	in 2001-201	3	
Democracy Index (V-Dem, 2000)	-2.8	-5.3	-1.8	-1.3	-2.5	-3.2	-1.7	-2.6	-2.8	-2.7	-2.1	-2.2
	(0.7)	(3.8)	(0.4)	(0.6)	(0.6)	(1.0)	(1.1)	(0.9)	(0.3)	(0.4)	(0.1)	(0.3)
	0.00	0.17	0.00	0.02	0.00	0.00	0.13	0.00	0.00	0.00	0.00	0.00
			De	pendent Var	iable is N	Mean GE	P Growt	h Rate i	n 2020-20	022		
Democracy Index (V-Dem, 2019)	-0.9	-0.8	-0.9	-1.0	-0.8	-0.7	-0.7	-0.6	-1.0	-0.9	-0.9	-0.9
	(0.1)	(0.2)	(0.1)	(0.1)	(0.2)	(0.1)	(0.3)	(0.2)	(0.1)	(0.2)	(0.1)	(0.1)
	0.00	0.00	0.00	0.00	0.00	0.00	0.02	0.01	0.00	0.00	0.00	0.00
IVs	settler	mortality	populati	on density	legal	origin	lang	uage	crops &	& minerals	all	IVs
Panel C: IV Estimation with Po	tentially											
		I) Two	-Stage Ha	rd Thresho	lding			Ι	I) Search	ing-Sampli	ng	
			De	pendent Var	iable is N	Mean GE	P Growt	h Rate i	n 2001-20	019		
Democracy Index (V-Dem, 2000)		-1.3	-0.8	-0.5	-2.8	-1.0	•				-2.2	-4.6
		(1.0)	(1.0)	(1.0)	(0.9)	(1.9)					(1.3)	(1.8)
		0.21	0.43	0.61	0.00	0.62					0.10	0.01
		De	pendent V	ariable is M	ean Nigh	ttime Li	ght Inten	sity Gro	wth Rate	in 2001-201	.3	
Democracy Index (V-Dem, 2019)		-1.3	-1.4	-1.0	-3.2	-1.3				-1.3	-3.4	-3.6
		(1.5)	(2.2)	(1.3)	(1.1)	(1.1)				(1.8)	(1.0)	(1.8)
		0.41	0.52	0.45	0.00	0.22		1 D / '		0.47	0.00	0.05
		0.5		pendent Var			PP Grow	in Kate i			a =	
Democracy Index (V-Dem, 2000)	-1.0	-0.9	-0.9	-0.8	-0.9	-0.8		•	-0.7	-0.1	-3.7	-1.6
	(0.3)	(0.4)	(0.2)	(0.2)	(0.3)	(0.7)	•	•	(0.2)	(0.7)	(1.9)	(1.2)
n./	0.00	0.02	0.00	0.00	0.01	0.23		•	0.00	0.93	0.05	0.20
IVs		guage	-	t minerals		IVs	lang	•		k minerals		IVs
Number of IVs	2	2	10	10	15	15	2	2	10	10	15	15

Table 1	: 2SLS	and	GMM	Estimates	of l	Democracy'	s Effects
---------	--------	-----	-----	-----------	------	------------	-----------

*Notes:* Panel A reports the 2SLS estimates of democracy's effect on the mean GDP growth rate in 2001-2019, the mean nighttime light intensity growth rate in 2001-2013, and the mean GDP growth rate in 2020-2022, using five different IV strategies. The Democracy Index (V-Dem) is normalized to have mean zero and standard deviation one in each year. For IVs, columns 1 and 2 use log European settler mortality, columns 3 and 4 use log population density in the 1500s, columns 5 and 6 use British legal origin, columns 7 and 8 use the fraction speaking English and the fraction speaking European, columns 9 and 10 use the ability to grow crops and mine minerals, and columns 11 and 12 use all the IVs together. Robust standard errors are in parentheses. The p-values, presented under the standard errors, are displayed as 0.00 if they are strictly smaller than the 0.005 threshold. The F-statistics are from the first-stage regressions of the IVs against the democracy index in 2019. The corresponding first-stage coefficients are in Appendix Table A7. The J-statistic (p-value) shows the p-value from the overidentification J test of the null hypothesis that the instruments are valid. A low p-value suggests that some of the instruments may be invalid. Columns 1, 3, 5, 7, 9, and 11 have no controls, while columns 2, 4, 6, 8, 10, and 12 have the following baseline controls: absolute latitude, mean temperature, mean precipitation, population density, and median age. For the mean GDP growth rate in 2020-2022, we also control for diabetes prevalence. N refers to the number of countries for which IVs, controls, and outcomes are available. Panel B reports the two-step efficient GMM estimates. Panel C reports the estimates of democracy's effect using two methods to allow for potentially invalid IVs, columns 1–2 and 7–8 use the fraction speaking European, columns 7–12 use searching-sampling to select valid IVs. For potential IVs, columns 1–2 and 7–8 use the fraction speaking European. Variable definitions and data sources are in Appendix

		1) Politica	al Channels			2)	Economic Ch	annels			3) Educati	on Channels	4) Demogra	phic Channels
					Capital	R&D					Primary	Secondary		
	Protectionism Growth	Populism Growth	Hate Speech Growth	Polarization Growth	Investment Growth	Investment Growth	Labor Force Growth	TFP Growth	Imports Growth	Exports Growth	Education Growth	Education Growth	Population Growth	Median Age Growth
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)
						Panel	A: OLS			. ,		. /	. ,	. ,
Democracy Index (V-Dem, 2000)	0.2	0.4	0.4	0.4	-0.5	-3.7	-0.5	-0.2	-3.2	-3.0	-0.1	-0.6	-0.4	-0.3
• • • •	(0.1)	(0.2)	(0.1)	(0.1)	(0.10)	(2.0)	(0.3)	(0.3)	(0.6)	(0.6)	(0.05)	(0.1)	(0.2)	(0.06)
Ν	115	148	160	160	125	125	159	113	115	115	146	138	154	154
				Р	anel B: Instru	ument for De	mocracy by Se	ttler Mor	ality					
Democracy Index (V-Dem, 2000)	0.7	0.6	-0.3	-0.08	-0.6	-2.9	-0.4	-0.6	-5.0	-4.5	-0.1	-1.6	-0.3	-0.6
	(0.3)	(0.3)	(0.6)	(0.4)	(0.2)	(1.0)	(0.4)	(0.2)	(0.8)	(0.7)	(0.1)	(0.3)	(0.2)	(0.2)
N	69	77	82	82	71	65	82	59	61	61	78	71	82	82
				Panel C	C: Instrumen	t for Democra	cy by Populat	ion Densi	y in 1500s					
Democracy Index (V-Dem, 2000)		0.7	0.2	0.4	-0.6	-2.2	-0.2	-0.4	-4.3	-3.8	-0.4	-0.9	-0.1	-0.4
	(0.2)	(0.3)	(0.3)	(0.2)	(0.2)	(0.6)	(0.3)	(0.2)	(0.8)	(0.8)	(0.2)	(0.3)	(0.2)	(0.1)
N	69	80	87	87	71	67	87	60	62	62	81	73	86	86
							Democracy by		_					
Democracy Index (V-Dem, 2000)		0.8	0.3	0.3	-0.6	-2.7	-0.4	0.05	-4.4	-3.8	-0.03	-2.3	-0.01	-0.6
	(0.4)	(0.4)	(0.4)	(0.3)	(0.2)	(1.0)	(0.4)	(0.5)	(0.8)	(0.7)	(0.2)	(0.6)	(0.1)	(0.2)
N	71	84	91	91	73	68	91	62	65	65	85	77	90	90
							Democracy by	0 0						
Democracy Index (V-Dem, 2000)		2.3	0.7	1.4	-0.7	-4.0	0.05	-0.2	-4.0	-2.8	-0.2	-0.5	0.3	-0.6
	(0.4)	(1.4)	(0.5)	(0.8)	(0.2)	(1.7)	(0.5)	(0.4)	(0.9)	(1.3)	(0.1)	(0.2)	(0.5)	(0.2)
N	107	122	133	133	106	106	132	101	97	97	125	118	130	130
	0.6	0.7	0.00				ocracy by Cro	-		4.2	0.0	0.5	0.0	0.5
Democracy Index (V-Dem, 2000)		0.7	-0.08	0.2	-0.6	-2.4	-0.5	-0.3	-4.7	-4.3	-0.2	-0.5	-0.3	-0.5
	(0.3)	(0.4)	(0.4)	(0.3)	(0.2)	(0.9)	(0.3)	(0.3)	(0.8)	(0.8)	(0.07)	(0.3)	(0.2)	(0.2)
N	107	127	139	139	109	108	138	99	101	101	130	122	137	137
Democratic Later (V. Democratic)	0.4	0.6	0.5	0.6	0.6		Use all IVs	0.4	2.0	2.4	-0.3	1.0	0.05	0.2
Democracy Index (V-Dem, 2000)		0.6	0.5	0.6 (0.08)	-0.6	-2.1	-0.1	-0.4 (0.1)	-3.9	-3.4	-0.3 (0.1)	-1.0	-0.05	-0.3
N	(0.1)	(0.2) 67	(0.1) 72	(0.08)	(0.2) 63	(0.5) 58	(0.3) 72	(0.1)	(0.4) 52	(0.5) 52	(0.1)	(0.2) 62	(0.1) 72	(0.06) 72
N	62	0/	12	12	63	58	12	52	52	52	68	62	12	12
Outcome Mean	N/A	N/A	N/A	N/A	1.7	5.7	1.9	0.8	9.6	9.8	0.5	1.7	1.6	0.9

#### Table 2: Mechanisms Behind Democracy's Effect in 2001-2019

*Notes:* This table reports the OLS (Panel A) and 2SLS regression (Panels B-G) estimates of democracy's effect on potential mechanisms in 2001-2019. The Democracy Index (V-Dem) is normalized to have mean zero and standard deviation one in each year. Protectionism measures a country's aversion to international trade; Populism measures the extent to which representatives of a political party use populist rhetoric; Hate Speech measures how often major political parties use hate speech in their rhetoric; Polarization indicates how significant the differences of opinions are on major political issues among major political parties in a society; Capital Investment is the gross capital formation for a specific country; R&D Investment is the R&D expenditure expressed as a share of GDP; Labor Force is the total number of workers above the age of 15 in a specific country; Exports is the total value of exports of goods and services in US dollars; Primary Education and Secondary Education represent the enrollment rates for primary education, respectively. For IVs, Panel B uses log European settler mortality, Panel C uses log population density in the 1500s, Panel C uses British legal origin, Panel E uses the fraction speaking English and the fraction speaking European, Panel F uses the ability to grow crops and mine minerals, and Panel G uses all the IVs together. Robust standard errors are in parentheses. We report the global mean of the dependent variables in the bottom row. The specifications do not include controls. N refers to the number of countries for which IVs and outcomes are available. Variable definitions and data sources are in Appendix Table A1.

### References

- Acemoglu, D., S. Johnson, and J. A. Robinson (2001). The Colonial Origins of Comparative Development: An Empirical Investigation. *American Economic Review 91*(5), 1369–1401.
- Acemoglu, D., S. Johnson, and J. A. Robinson (2002). Reversal of Fortune: Geography and Institutions in the Making of the Modern World Income Distribution. *Quarterly Journal of Economics 117*(4), 1231–1294.
- Acemoglu, D. and C. Molina (2021). Converging to Converge? A Comment. In Eichenbaum and Hurst (2021).
- Acemoglu, D., S. Naidu, P. Restrepo, and J. A. Robinson (2019). Democracy Does Cause Growth. *Journal of Political Economy* 127(1), 47–100.
- Aghion, P., A. Alesina, and F. Trebbi (2007). Democracy, Technology, and Growth. Working Paper 13180, National Bureau of Economic Research.
- Allcott, H., L. Boxell, J. Conway, M. Gentzkow, M. Thaler, and D. Yang (2020). Polarization and Public Health: Partisan Differences in Social Distancing During the Coronavirus Pandemic. *Journal of Public Economics 191*, 104254.
- Anderson, K., G. Rausser, and J. Swinnen (2013). Political Economy of Public Policies: Insights from Distortions to Agricultural and Food Markets. *Journal of Economic Literature* 51(2), 423–77.
- Autor, D. H., D. Dorn, and G. H. Hanson (2016). The China Shock: Learning from Labor-market Adjustment to Large Changes in Trade. Annual Review of Economics 8, 205–240.
- Barro, R. (1997). Determinants of Economic Growth: A Cross-Country Empirical Study. MIT Press.

Barro and Sala-i-Martin (2004). Economic Growth (Second Edition). MIT Press. .

- Baum, M. A. and D. A. Lake (2003). The Political Economy of Growth: Democracy and Human Capital. *American Journal of Political Science* 47(2), 333–347.
- Besley, T. and A. Case (2003). Political Institutions and Policy Choices: Evidence from the United States. *Journal of Economic Literature 41*(1), 7–73.
- Besley, T. and M. Kudamatsu (2006). Health and Democracy. American Economic Review 96(2), 313-318.
- Burgess, R., R. Jedwab, E. Miguel, A. Morjaria, and G. Padró i Miquel (2015). The Value of Democracy: Evidence from Road Building in Kenya. American Economic Review 105(6), 1817–1851.
- Curtin, P. D. (1989). *Death By Migration: Europe's Encounter with the Tropical World in the Nineteenth Century*. Cambridge University Press.
- Doucouliagos, H. and M. A. Ulubaşoğlu (2008). Democracy and Economic Growth: A Meta-Analysis. *American Journal of Political Science* 52(1), 61–83.
- Easterly, W. (2019). In Search of Reforms for Growth: New Stylized Facts on Policy and Growth Outcomes. Working Paper 26318, National Bureau of Economic Research.
- Easterly, W. and R. Levine (2003). Tropics, Germs, and Crops: How Endowments Influence Economic Development. *Journal of Monetary Economics 50*(1), 3–39.

Eichenbaum, M. S. and E. Hurst (2021). NBER Macroeconomics Annual 2021, Volume 36. University of Chicago Press.

- Frey, C. B., C. Chen, and G. Presidente (2020). Democracy, Culture, and Contagion: Political Regimes and Countries Responsiveness to Covid-19. Covid Economics 18, 222–238.
- Gerring, J., S. C. Thacker, and R. Alfaro (2012). Democracy and Human Development. Journal of Politics 74(1), 1–17.
- Grossman, G., S. Kim, J. M. Rexer, and H. Thirumurthy (2020). Political Partisanship Influences Behavioral Responses to Governors Recommendations for Covid-19 Prevention in the United States. *Proceedings of the National Academy of Sciences 117*(39), 24144–24153.
- Guo, Z. (2023). Causal Inference with Invalid Instruments: Post-selection Problems and a Solution Using Searching and Sampling. *Journal* of the Royal Statistical Society Series B: Statistical Methodology 85(3), 959–985.
- Guo, Z., H. Kang, T. Tony Cai, and D. S. Small (2018). Confidence Intervals for Causal Effects with Invalid Instruments by Using Two-stage Hard Thresholding with Voting. *Journal of the Royal Statistical Society Series B: Statistical Methodology* 80(4), 793–815.
- Haggard, S. and R. Kaufman (2021). Backsliding: Democratic Regress in the Contemporary World. Cambridge University Press.
- Hall, R. E. and C. I. Jones (1999). Why Do Some Countries Produce So Much More Output Per Worker Than Others? *Quarterly Journal of Economics 114*(1), 83–116.
- Hayashi, F. (2000). Econometrics. Princeton University Press.
- Jorgenson, D. W. (2008). Productivity and Economic Growth. University of Chicago Press.

Kremer, M., J. Willis, and Y. You (2021). Converging to Convergence. In Eichenbaum and Hurst (2021).

- LaPorta, Rafael, F. L.-d.-S., A. Shleifer, and R. W. Vishny (1998). Law and Finance. Journal of Political Economy 106, 1113-55.
- Lee, D. S., J. McCrary, M. J. Moreira, and J. Porter (2022). Valid t-ratio Inference for IV. American Economic Review 112(10), 3260–3290.
- Martinez, L. R. (2022). How Much Should We Trust the Dictator's GDP Growth Estimates? Journal of Political Economy 130(10), 2731–2769.

Papaioannou, E. and G. Siourounis (2008). Democratisation and Growth. The Economic Journal 118(532), 1520–1551.

- Persson, T. and G. Tabellini (2006). Democracy and Development: The Devil in the Details. American Economic Review 96(2), 319-324.
- Przeworski, A., R. M. Alvarez, M. E. Alvarez, J. A. Cheibub, F. Limongi, and F. P. L. Neto (2000). Democracy and Development: Political Institutions and Well-being in the World, 1950-1990, Volume 3. Cambridge University Press.
- Przeworski, A. and F. Limongi (1993). Political Regimes and Economic Growth. Journal of Economic Perspectives 7(3), 51-69.
- Quinn, D. P. and J. T. Woolley (2001). Democracy and National Economic Performance: The Preference for Stability. American Journal of Political Science 45(3), 634–657.
- Rodrik, D. and R. Wacziarg (2005). Do Democratic Transitions Produce Bad Economic Outcomes? American Economic Review 95(2), 50-55.
- Schmelz, K. (2021). Enforcement May Crowd Out Voluntary Support for Covid-19 Policies, Especially Where Trust in Government Is Weak and in a Liberal Society. *Proceedings of the National Academy of Sciences 118*(1).
- Sokoloff, K. L. and S. L. Engerman (2000). Institutions, Factor Endowments, and Paths of Development in the New World. *Journal of Economic Perspectives* 14(3), 217–232.
- Song, Z., K. Storesletten, and F. Zilibotti (2011). Growing Like China. American Economic Review 101(1), 196-233.
- Tavares, J. and R. Wacziarg (2001). How Democracy Affects Growth. European Economic Review 45(8), 1341–1378.

### **A** Appendix

#### A.1 Extending Acemoglu et al. (2019)'s Analysis of the Effect of Democratization

Acemoglu et al. (2019) document that transitions to democracy cause per-capita GDP growth from 1960 to 2010. They use dynamic panel strategies that control for country fixed effects and utilize regional waves of democratization as an IV. While our finding that democracy has had a negative cross-country impact on growth since the beginning of the century neither denies nor corroborates Acemoglu et al. (2019)'s finding, it is interesting never-theless to explore whether their positive effect of democratization persists or weakens in recent years. We do so by extending their analysis to include 2010-2020. We examine how the impact of democratization on growth has evolved over time.

For the extended period of 1960 to 2020, we confirm that democratization positively impacts GDP per capita. Specifically, using Acemoglu et al. (2019)'s preferred model — which includes country fixed effects and four lags of GDP — our analysis in Column 3 in Table A24 (Panel (a), first line) suggests a 1.16% immediate increase in GDP per capita following a permanent shift to democracy. Over the long term, this effect is substantially larger, resulting in a 28.63% (s.e. = 7.87%) increase in GDP per capita, as detailed in the fourth line of Panel (a) in Table A24. In Panel (b), where we limit the analysis to the years 1980 to 2020, the long-term effect remains comparable at 27.03% (s.e. = 7.15%). This result is similar to Table 2 in Acemoglu et al. (2019)

For the more recent periods 1995-2020 and 2001-2020, however, the estimates are statistically insignificant and exhibit larger standard errors (Panels (c) and (d)). Notably, in Panel (d), the sign of the democratization coefficient reverses, suggesting a potential shift in the relationship in recent years. Table A25 provides additional robustness checks and finds a similar pattern to Table A24. Figure A7 presents corroborating evidence graphically. Overall, the impact of democratization on GDP per capita is inconclusive for 2001-2020.

#### A.2 Policy Mechanisms Behind Democracy's Effect during the Pandemic

Why does having a stronger democracy cause worse economic and public health outcomes during the Covid-19 pandemic? Media and policy discussions point to the speed, coverage, and severity of containment policies as potential proximate mechanisms. For example, Paul Krugman blames "*catastrophically slow and inadequate*" responses by the US government for its failure.<sup>20</sup> We explore whether this differential in policy responses explains democracy's negative effect. Our findings suggest that a key channel for the negative impact of democracy is weaker and narrower containment policies at the beginning of the outbreak. In contrast, the speed of containment policies appears to be less important.

To measure the severity of the policy, we use the Containment Health Index for the 10th confirmed case of Covid-19. The Containment Health Index is a measure of how strict the government's response to containing Covid-19 was (Blavatnik School of Government at the University of Oxford, 2021).<sup>21</sup> To quantify how widely

<sup>&</sup>lt;sup>20</sup>Krugman, Paul. 2020. "3 Rules for the Trump Pandemic." New York Times. March 19. https://www.nytimes.com/2020/03/19/ opinion/trump-coronavirus.html

<sup>&</sup>lt;sup>21</sup>We get similar results when we use the index at the 100th confirmed case or the index's mean during 2020.

initial responses cover aspects of civilian life, we look at the percentage of 13 domains in which the government introduced containment measures at the 10th Covid-19 case. The domains are schools, workplaces, public events, gatherings, public transport, stay-at-home requirements, domestic travel, international travel, public information campaigns, testing, contact tracing, facial coverings, and vaccinations. To assess policy speed, we consider the number of days between the 10th confirmed case and the introduction of any containment policy.<sup>22</sup>

For each policy response mechanism M (severity, coverage, or speed of containment response), we estimate the following 2SLS equations:

$$M_i = \alpha_2 + \beta_2 Democracy_i + \gamma_2 X_i + \varepsilon_{2i}$$
(5)

First Stage: 
$$Democracy_i = \alpha_1 + \beta_1 Z_i + \gamma_1 X_i + \varepsilon_{1i}$$
. (6)

Table A26 summarizes the results of this analysis.<sup>23</sup> Panel A shows that democracy causes less severe responses at the 10th confirmed case of Covid-19. The median estimate is that a standard deviation increase in democracy causes the Containment Health Index to decrease by 0.4 standard deviations, which corresponds to 20% of the mean. Democracy also narrows the scope of containment policies. The median estimate in Panel B suggests that a standard deviation increase in democracy causes a 9.3 percentage-point decrease in the coverage of the initial policy. On the other hand, democracy does not appear to cause slower responses. In fact, in Panel C, all columns predict that democracy causes faster responses. This leads to the bottom line that the severity and coverage of initial containment policies is a more important mechanism for the adverse effect of democracy than their speed.

<sup>&</sup>lt;sup>22</sup>We get similar results with the 100th confirmed case and January 1st, 2020 as the start date. The introduction date of any containment policy is the date when the Containment Health Index becomes positive. <sup>23</sup>We get similar results with alternative democracy indices, weighting, and sample definitions.

## A.3 Additional Results

	Variable	Data Source	Short Description			
Outcomes	Mean GDP Growth Rate	World Bank (2021c)	Annual percentage change in real GDP. It is calcu- lated using GDP and GDP deflator (base year varies by country) from the World Bank's WDI dataset.			
	Mean Nighttime Light In- tensity Growth Rate	Martinez (2022)	Nighttime light intensity as measured by satellite images			
Additional Out- comes	Mean Excess Deaths Per One Hundred Thousand	The Economist (2024)	Number of deaths per one hundred thousand be tween 2020/01/01 and 2022/12/31 in excess of the the baseline number of deaths we might normally			
	Life Satisfaction	Helliwell et al. (2023)	have expected in this period. Life Satisfaction (life ladder) measures the well being by asking respondents to imagine a ladde from 0 to 10, with 10 being their best possible lif and 0 their worst. Respondents then rate their cur rent life on this scale, capturing their overall life satisfaction.			
	Top 1% Income Share	World Inequality Database (2023)	Percentage of total income held by the wealthies			
	CO2 Emissions Per-Capita	Global Carbon Budget (2024)	1% of the population. Carbon dioxide (CO2) emissions from fossil fuels			
	Energy Consumption Per- Capita	US Energy Information Administra- tion (2023)	and industry, measured in tonnes per person. Energy consumption is expressed in kilowatt-hour per person. It refers to primary energy—that is the energy required by the end user for electricity transportation, and heating—plus the energy los due to inefficiencies during the conversion of raw resources into a usable form.			
Treatments	Democracy Index (V-Dem)	Coppedge et al. (2021)	Index aggregating indices measuring freedom o association, clean elections, freedom of expression			
	Democracy Index (Polity)	Center for Systemic Peace (2018)	elected officials, and suffrage. Index measuring the level of democracy by having the lower extreme being "hereditary monarchy"			
	Democracy Index (Freedom House)	Freedom House (2020)	and the upper "consolidated democracy". Index measuring the degree of democratic freedon by considering political rights and civil liberties.			
	Democracy Index (Economist Intelligence Unit)	Economist Intelligence Unit (2021)	Index measuring the state of democracy. Range from least democratic to most democratic.			
Weightings & Con- trols	Population (Millions)	United Nations Department of Eco- nomic and Social Affairs, Population Division (2019)				
	Absolute Latitude	Google Dataset Publishing Language (2021) World Bank (2021a)	Latitude of the centroid of each country (i.e., a mea sure of distance from the equator). Average monthly temperature in degrees Celcius.			
	Mean Temperature Mean Precipitation Population Density	World Bank (2021a) United Nations Department of Eco- nomic and Social Affairs, Population Division (2019)	Average monthly emperature in degrees of the second			
	Median Age	United Nations Department of Eco- nomic and Social Affairs, Population	United Nations projections of the median age of the population.			
	Diabetes Prevalence	Division (2019) International Diabetes Federation (2019)	% of population with diabetes aged 20 to 79.			
IVs	Log European Settler Mor- tality	Acemoglu et al. (2001)	Log annualized deaths per thousand mean strength of European settlers between the seventeenth and nineteenth century.			
	Log Population Density in 1500s	Acemoglu and Johnson (2005)	Log population density in the 1500s measured a the number of inhabitants per square kilometer.			
	British Legal Origin	LaPorta et al. (2008)	Binary indicator that the country's legal origin is British.			
	Fraction Speaking English	Hall and Jones (1999)	The fraction of the population speaking English as a mother tongue in 1992.			
	Fraction Speaking European	Hall and Jones (1999)	The fraction of the population speaking English French, German, Portuguese or Spanish as a mother tongue in 1992.			
	Bananas, Coffee, Maize, Millet, Rice, Sugarcane, Rubber, Wheat	Easterly and Levine (2003); Food and Agriculture Association of the United Nations (2020)	Binary indicator that the country produced any o the particular commodity in 1990.			
	Copper, Silver	Easterly and Levine (2003); World Bureau of Metal Statistics (2019a,b)	Binary indicator that the country mined any of the particular commodity in 1990.			
Mechanisms	Protectionism Growth	Fraser Institute (2021)	A measure of a country's aversion to trade internationally.			
	Populism Growth	Lührmann et al. (2020)	A measure of the extent to which members ar representatives of a political party use populi rhetoric.			
	Hate Speech Growth	Mechkova et al. (2021)	A measure of the extent to which hate speech is used in the rhetoric of major political parties.			

## Table A1: Data Sources and Description

	Variable	Data Source	Short Description
	Polarization Growth	Mechkova et al. (2021)	A measure of how significantly opinions differ or major political issues among major political parties
	Capital Investment Growth	World Bank (2021c)	in a society. Annual growth rate of gross capital formation ex- pressed as a % of the GDP. Gross capital forma- tion (formerly gross domestic investment) consists of outlays on additions to the fixed assets of the economy plus net changes in the level of inven- tories. Fixed assets include land improvements (fences, ditches, drains, and so on); plant, machin- ery, and equipment purchases; and the construction of roads, railways, and the like, including schools offices, hospitals, private residential dwellings, and commercial and industrial buildings. Inventories are stocks of goods held by firms to meet tem- porary or unexpected fluctuations in production on sales, and "work in progress." According to the 1993 SNA, net acquisitions of valuables are also
	R&D Expenditure Growth	UNESCO Institute for Statistics (2021)	considered capital formation. Annual growth rate of gross domestic expenditures on R&D, expressed as a percent of GDP. They in clude both capital and current expenditures in the four main sectors: business enterprise, government higher education and private non-profit. R&D cov- ers basic research, applied research, and experi-
	Labor Force Growth	World Bank (2021c)	mental development. Annual growth rate of the labor force, which com prises people ages 15 and older who supply labo for the production of goods and services during a specified period. It includes people currently em ployed and people unemployed but seeking worl as well as first-time job-seekers. Not everyom who works is included, however. Unpaid work ers, family workers, and students are often omitted Some countries do not count members of the armer
	TFP Growth	Feenstra et al. (2015)	forces. Annual growth rate of total factor productivity cal culated using real GDP and factor input growth
	Import Value Growth	World Bank (2021c)	rates obtained from national accounts data. Annual growth rate of the imports of goods and ser vices (BoP, in current US\$).
	Export Value Growth	World Bank (2021c)	Annual growth rate of the exports of goods and services (BoP, in current US\$).
	Population Growth	United Nations, World Population Prospects (2024b)	Annual growth rate of population.
	Median Age Growth	United Nations, World Population Prospects (2024a)	Annual growth rate of median age.
	Primary Education Growth	World Bank (2021c)	Annual growth rate of the ratio of children of of ficial school age enrolled in primary school to th total population of children in the corresponding of facial school age group
	Secondary Education Growth	World Bank (2021c)	ficial school age group. Annual growth rate of the ratio of children of offi cial school age enrolled in secondary school to th total population of children in the corresponding of ficial school age group.
Additional Mecha- nisms	Value Added, Agriculture Growth	World Bank (2021c)	Annual growth rate for agricultural value added ex pressed in current US dollars. Agriculture corre sponds to ISIC divisions 1-5 and includes forestry hunting, and fishing, as well as cultivation of crop and livestock production. Value added is the ne output of a sector after adding up all outputs an subtracting intermediate inputs. It is calculated without making deductions for depreciation of fab ricated assets or depletion and degradation of nat ural resources. The origin of value added is de termined by the International Standard Industria
	Value Added, Manufactur- ing Growth	World Bank (2021c)	Classification (ISIC), revision 3. Annual growth rate for manufacturing value adde expressed in current US dollars. Manufacturin refers to industries belonging to ISIC divisions 15 37.

## Table A1: Data Sources and Description

	Variable	Data Source	Short Description
	Value Added, Services Growth	World Bank (2025b)	Annual growth rate for value added in services ex- pressed in current US dollars. Services correspond to ISIC divisions 50-99. They include value added in wholesale and retail trade (including hotels and restaurants), transport, and government, financial, professional, and personal services such as educa- tion, health care, and real estate services. Also in- cluded are imputed bank service charges, import duties, and any statistical discrepancies noted by national compilers as well as discrepancies arising
	Tex Revenue Growth	UNU-WIDER Government Revenue Dataset (2023)	from rescaling. Annual growth rate of the ratio of tax revenues to GDP. It includes both direct and indirect taxes, as well as social contributions. Taxes are defined as compulsory, unrequited payments to the govern- ment, while social contributions encompass both compulsory and voluntary social insurance contri- butions from employers, employees, and the self- amplexed.
	R&D Researchers Growth	UNESCO Institute for Statistics (2021)	employed. Annual growth rate of the number of researchers engaged in R&D per million. Researchers are pro- fessionals who conduct research and improve or de- velop concepts, theories, models techniques instru-
	New Business Registrations Growth	World Bank (2021b)	mentation, software of operational methods. Annual growth rate of new businesses per 1,000 people. New businesses registered are the number of new limited liability corporations registered in the calendar year. For cross-country comparabil- ity, only limited liability corporations that operate
	FDI Growth	World Bank (2025a)	in the formal sector are included. Annual growth rate of foreign direct investments (FDI). FDI are defined as the net inflows of invest- ment used to acquire a lasting management interest (10 percent or more of the voting stock) in an enter- prise operating in a foreign economy. It comprises equity capital, reinvested earnings, other long-term capital, and short-term capital, as recorded in the balance of payments. The series represents the net inflows from foreign investors in the reporting
	Conflict Index Growth	Banks (2024)	economy, as a share of GDP. Weighted average of indices measuring assassina- tions, strikes, guerilla warfare, government crises, purges, riots, revolutions, and anti-government
	Child Mortality Growth	World Bank (2021c)	demonstrations. Annual growth rate of the number of infants who die before one year of age, per 1,000 live births.
Policy Re- sponses in 2020	Containment Health Index at 10th Covid-19 Case	Blavatnik School of Government at the University of Oxford (2021)	Average of 13 sub-scores which record severity of government responses in a specific domain on ar ordinal scale (for example, the school sub-index is on a 0 (no measure) to 4 (require closing) scale) and subtracts 0.5 if it is targeted. It is scaled to take a value between 0 and 100. The domains are schools workplaces, public events, gatherings, public trans- port, stay-at-home requirements, domestic travel international travel, public information campaigns testing, contact tracing, facial coverings, and vacci- nations. We use the index at the date when the 10th case of Covid-19 is confirmed.
	Coverage of Containment Measures at 10th Covid-19 Case	Blavatnik School of Government at the University of Oxford (2021)	The percentage of the 13 domains in which the dat records any policy introduction at the date when th 10th case of Covid-19 is confirmed.
	Days between 10th Covid- 19 Case and Any Contain- ment Measure	Blavatnik School of Government at the University of Oxford (2021)	The number of days between the date when the 10t Covid-19 case is confirmed and the date when th containment health index becomes positive.

## Table A1: Data Sources and Description

## Table A2: Descriptive Statistics

	Variable	Ν	Mean	St. Dev.	Min	Median	Max
Outcomes	Mean GDP Growth Rate in 2001-2019	160	3.8	1.8	0.1	3.6 (Chile)	8.7 (China)
	Mean GDP Growth Rate in 2020-2022	160	1.9	4.1	(Greece) -7.3	1.7	(China) 44.0
	Mean Nighttime Light Intensity Growth Rate in	156	6.8	6.0	(Ukraine) 0.8	(Burundi) 5.5	(Guyana) 48.1
	2001-2013				(United Kingdom)	(Bolivia)	(Cambodia)
Freatments	Democracy Index (V-Dem, 2000)	160	0	1.0	-1.8	-0.01	1.4
	Democracy Index (V-Dem, 2019)	160	0	1.0	(Saudi Arabia) -2.2	(Madagascar) 0.02	(Sweden) 1.5
	• • • •				(Saudi Arabia)	(Bhutan)	(Denmark)
Controls	GDP (Current USD, Billions, 2000)	160	206.7	931.2	0.1	11.7	10,251.0
	GDP (Current USD, Billions, 2019)	160	532.3	2,116.4	(Sao Tome and Principe) $0.4$	(El Salvador) 54.6	(United States 21,381.0
	Absolute Latitude	160	26.4	17.5	(Sao Tome and Principe) 0.0	(Slovenia) 23.5	(United States 65.0
	Mean Temperature (°c, 1991-2000)	160	18.5	8.4	(Democratic Republic of the Congo) -6.2	(Bangladesh) 22.0	(Iceland) 28.6
	•				(Canada)	(Angola)	(Mali)
	Mean Temperature (°c, 1991-2016)	160	18.7	8.3	-6.0 (Canada)	22.3 (Botswana)	28.9 (Mali)
	Mean Precipitation (mm per Month, 1991-2000)	160	92.1	63.7	2.7 (Egypt)	78.6 (Albania)	252.7 (Malaysia)
	Mean Precipitation (mm per Month, 1991-2016)	160	93.4	64.6	2.5	79.8	259.1
	Population Density (No. of People per km <sup>2</sup> , 2000)	160	155.3	481.2	(Egypt) 1.5	(Angola) 61.7	(Malaysia) 5,755.5
	Population Density (No. of People per km <sup>2</sup> , 2019)	160	213.3	701.0	(Mongolia) 2.1	(Benin) 81.5	(Singapore) 8,291.9
	Median Age (2000)	160	213.5	8.0	(Mongolia) 15.0	(Greece) 22.7	(Singapore) 41.2
					(Burundi)	(Guyana)	(Japan)
	Median Age (2019)	160	30.5	9.3	15.2 (Niger)	29.6 (Lebanon)	48.4 (Japan)
	Diabetes Prevalence (%, 2019)	160	7.5	4.0	1.0 (Benin)	6.4 (Cambodia)	22.1 (Sudan)
Vs	Log European Settler Mortality (Annual No. of Deaths per Thousand)	82	4.6	1.3	0.9	4.5	8.0
	Log Population Density in 1500s (No. of Inhabi- tants per km <sup>2</sup> )	87	0.6	1.6	(United Kingdom) -3.8	(Barbados) 0.4	(Mali) 4.6
	British Legal Origin	91	0.4	0.5	(Canada) 0	(Costa Rica) 0	(Egypt)
	0 0				(Algeria)	(Algeria)	(Australia) 1
	Fraction Speaking English	133	0.1	0.2	(Algeria) 0	(Algeria) 0	(Barbados)
	Fraction Speaking European Bananas	133 139	0.2	0.4 0.5	(Angola)	(Angola)	1 (France)
			0.7		0 (Albania)	(Angola)	l (Angola)
	Coffee	139	0.5	0.5	0 (Albania)	0 (Albania)	l (Angola)
	Copper	147	0.3	0.5	0 (Algeria)	0 (Algeria)	1 (Albania)
	Maize	139	0.9	0.3	0 (Bahrain)	1 (Albania)	1 (Albania)
	Millet	139	0.5	0.5	0 (Albania)	0 (Albania)	1 (Angola)
	Rice	139	0.7	0.5	0 (Austria)	(Albania)	(Albania)
	Rubber	139	0.2	0.4	(Albania) (Albania)	(Albania) (Albania)	(Bangladesh)
	Silver	145	0.4	0.5	0	0	1
	Sugarcane	139	0.6	0.5	(Albania) 0 (Albania)	(Albania) 1 (Angala)	(Algeria) 1
					(Albania)	(Angola)	(Angola)

Table A2 (	(Continued)
------------	-------------

	Variable	Ν	Mean	St. Dev.	Min	Median	Max
Mechanisms							
in 2001-2019	Capital Investments (Mean Annual % Growth)	125	1.7	3.2	-6.1	0.8	14.3
	-				(Equatorial Guinea)	(South Africa)	(Ivory Coast)
	R&D Expenditure (Mean Annual % Growth)	125	5.7	13.1	-7.9	1.9	74.0
					(Madagascar)	(Spain)	(Bosnia and Herzegovina)
	Total Labor Force (Mean Annual % Growth)	159	1.9	1.6	-1.4	2.0	10.4
					(Romania)	(Colombia)	(Qatar)
	TFP (Mean Annual % Growth)	113	0.8	1.6	-4.7	0.5	8.0
					(Qatar)	(Kenya)	(Tajikistan)
	Total Import Value (Mean Annual % Growth)	115	9.6	3.7	4.3	9.1	33.0
					(Jamaica)	(Colombia)	(Djibouti)
	Total Export Value (Mean Annual % Growth)	115	9.8	4.6	3.7	8.8	41.3
					(Jamaica)	(Paraguay)	(Djibouti)
	Median Age (Mean Annual % Growth)	154	1.6	1.4	-1.6	1.4	8.2
					(Moldova)	(Ireland)	(Qatar)
	Population (Mean Annual % Growth)	154	0.9	0.5	-0.6	0.8	2.5
					(Central African Republic)	(Barbados)	(Maldives)
	Primary School Enrollment (Mean Annual % Growth)	146	0.5	1.1	-2.5	0.1	4.8
					(Equatorial Guinea)	(Mauritius)	(Djibouti)
	Secondary School Enrollment (Mean Annual % Growth)	138	1.7	2.2	-1.5	0.9	11.1
	,				(Liberia)	(Mali)	(Mozambique)

Notes: Parentheses contain country names corresponding to the minimum, median and maximum values of each variable. When we observe multiple countries corresponding to the same minimum, median or maximum, we choose the first country in alphabetical order. When we do not find a country that corresponds exactly to the median, we choose the country with the closest value. Variable definitions and data sources are in Appendix Table A1.

## Table A3: Additional Descriptive Statistics

	Variable	Ν	Mean	St. Dev.	Min	Median	Max
Outcomes	Mean GDP Growth Rate in 1981-1990	130	2.9	2.3	-2.5	2.6	9.9
	Mean GDP Growth Rate in 1991-2000	155	3.7	4.2	(Guyana) -1.8	(Jamaica) 3.1	(Botswana) 38.4
	Mean GDP Growth Rate in 2001-2010	159	4.1	2.5	(Moldova) -3.0	(New Zealand) 3.8	(Equatorial Guinea) 18.0
	Mean GDP Growth Rate in 2011-2019	160	3.3	1.9	(Zimbabwe) -3.0	(Benin) 3.3	(Equatorial Guinea) 8.5
	Mean GDP Per Capita Growth Rate in 2001-2019		2.4	1.9	(Equatorial Guinea) -1.5	(Egypt) 2.1	(Ethiopia) 8.7
	1	160			(United Arab Emirates)	(Libya)	(Burma)
	Mean GDP Per Capita Growth Rate in 2020-2022	160	0.8	4.2	-12.8 (Lebanon)	0.9 (Burkina Faso)	41.7 (Guyana)
	Mean Excess Deaths per 100k People 2020-2022	154	87.9	64.7	-33.8 (Bhutan)	71.7 (Kenya)	363.9 (Bulgaria)
	Life Satisfaction Growth Rate 2010-2019	144	0.5	2.5	-11.8 (Angola)	0.5 (Croatia)	12.2 (Gambia)
	Life Satisfaction Growth Rate 2020-2022	129	-0.9	5.4	-25.6 (Democratic Republic of the Congo)	-0.4 (Iceland)	13.8 (Mauritania)
	Top 1% Income Share Growth Rate in 2001-2019	158	-0.1	1.2	-3.1	-0.01	6.1
	Top 1% Income Share Growth Rate in 2020-2022	158	-0.4	3.1	(Maldives) -16.3	(Nigeria) 0.0	(Cyprus) 16.0
	CO2 Emissions Growth Rate in 2001-2019	154	1.9	3.2	(Costa Rica) -3.4	(Albania) 1.8	(Ivory Coast) 17.8
	CO2 Emissions Growth Rate in 2020-2022	154	-0.5	4.1	(Gabon) -10.0	(Latvia) -0.9	(Laos) 19.1
	Energy Consumption Growth Rate in 2001-2019	154	2.2	6.5	(Ukraine) -2.2	(Egypt) 1.5	(Singapore) 76.1
					(Zimbabwe)	(Belarus)	(Equatorial Guinea)
	Energy Consumption Growth Rate in 2020-2022	154	-1.6	4.0	-16.7 (Papua New Guinea)	-1.4 (Comoros)	11.3 (Albania)
ontrols	GDP (Current USD, Billions, 1980)	128	85.7	293.4	0.03	7.4	2,857.3
	GDP (Current USD, Billions, 1990)	135	163.5	619.2	(Equatorial Guinea) 0.1	(Ethiopia) 8.0	(United States) 5,963.1
	GDP (Current USD, Billions, 2000)	160	206.7	931.2	(Sao Tome and Principe) 0.1	(Dominican Republic) 11.7	(United States) 10,251.0
	GDP (Current USD, Billions, 2010)	160	402.8	1,427.1	(Sao Tome and Principe) 0.2	(El Salvador) 37.4	(United States) 15,049.0
	GDP Per Capita (Current USD, 1980)	124	4,241.3	6,214.0	(Sao Tome and Principe) 123.4	(Costa Rica) 1,431.8	(United States) 41,311.9
	• • • • •				(Equatorial Guinea)	(Tunisia)	(United Arab Emirate
	GDP Per Capita (Current USD, 1990)	133	5,848.5	8,769.9	87.2 (Sudan)	1,310.4 (Republic of the Congo)	39,842.8 (Switzerland)
	GDP Per Capita (Current USD, 2000)	159	6,655.8	10,252.9	130.2 (Burundi)	1,675.8 (Paraguay)	48,984.2 (Luxembourg)
	GDP Per Capita (Current USD, 2010)	160	13,182.9	18,936.0	231.5 (Burundi)	4,604.7 (Ecuador)	112,049.1 (Luxembourg)
	GDP Per Capita (Current USD, 2019)	160	14,861.5	20,346.0	261.3 (Burundi)	5,906.3 (Bosnia and Herzegovina)	113,860.5 (Luxembourg)
	Population (Millions, 2000)	159	37.1	135.4	0.1	8.1	1,290.6
	Population (Millions, 2019)	159	47.0	162.4	(Seychelles) 0.1	(Azerbaijan) 10.1	(China) 1,439.3
	Mean Temperature (°c, 1971-1980)	160	17.9	8.5	(Seychelles) -7.4	(Azerbaijan) 21.4	(China) 28.2
	Mean Temperature (°c, 1981-1990)	160	18.2	8.5	(Canada) -7.0	(Angola) 21.8	(Mali) 28.6
	Mean Temperature (°c, 1991-2000)	160	18.5	8.4	(Canada) -6.2	(Angola) 22.0	(Mali) 28.6
	Mean Temperature (°c, 2001-2010)	160	18.9	8.4	(Canada) -5.8	(Angola) 22.5	(Mali) 29.1
					(Canada)	(Botswana)	(Mali)
	Mean Precipitation (mm per Month, 1971-1980)	160	93.4	64.5	3.0 (Egypt)	83.8 (Angola)	260.3 (Costa Rica)
	Mean Precipitation (mm per Month, 1981-1990)	160	92.1	64.0	3.1 (Egypt)	79.8 (Albania)	256.5 (Papua New Guinea
	Mean Precipitation (mm per Month, 1991-2000)	160	92.1	63.7	2.7 (Egypt)	78.6 (Albania)	252.7 (Malaysia)
	Mean Precipitation (mm per Month, 2001-2010)	160	94.9	66.3	2.2	82.5	265.7
	Population Density (No. of People per km <sup>2</sup> , 1980)	160	110.7	297.8	(Egypt) 1.1	(Angola) 42.6	(Malaysia) 3,445.3
	Population Density (No. of People per km <sup>2</sup> , 1990)	160	131.4	367.2	(Mongolia) 1.4	(Malaysia) 50.9	(Singapore) 4,304.2
	Population Density (No. of People Per km <sup>2</sup> , 2000)	160	155.3	481.2	(Mongolia) 1.5	(Cambodia) 61.7	(Singapore) 5,755.5
	Population Density (No. of People Per km <sup>2</sup> , 2010)	160	186.1	612.3	(Mongolia) 1.8	(Benin) 73.3	(Singapore) 7,330.2
					(Mongolia)	(Bosnia and Herzegovina)	(Singapore)
	Median Age (1980)	160	22.4	6.3	15.0 (Kenya)	19.3 (Haiti)	36.5 (Germany)
	Median Age (1990)	160	23.7	7.2	15.3 (Rwanda)	20.8 (Lebanon)	38.4 (Sweden)
	Median Age (2000)	160	25.7	8.0	15.0 (Burundi)	22.7 (Guyana)	41.2 (Japan)
	Median Age (2010)	160	28.1	8.7	15.0	26.2	44.7

	Variable	Ν	Mean	St. Dev.	Min	Median	Max
Mechanisms							
in 2001–2019	Value Added Agriculture (Mean Annual % Growth)	135	6.9	3.5	0.5	6.8	18.9
					(Sudan)	(Cameroon)	(Chad)
	Value Added Manufacturing (Mean Annual % Growth)	124	8.4	6.0	1.2	6.9	43.7
					(United Kingdom)	(Egypt)	(Gabon)
	Value Added Services (Mean Annual % Growth)	138	9.1	3.6	1.4	8.7	23.1
					(Japan)	(Cape Verde)	(Ghana)
	Tax Revenue Share (Mean Annual % Growth)	125	1.4	2.2	-2.6	1.0	16.4
					(Angola)	(Cape Verde)	(Burma)
	R&D Researchers (Mean Annual % Growth)	118	5.0	11.3	-7.9	1.9	74.0
					(Mauritius)	(Bulgaria)	(Brazil)
	No. of New Business Registrations (Mean Annual % Growth)	124	5.7	12.1	-7.9	2.0	74.0
					(Luxembourg)	(Paraguay)	(Bahrain)
	Foreign Direct Investments (Mean Annual % Growth)	99	5.5	395.5	-3,205.0	31.4	890.5
					(Belarus)	(Cyprus)	(Paraguay)
	Value of Imports from China (Mean Annual % Growth)	134	23.2	9.9	6.7	21.2	56.7
					(Japan)	(Egypt)	(Armenia)
	Value of Exports to China (Mean Annual % Growth)	104	39.9	33.7	7.8	26.6	149.1
			0.1	0.0	(Finland)	(Barbados)	(Jamaica)
	Conflict Index (Mean Annual % Growth)	155	0.1	0.8	-0.1	-0.1	7.7
		1.50			(Laos)	(Senegal)	(Iraq)
	Child Mortality Rate (Mean Annual % Growth)	153	-3.4	1.8	-9.5	-3.1	0.7
					(Montenegro)	(Burma)	(Botswana)
Policy Responses							
in 2020	Containment Health Index at 10th Covid-19 Case	151	1.8	1.0	0.0	1.7	3.9
					(Algeria)	(Burkina Faso)	(Djibouti)
	Coverage of Containment Policy at 10th Covid-19 Case	152	48.4	23.7	0.0	46.2	92.3
					(Algeria)	(Azerbaijan)	(Bhutan)
	Days Between 10th Covid-19 Case Until Any Containment Measure	152	-42.8	33.3	-270	-40	34
					(Solomon Islands)	(Azerbaijan)	(Thailand)

### Table A3 (Continued)

Notes: Parentheses contain country names corresponding to the minimum, median, and maximum values of each variable. When we observe multiple countries corresponding to the same minimum, median, or maximum, we choose the first country in alphabetical order. When we do not find a country that corresponds exactly to the median, we choose the country with the closest value. Variable definitions and data sources are in Appendix Table A1.

	Democracy Index		Democracy Index	
	(V-Dem, 2000)	Rank	(V-Dem, 2019)	Rank
United States	1.3	22	1.1	32
China	-1.6	150	-1.9	158
Japan	1.2	35	1.1	30
Germany	1.4	2	1.4	11
United Kingdom	1.3	26	1.3	17
India	0.8	50	-0.3	94
France	1.3	16	1.3	13
Italy	1.2	30	1.3	21
Brazil	1.3	21	0.6	55
Canada	1.2	33	1.2	24
Russia	-0.4	96	-1.2	133
South Korea	1.2	31	1.3	23
Spain	1.4	5	1.3	14
Australia	1.4	9	1.2	26
Mexico	0.4	62	0.5	59
Indonesia	0.6	56	0.3	70
Netherlands	1.3	27	1.3	16
Saudi Arabia	-1.8	159	-2.2	159
Switzerland	1.4	7	1.4	4
Poland	1.3	17	0.6	58
Thailand	0.0	79	-1.4	144
Belgium	1.3	14	1.4	8
Sweden	1.4	1	1.4	2
Argentina	1.2	34	0.9	40
Nigeria	-0.2	87	-0.1	85
Austria	1.3	25	1.2	25
United Arab Emirates	-1.8	157	-1.8	156
Norway	1.4	6	1.4	6
Israel	0.9	43	0.6	54
Ireland	1.3	11	1.3	12

Table A4: Democracy Indices for 30 Countries with Largest Total GDP in 2019

*Notes:* This table reports the democracy index in 2000 and 2019 and the corresponding rank in the dataset (N=159) for 30 countries with the largest total GDP in 2019. The countries are ordered by GDP size. The Democracy Index (V-Dem) is normalized to have mean zero and standard deviation one in each year.

	V-Dem	Polity	Freedom	Economist
			House	Intelligence
				Unit
Panel A: Democracy Index for 2019				
V-Dem (2019)	1			
Polity (2018)	0.858	1		
Freedom House (2019)	0.944	0.838	1	
Economist Intelligence Unit (2019)	0.891	0.776	0.946	1
Panel B: Democracy Index for 2000				
V-Dem (2000)	1			
Polity (2000)	0.898	1		
Freedom House (2003)	0.935	0.889	1	
Economist Intelligence Unit (2006)	0.909	0.853	0.917	1

### Table A5: Correlation Among Democracy Indices

*Notes:* This table reports the pairwise correlations among the V-Dem, Polity, Freedom House, and Economist Intelligence Unit's democracy indices in 2019 (Panel A) and 2000 (Panel B). The publication year of each index is in parentheses. When data for democracy levels in 2019 or 2000 are unavailable, we use the index from the nearest available year. Variable definitions and data sources are in Appendix Table A1.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)		
Dependent Variable is Mean GDP Growth Rate in 2001-2019										
Democracy Index (V-Dem, 2000)	-1.6	-1.2	-1.3	-0.9	-1.7	-1.0	-1.1	-1.0		
	(0.4)	(0.6)	(0.5)	(0.4)	(0.4)	(0.5)	(0.4)	(0.4)		
Dependent Variable is Mean GDP Growth Rate in 2020-2022										
Democracy Index (V-Dem, 2019)	-0.8	-0.7	-1.0	-0.9	-0.8	-0.8	-0.8	-0.9		
	(0.2)	(0.2)	(0.2)	(0.3)	(0.2)	(0.2)	(0.2)	(0.3)		
Dependent Variable is Mean Nighttime Light Intensity Growth Rate in 2001-2013										
Democracy Index (V-Dem, 2000)	-1.6	-1.5	-0.8	-1.0	-1.3	-0.9	-0.8	-0.9		
	(0.3)	(0.4)	(0.3)	(0.2)	(0.3)	(0.3)	(0.3)	(0.3)		
Baseline Controls Other Than Baseline GDP		$\checkmark$		$\checkmark$		$\checkmark$		$\checkmark$		
Baseline GDP Per Capita Control			$\checkmark$	$\checkmark$			$\checkmark$	$\checkmark$		
Baseline Total GDP Control					$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$		
Ν	150	150	150	150	150	150	150	150		

Table A6: Correlation Between Democracy and Economic Growth With Control for Baseline GDP

*Notes:* This table reports the results of OLS regressions of GDP and nightime light intensity growth rates on the democracy index with additional controls. The Democracy Index (V-Dem) is normalized to have mean zero and standard deviation one in each year. Columns 2, 4, 6, 8 have the following baseline controls: absolute latitude, mean temperature, mean precipitation, population density, and median age. For the mean GDP growth rate in 2020-2022, we also control for diabetes prevalence. Columns 3, 4, 7, and 8 additionally control for baseline GDP per capita. Columns 5, 6, 7, 8 additionally control for baseline total GDP. N refers to the number of countries for which IVs, controls, and outcomes are available. Robust standard errors are in parentheses. Variable definitions and data sources are in Appendix Table A1.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
			Dependen	t Variable	is Demo	cracy In	dex (V-D	em, 201	9)			
Log European Settler Mortality	-0.8	-0.9									0.09	-0.3
	(0.2)	(0.3)									(0.2)	(0.1)
Log Population Density in 1500s			-0.5	-0.5							0.008	-0.004
			(0.09)	(0.04)							(0.08)	(0.06)
British Legal Origin					2.0	2.0					1.3	0.7
					(0.6)	(0.5)					(0.3)	(0.2)
Fraction Speaking English							-0.05	0.7			-1.0	-0.4
							(0.2)	(0.5)			(0.5)	(0.5)
Fraction Speaking European							1.8	1.2			2.3	2.1
							(0.6)	(0.3)			(0.5)	(0.3)
Bananas									-0.08	0.3	-0.7	-0.6
									(0.5)	(0.4)	(0.2)	(0.2)
Coffee									-0.08	0.8	-0.3	0.002
									(0.3)	(0.3)	(0.2)	(0.2)
Copper									-0.6	-0.01	-0.06	0.5
									(0.4)	(0.4)	(0.3)	(0.4)
Maize									0.7	1.2	1.0	-2.6
									(0.4)	(0.4)	(0.8)	(0.4)
Millet									-0.6	-0.3	-0.2	-0.2
									(0.4)	(0.3)	(0.3)	(0.2)
Rice									-0.8	-0.8	-0.07	0.5
									(0.6)	(0.5)	(0.4)	(0.3)
Rubber									-2.2	-2.2	-0.4	0.03
									(0.5)	(0.3)	(0.2)	(0.3)
Silver									1.1	0.6	0.6	0.1
								(0.4)	(0.4)	(0.5)	(0.4)	
Sugarcane									1.0	0.4	1.0	0.4
6								(0.6)	(0.6)	(0.4)	(0.4)	
Wheat									-0.4	0.9	-1.1	-1.0
									(0.5)	(0.6)	(0.5)	(0.3)
F-Statistic (First stage)	9.7	7.3	27.2	135.2	12.0	16.8	4.3	14.0	6.4	6.0	57.5	351.8
Baseline Controls	x	1	X	1	x	1	×	1	x	1	X	1
N	81	81	86	86	90	90	132	132	138	138	71	71

Table A7: First-stage Regression Estimates of IVs' Effects on Democracy

*Notes:* This table reports the first-stage regression estimates of the effect of the five different sets of IVs on democracy levels in 2019. The Democracy Index (V-Dem) is normalized to have mean zero and standard deviation one in each year. Columns 1, 3, 5, 7, 9, and 11 have no controls, while columns 2, 4, 6, 8, 10, and 12 have the following baseline controls: absolute latitude, mean temperature, mean precipitation, population density, median age, and diabetes prevalence. N refers to the number of countries for which IVs, controls, and outcomes are available. Robust standard errors are in parentheses. Variable definitions and data sources are in Appendix Table A1.
Panel A: Africa & Asia vs. Amer	(1) icas & Oc	(2) eania	(3)	(4)	(5)	(6)
i undi iti itiridu tu itisiu (si itinici			le is Demo	cracy Inde	ex (V-Dem	, 2019)
Log European Settler Mortality	-0.2 (0.3)	-0.3 (0.3)				
Log Population Density in 1500s			-0.3 (0.2)	-0.2 (0.1)		
British Legal Origin					1.4 (0.3)	0.5 (0.2)
Ν	49 Depend	49 ant Variab	58 le is Demo	58 cracy Inde	61	61
Log European Settler Mortality	-0.3	-0.04	le 13 Denio	cracy mad	X (V-Dem	, 2017)
Log Population Density in 1500s	(0.04)	(0.08)	-0.2	0.1		
British Legal Origin			(0.02)	(0.04)	0.5	0.03
N	29	29	28	28	(0.06) 29	(0.1)
Baseline Controls Panel B: Africa & Oceania vs. An	X moricoc 8	✓ A sig	×	1	X	1
ranei D. An ica & Oceania vs. A			le is Demo	cracy Inde	ex (V-Dem	, 2019)
Log European Settler Mortality	-0.3 (0.1)	-0.4 (0.3)				
Log Population Density in 1500s	(0.1)	(0.5)	-0.3 (0.02)	-0.3 (0.07)		
British Legal Origin			(0.02)	(0.07)	1.6 (0.3)	1.0 (0.3)
N	40	40	47	47	50	50
Log European Settler Mortality	-1.1	-1.2	le is Demo	cracy inde	ex (v-Dem	, 2019)
Log Population Density in 1500s	(0.3)	(0.3)	-0.5	-0.6		
British Legal Origin			(0.10)	(0.05)	2.0	2.1
					(0.6)	(0.8)
N Baseline Controls	38 ×	38 ✓	39 X	39 ✓	40 ×	40 ✓
Panel C: Africa & S. America vs.	N. Amer			a		
	Depend	ent Variab.	le 1s Demo	cracy Inde	ex (V-Dem	, 2019)
Log European Settler Mortality	-0.2	0.02				
Log European Settler Mortality Log Population Density in 1500s	-0.2 (0.07)	0.02 (0.2)	-0.3	-0.3		
				-0.3 (0.06)	-0.2	0.9
Log Population Density in 1500s British Legal Origin	(0.07)	(0.2)	-0.3 (0.04)	(0.06)	(0.3)	(0.2)
Log Population Density in 1500s	(0.07)	(0.2)	-0.3	(0.06)	(0.3) 57	(0.2)
Log Population Density in 1500s British Legal Origin	(0.07) 47 Depende -1.1	(0.2) 47 ent Variable -0.9	-0.3 (0.04) 55	(0.06)	(0.3) 57	(0.2)
Log Population Density in 1500s British Legal Origin N	(0.07) 47 Depende	(0.2) 47 ent Variab	-0.3 (0.04) 55 le is Demo -0.5	(0.06) 55 cracy Inde -0.5	(0.3) 57	(0.2)
Log Population Density in 1500s British Legal Origin N Log European Settler Mortality	(0.07) 47 Depende -1.1	(0.2) 47 ent Variable -0.9	-0.3 (0.04) 55 le is Demo	(0.06) 55 cracy Inde	(0.3) 57	(0.2) 57
Log Population Density in 1500s British Legal Origin N Log European Settler Mortality Log Population Density in 1500s	(0.07) 47 Depende -1.1 (0.3)	(0.2) 47 ent Variable -0.9 (0.3)	-0.3 (0.04) 55 le is Demo -0.5 (0.09)	(0.06) 55 cracy Inde -0.5 (0.06)	(0.3) 57 ex (V-Dem 2.4 (0.4)	(0.2) 57 , 2019) 1.8 (1.0)
Log Population Density in 1500s British Legal Origin N Log European Settler Mortality Log Population Density in 1500s British Legal Origin N Baseline Controls	(0.07) 47 Depende -1.1 (0.3) 31 <b>X</b>	$(0.2)$ $47$ ent Variab $-0.9$ $(0.3)$ $31$ $\checkmark$	-0.3 (0.04) 55 le is Demo -0.5 (0.09) 31 X	(0.06) 55 cracy Inde -0.5 (0.06) 31 ✓	(0.3) 57 ex (V-Dem 2.4	(0.2) 57 , 2019) 1.8 (1.0)
Log Population Density in 1500s British Legal Origin N Log European Settler Mortality Log Population Density in 1500s British Legal Origin N	(0.07) 47 Depende -1.1 (0.3) 31 X . S. Ameri	(0.2) $47$ ent Variabi -0.9 (0.3) $31$ $\checkmark$ ica, Asia,	$ \begin{array}{r}     -0.3 \\     (0.04) \\ \hline     55 \\     le is Demo \\     -0.5 \\     (0.09) \\ \hline     31 \\     \times \\     & Oceania \\ \end{array} $	$(0.06)$ $\overline{55}$ cracy Inde $-0.5$ $(0.06)$ $\overline{31}$ $\checkmark$ a	(0.3) 57 ex (V-Dem 2.4 (0.4) 33 <b>x</b>	(0.2) 57 , 2019) 1.8 (1.0) 33 ✓
Log Population Density in 1500s British Legal Origin N Log European Settler Mortality Log Population Density in 1500s British Legal Origin N Baseline Controls	(0.07) 47 Depende -1.1 (0.3) 31 <b>X</b> <b>. S. Ameri</b> Depende -0.4	$(0.2)$ $47$ ent Variab $-0.9$ $(0.3)$ $31$ $\checkmark$ ica, Asia, ent Variab $0.08$	-0.3 (0.04) 55 le is Demo -0.5 (0.09) 31 X	$(0.06)$ $\overline{55}$ cracy Inde $-0.5$ $(0.06)$ $\overline{31}$ $\checkmark$ a	(0.3) 57 ex (V-Dem 2.4 (0.4) 33 <b>x</b>	(0.2) 57 , 2019) 1.8 (1.0) 33 ✓
Log Population Density in 1500s British Legal Origin N Log European Settler Mortality Log Population Density in 1500s British Legal Origin N Baseline Controls Panel D: Africa & N. America vs	(0.07) 47 Depend -1.1 (0.3) 31 X . S. Amer Depend	$(0.2)$ $47$ ent Variab $-0.9$ $(0.3)$ $31$ $\checkmark$ ica, Asia, ent Variab	-0.3 (0.04) 55 le is Demo -0.5 (0.09) 31 <b>x</b> <b>&amp; Oceania</b> le is Demo	(0.06) 55 cracy Inde -0.5 (0.06) 31 $\checkmark$ a cracy Inde -0.4	(0.3) 57 ex (V-Dem 2.4 (0.4) 33 <b>x</b>	(0.2) 57 , 2019) 1.8 (1.0) 33 ✓
Log Population Density in 1500s British Legal Origin N Log European Settler Mortality Log Population Density in 1500s British Legal Origin N Baseline Controls Panel D: Africa & N. America vs Log European Settler Mortality	(0.07) 47 Depende -1.1 (0.3) 31 <b>X</b> <b>. S. Ameri</b> Depende -0.4	$(0.2)$ $47$ ent Variab $-0.9$ $(0.3)$ $31$ $\checkmark$ ica, Asia, ent Variab $0.08$	-0.3 (0.04) 55 le is Demo -0.5 (0.09) 31 <b>x</b> <b>&amp; Oceania</b> le is Demo	$(0.06)$ $\overline{55}$ cracy Inde $-0.5$ $(0.06)$ $\overline{31}$ $\checkmark$ a cracy Inde	(0.3) 57 ex (V-Dem 2.4 (0.4) 33 x ex (V-Dem 1.2	$(0.2) = \frac{1.8}{57}$ $(1.0) = \frac{1.8}{33}$ $(1.0) = \frac{3}{5}$ $(1.0) = \frac{3}{5}$ $(1.0) = \frac{3}{5}$ $(1.0) = \frac{3}{5}$
Log Population Density in 1500s British Legal Origin N Log European Settler Mortality Log Population Density in 1500s British Legal Origin N Baseline Controls Panel D: Africa & N. America vs Log European Settler Mortality Log Population Density in 1500s	(0.07) 47 Depende -1.1 (0.3) 31 <b>X</b> <b>. S. Ameri</b> Depende -0.4	$(0.2)$ $47$ ent Variab $-0.9$ $(0.3)$ $31$ $\checkmark$ ica, Asia, ent Variab $0.08$	-0.3 (0.04) 55 le is Demo -0.5 (0.09) 31 <b>x</b> <b>&amp; Oceania</b> le is Demo	(0.06) 55 cracy Inde -0.5 (0.06) 31 $\checkmark$ a cracy Inde -0.4	(0.3) 57 ex (V-Dem 2.4 (0.4) 33 x ex (V-Dem	$(0.2) = \frac{1.8}{57}$ (1.0) (1
Log Population Density in 1500s British Legal Origin N Log European Settler Mortality Log Population Density in 1500s British Legal Origin N Baseline Controls Panel D: Africa & N. America vs Log European Settler Mortality Log Population Density in 1500s British Legal Origin N	(0.07) 47 Depende -1.1 (0.3) 31 <b>X</b> . S. Ameri Depende -0.4 (0.10) 50 Depende	(0.2) $47$ ent Variabi -0.9 (0.3) $31$ $\checkmark$ icca, Asia, ent Variabi 0.08 (0.2) $50$ Ent Variabi	$ \begin{array}{r} -0.3 \\ (0.04) \\ \hline 55 \\ e is Demo \\ \hline -0.5 \\ (0.09) \\ \hline 31 \\ \times \\ \& Oceania \\ le is Demo \\ \hline -0.3 \\ (0.04) \\ \end{array} $	$(0.06) \\ \hline 55 \\ cracy Inde \\ -0.5 \\ (0.06) \\ \hline 31 \\ \checkmark \\ a \\ cracy Inde \\ -0.4 \\ (0.06) \\ \hline 58 \\ \hline \end{tabular}$	(0.3) = 57 (V-Dem) $(0.4) = 33$ (0.4) (0.4) (0.4) (0.4) (0.4) (0.5) (	$(0.2) \\ 57 \\ (.2019) \\ (.10) \\ 33 \\ \checkmark \\ (.10) \\ 33 \\ \checkmark \\ (.2019) \\ 0.5 \\ (0.3) \\ 60 \\ (.2010) $
Log Population Density in 1500s British Legal Origin N Log European Settler Mortality Log Population Density in 1500s British Legal Origin N Baseline Controls Panel D: Africa & N. America vs Log European Settler Mortality Log Population Density in 1500s British Legal Origin N Log European Settler Mortality	(0.07) 47 Depende -1.1 (0.3) 31 <b>x</b> . S. Ameri Depende -0.4 (0.10) 50	(0.2) 47 ent Variab -0.9 (0.3) 31 × ica, Asia, ent Variab 0.08 (0.2) 50	$ \begin{array}{r} -0.3 \\ (0.04) \\ \hline 55 \\ \hline 1000 \\ \hline 55 \\ \hline 0.05 \\ (0.09) \\ \hline 31 \\ \times \\ & & \\ & \\ & \\ & \\ & \\ & \\ & \\ & \\ & \\$	$(0.06) \\ \hline 55 \\ cracy Inde \\ \hline -0.5 \\ (0.06) \\ \hline 31 \\ \checkmark \\ \hline 4 \\ cracy Inde \\ \hline -0.4 \\ (0.06) \\ \hline 58 \\ cracy Inde \\ \hline \end{cases}$	(0.3) = 57 (V-Dem) $(0.4) = 33$ (0.4) (0.4) (0.4) (0.4) (0.4) (0.5) (	$(0.2) \\ 57 \\ (.2019) \\ (.10) \\ 33 \\ \checkmark \\ (.10) \\ 33 \\ \checkmark \\ (.2019) \\ 0.5 \\ (.0.3) \\ 60 \\ (.2019)$
Log Population Density in 1500s British Legal Origin N Log European Settler Mortality Log Population Density in 1500s British Legal Origin N Baseline Controls Panel D: Africa & N. America vs Log European Settler Mortality Log Population Density in 1500s British Legal Origin N	(0.07) 47 Dependi -1.1 (0.3) 31 X . S. Amer Dependi -0.4 (0.10) 50 Dependi -0.9	(0.2) $47$ ent Variabi -0.9 (0.3) $31$ $\checkmark$ ica, Asia, ent Variabi 0.08 (0.2) 50 ent Variabi -0.7	$ \begin{array}{r} -0.3 \\ (0.04) \\ \hline 55 \\ e is Demo \\ -0.5 \\ (0.09) \\ \hline 31 \\ \times \\ & & Oceania \\ te is Demo \\ \hline -0.3 \\ (0.04) \\ \hline 58 \\ \end{array} $	$(0.06) \\ \hline 55 \\ cracy Inde \\ -0.5 \\ (0.06) \\ \hline 31 \\ \checkmark \\ a \\ cracy Inde \\ -0.4 \\ (0.06) \\ \hline 58 \\ \hline \end{tabular}$	(0.3) = 57 (V-Dem) $(0.4) = 33$ (0.4) (0.4) (0.4) (0.4) (0.4) (0.5) (	$(0.2) \\ 57 \\ (.2019) \\ (.10) \\ 33 \\ \checkmark \\ (.10) \\ 33 \\ \checkmark \\ (.2019) \\ 0.5 \\ (0.3) \\ 60 \\ (.2010) $
Log Population Density in 1500s British Legal Origin N Log European Settler Mortality Log Population Density in 1500s British Legal Origin N Baseline Controls Panel D: Africa & N. America vs Log European Settler Mortality Log Population Density in 1500s British Legal Origin N Log European Settler Mortality	(0.07) 47 Dependi -1.1 (0.3) 31 X . S. Amer Dependi -0.4 (0.10) 50 Dependi -0.9	(0.2) $47$ ent Variabi -0.9 (0.3) $31$ $\checkmark$ ica, Asia, ent Variabi 0.08 (0.2) 50 ent Variabi -0.7	$ \begin{array}{r} -0.3 \\ (0.04) \\ \hline 55 \\ le is Demo \\ -0.5 \\ (0.09) \\ \hline 31 \\ \times \\ \& Oceania \\ le is Demo \\ \hline -0.3 \\ (0.04) \\ \hline 58 \\ le is Demo \\ \hline -0.4 \\ \end{array} $	$(0.06) = \frac{55}{\text{cracy Inde}} -0.5 \\ (0.06) = \frac{31}{\sqrt{a}} -\frac{31}{\sqrt{a}} -\frac{31}{a$	$(0.3) = \frac{57}{57}$ ex (V-Dem) $(0.4) = \frac{2.4}{33}$ ex (V-Dem) $(0.4) = \frac{1.2}{3}$ ex (V-Dem) $(0.5) = \frac{1.2}{60}$ ex (V-Dem) $1.3 = \frac{1.3}{50}$	(0.2) - 57 - 57 - 57 - 57 - 57 - 57 - 57 - 5
Log Population Density in 1500s British Legal Origin N Log European Settler Mortality Log Population Density in 1500s British Legal Origin N Baseline Controls Panel D: Africa & N. America vs Log European Settler Mortality Log Population Density in 1500s British Legal Origin N Log European Settler Mortality Log European Settler Mortality Log Population Density in 1500s	(0.07) 47 Dependi -1.1 (0.3) 31 X . S. Amer Dependi -0.4 (0.10) 50 Dependi -0.9	(0.2) $47$ ent Variabi -0.9 (0.3) $31$ $\checkmark$ ica, Asia, ent Variabi 0.08 (0.2) 50 ent Variabi -0.7	$ \begin{array}{r} -0.3 \\ (0.04) \\ \hline 55 \\ le is Demo \\ -0.5 \\ (0.09) \\ \hline 31 \\ \times \\ \& Oceania \\ le is Demo \\ \hline -0.3 \\ (0.04) \\ \hline 58 \\ le is Demo \\ \hline -0.4 \\ \end{array} $	$(0.06) = \frac{55}{\text{cracy Inde}} -0.5 \\ (0.06) = \frac{31}{\sqrt{a}} -\frac{31}{\sqrt{a}} -\frac{31}{a$	(0.3) = 57 (V-Dem) $(0.4) = 33$ (0.4) (0.4) (0.4) (0.4) (0.4) (0.5) (	$(0.2) \\ 57 \\ (2019) \\ (10) \\ 33 \\ \checkmark \\ (10) \\ 33 \\ \checkmark \\ (10) \\ 33 \\ \checkmark \\ (10) \\ (0, 0$

Table A8: First-stage Monotonicity Check: By Continent

Notes: This table conducts monotonicity checks for the relationship between the univariate IVs and democracy by dividing the sample into two by random combinations of continents. Panel A has Africa and Asia as the sample for the regressions in the first three rows and Americas and Oceania as the sample in the following three rows. Similarly, Panel B compares Africa and Oceania, with the Americas and Asia; Panel C compares Africa and South America with North America, Asia, and Oceania; Panel D compares Africa and North America with South America, Asia, and Oceania; Panel D compares Africa and North America with South America, Asia, and Oceania; Panel D compares Africa and North America with South America, Asia, and Oceania; Panel D compares Africa and North America with South America, Asia, and Oceania; Panel D compares Africa and North America with South America, Asia, and Oceania; Panel D compares Africa and North America with South America, Asia, and Oceania; Panel D compares Africa and North America with South America, Asia, and Oceania; Panel D compares Africa and North America with South America, Asia, and Oceania; Panel D compares Africa and North America with South America, Asia, and Oceania, The Democracy Index (V-Dem) is normalized to have mean zero and standard deviation one in each year. Columns 1, 3, 5 have no controls, while columns 2, 4, 6 have the following baseline controls: absolute latitude, mean temperature, mean precipitation, population density, median age, and diabetes prevalence. N refers to the number of countries for which IVs, controls, and outcomes are available. Robust standard errors are in parentheses. Variable definitions and data sources are in Appendix Table A1.

Table A9:	Correlation	Among IVs
-----------	-------------	-----------

	Log European Settler Mortality	Log Population Density in 1500s	British Legal Origin	Fraction Speaking English	Fraction Speaking European	Bananas	Coffee	Copper	Maize	Millet	Rice	Rubber	Silver	Sugarcane	Wheat
Log European Settler Mortality	1														
Log Population Density in 1500s	0.367	1													
British Legal Origin	-0.233	-0.0732	1												
Fraction Speaking English	-0.418	-0.407	0.504	1											
Fraction Speaking European	-0.437	-0.545	0.0458	0.564	1										
Bananas	-0.112	0.0502	-0.0388	0.0422	0.0913	1									
Coffee	0.163	0.0814	-0.119	-0.121	0.0233	0.442	1								
Copper	-0.388	-0.227	0.0388	0.0650	0.163	0.0779	0.0957	1							
Maize	0.181	0.0677	-0.0583	0.0364	0.102	0.282	0.238	0.103	1						
Millet	0.325	0.325	0.149	-0.194	-0.488	-0.0700	-0.149	-0.0574	0.178	1					
Rice	0.385	0.203	-0.211	-0.358	-0.138	0.274	0.318	0.0693	0.560	0.216	1				
Rubber	0.0642	0.154	-0.00643	-0.239	-0.188	0.183	0.333	0.305	0.110	0.0653	0.197	1			
Silver	-0.459	-0.247	-0.0487	0.0788	0.299	-0.00279	-0.0131	0.795	-0.0473	-0.208	-0.0844	0.148	1		
Sugarcane	0.102	0.130	-0.0857	-0.100	-0.0177	0.535	0.444	0.0391	0.447	0.143	0.493	0.247	-0.150	1	
Wheat	-0.185	0.00169	-0.0725	-0.0964	0.0214	-0.179	-0.230	0.243	0.0317	0.236	-0.0464	-0.0265	0.275	-0.0582	1

Notes: This table reports the pairwise correlations among the IVs. Variable definitions and data sources are in Appendix Table A1.

IV	Base Sample	Cov(Z, Democracy)	Cov(Z, Potential Omitted Var)	Likely Direction of Bias <u>Cov(Z,PotentialOmittedVar)</u> <u>Cov(Z,Democracy</u> )
European settler mortality IV	N = 81 (countries for- merly under European rule with data available)	Negative (Higher settler mortality led settlers to establish extractive institutions, resulting in lower lev- els of democracy)	Likely negative (Worse disease en- vironments may directly hamper growth)	Likely positive
Population den- sity in 1500s IV	N = 86 (countries for- merly under European rule with data available)	Negative (Higher population den- sity at the beginning of colonial rule led European colonizers to es- tablish extractive institutions, re- sulting in lower levels of democ- racy)	Likely positive (Higher popula- tion density may positively affect growth through higher returns to scale and agglomeration effects)	Likely negative
British legal ori- gin IV	N = 90 (countries for- merly under European rule with data available)	Positive (British colonial rule led to the establishment of a common- law legal system, which is corre- lated with less restrictions on indi- vidual freedoms and higher levels of democracy)	Likely positive (Being formerly subjected to British rule may lead to greater advantages in an Anglo- centric world economy through lin- guistic or cultural influence)	Likely positive
Fraction speak- ing English or European IVs	N = 132 (all countries with data available)	Positive (The fraction of the pop- ulation speaking English or Euro- pean corresponds to the extent of Western influence, which is pos- itively related to higher levels of democracy)	Likely positive (Higher fractions of the population speaking English or a European language may result in more globally competitive human capital)	Likely positive
Crops and miner- als IVs	N = 138 (all countries with data available)	Depends on the commodity	Depends on the commodity	Depends on the commod- ity

Table A10: Directions of Potential Bias in the IV Estimates

\_

\_

#### (a) Reduced form: Mean GDP Growth Rate in 2001-2019



(b) Reduced form: Mean GDP Growth Rate in 2020–2022



(e) Reduced form: Mean GDP Growth Rate in 2020-2022



(c) Reduced form: Mean Nighttime Light Intensity Growth Rate in 2001-2013 (f) Reduced form: Mean Nighttime Light Intensity Growth Rate in 2001-2013



*Notes:* Panels (a), (b), and (c) illustrate the reduced-form relationship between the log of Population Density in the 1500s (IV) and three outcomes: the mean GDP growth rate from 2001 to 2019, mean GDP growth rate from 2020 to 2022, and mean night-time light intensity growth rate from 2001 to 2013. Panels (d), (e), and (f) illustrate the reduced-form relationship between the same outcomes and the British Legal Origin (IV). The Democracy Index (V-Dem) is normalized to have mean zero and standard deviation one in each year. The size of each circle (country) is proportional to its baseline GDP. The colors depend on the level of the democracy index (warmer colors for democracy and darker colors for autocracies). The line is the OLS regression fitted line without controls and weights countries by baseline GDP. The shaded area corresponds to the 95% confidence interval. Variable definitions and data sources are in Appendix Table A1.

(d) Reduced form: Mean GDP Growth Rate in 2001–2019

	1) V	alue Added Char	nels			2) Economic C	hannels			3) Ot	her Channels
	Value Added Growth Agriculture	Value Added Growth Manufacturing	Value Added Growth Services	Tax Revenue Growth	R&D Researchers Growth	New Business Registrations Growth	FDI Growth	Imports from China Growth	Exports from China Growth	Conflict Index Growth	Child Mortality Growth
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)
					Panel A: OL	-					
Democracy Index (V-Dem, 2000)	-1.8	-2.1	-3.1	-0.7	-0.3	1.8	-10.7	-4.8	-10.0	-0.2	1.4
	(0.4)	(0.3)	(0.6)	(0.3)	(1.0)	(3.6)	(9.8)	(0.7)	(1.8)	(0.09)	(0.3)
N	135	124	138	125	118	124	99	134	104	155	153
						cy by Settler Mo	•				
Democracy Index (V-Dem, 2000)	-3.8	-7.3	-4.6	-1.7	-5.1	-8.5	-55.0	-9.0	-14.8	-0.4	2.6
	(0.9)	(2.0)	(0.8)	(0.7)	(6.9)	(9.9)	(42.4)	(3.2)	(4.9)	(0.3)	(0.5)
N	74	68	76	69	62	60	50	70	53	78	81
			Panel C: I	nstrument for I	Democracy by	Population Dens	ity in 1500	)s			
Democracy Index (V-Dem, 2000)	-1.3	-2.6	-3.7	-1.5	-0.9	7.8	-21.8	-6.6	-15.1	-0.6	2.0
	(0.6)	(1.7)	(0.7)	(0.5)	(2.2)	(8.4)	(19.4)	(2.0)	(3.5)	(0.4)	(0.3)
N	77	69	80	71	66	63	48	74	53	83	85
			Pa	nel D: Instrum	ent for Democi	racy by Legal Oi	rigin				
Democracy Index (V-Dem, 2000)	-0.8	-38.9	-3.8	-2.5	-12.3	-112.9	-58.6	-8.6	-21.5	0.2	2.6
	(2.1)	(251.9)	(1.0)	(0.9)	(12.4)	(535.6)	(56.4)	(3.6)	(7.5)	(0.2)	(0.5)
Ν	80	72	83	75	67	65	50	77	55	87	89
			Р	anel E: Instrun	nent for Demo	cracy by Langua	ige				
Democracy Index (V-Dem, 2000)	-2.2	-4.4	-1.8	-3.6	3.6	9.8	-3.2	-3.4	-4.9	0.09	2.9
•	(0.8)	(1.7)	(2.2)	(1.7)	(5.7)	(13.9)	(21.2)	(3.5)	(7.9)	(0.3)	(0.8)
Ν	114	104	120	105	103	104	86	113	89	129	129
			Panel	F: Instrument	or Democracy	by Crops and M	linerals				
Democracy Index (V-Dem, 2000)	-3.0	-4.0	-4.2	-0.9	-6.3	-1.3	-52.0	-8.0	-15.3	-0.2	2.0
	(0.7)	(1.0)	(0.9)	(0.6)	(6.0)	(5.0)	(34.2)	(2.5)	(3.3)	(0.1)	(0.4)
Ν	119	109	122	110	105	107	87	118	92	135	136
					anel G: Use all		~ .		~ =		
Democracy Index (V-Dem, 2000)	-1.3	-2.0	-3.7	-1.0	0.02	8.2	-7.3	-5.1	-11.8	-0.4	1.9
	(0.4)	(0.7)	(0.4)	(0.4)	(2.1)	(7.7)	(11.7)	(1.4)	(3.2)	(0.2)	(0.2)
Ν	65	59	68	60	55	54	42	62	45	68	71
Outcome Mean	6.9	8.4	9.1	1.4	5.0	5.7	5.5	23.2	39.9	0.1	-3.4

Table A11.	Additional	Mechanisms	in	2001-2019

*Notes:* This table reports the OLS (Panel A) and 2SLS (Panels B-G) estimates of democracy's effect on potential mechanisms in 2001-2019. The Democracy Index (V-Dem) is normalized to have mean zero and standard deviation one in each year. Tax Revenue represents the total tax revenues as a percentage of GDP; R&D Researchers represents the number of researchers in R&D per million people; New Business Registrations indicates the density of new business registrations per 1,000 people; Foreign Direct Investment (FDI) represents net outflows as a percentage of GDP; Import from China represents the total value of imports from China in US dollars; Export to China represents the total value of exports from China in US dollars; Conflict Index measures the intensity of internal conflict within a country; Child Mortality represents the mortality rate of infants per 1,000 live births. For IVs, Panel B uses log European settler mortality, Panel C uses log population density in the 1500s, Panel C uses British legal origin, Panel E uses the fraction speaking English and the fraction speaking European, Panel F uses the ability to grow crops and mine minerals, and Panel G uses all the IVs together. We report the global mean of the dependent variables in the bottom row. The specifications do not include controls. N refers to the number of countries for which IVs, controls, and outcomes are available. Robust standard errors are in parentheses. Variable definitions and data sources are in Appendix Table A1.

# Figure A2: Correlation Between Democracy and Additional Outcomes

(a) Mean Excess Deaths per 100k in 2020-2022



(b) Mean Life Satisfaction Growth Rates in 2010-2019



(d) Mean Top 1% Income Share Growth Rates in 2001-2019



(c) Mean Life Satisfaction Growth Rates in 2020-2022



(e) Mean Top 1% Income Share Growth Rates in 2020-2022





(g) Mean CO2 Emissions per Capita Growth Rates in 2020-2022





(f) Mean CO2 Emissions per Capita Growth Rates in 2001-2019

(i) Mean Energy Consumption per Capita Growth Rates in 2020-2022



Energy Consumption Growth in 2001-19

5.0

2.5

0.0

-2.5

China

Pakistar

c

•



*Notes:* This figure shows the correlation between democracy and additional outcomes. The Democracy Index (V-Dem) is normalized to have mean zero and standard deviation one in each year. The size of each circle (country) is proportional to its baseline GDP. The colors depend on the level of the democracy index (warmer colors for democracy and darker colors for autocracies). The line is the fitted line from a univariate OLS regression of the outcome against the democracy index that weights observations by baseline GDP. The shaded area corresponds to the 95% confidence interval.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
Panel A: Two-Stage Least Squa	res		n	1 . 17			0.4	1001 B		2022		
Democracy Index (V-Dem, 2019)	21.2	30.2	22.4	pendent Vari 26.7	26.8	25.9	24.5	20.1	15.8	2022	27.8	29.7
Semocracy macx (V-Benn, 2017)	(8.1)	(6.3)	(8.1)	(3.4)	(6.4)	(4.9)	(11.5)	(11.8)	(9.8)	(7.0)	(4.5)	(2.9)
	0.01	0.00	0.01	0.00	0.00	0.00	0.03	0.09	0.11	0.00	0.00	0.00
			D	ependent Var	iable is M	ean Life Sa	tisfaction (	Growth Rat	e in 2010-2	2019		
Democracy Index (V-Dem, 2000)	0.1	-1.7	-0.1	-0.4	-0.5	-1.5	-0.4	-0.4	-0.1	-0.7	-0.4	-0.7
	(0.5)	(1.4)	(0.4)	(0.4)	(0.2)	(0.7)	(0.3)	(0.2)	(0.3)	(0.1)	(0.2)	(0.1)
	0.85	0.24	0.76 D	0.37 ependent Var	0.04 iable is M	0.05 ean Life Sa	0.21 tisfaction (	0.07 Frowth Rat	0.79 e in 2020-'	0.00	0.09	0.00
Democracy Index (V-Dem, 2019)	-2.5	-2.6	-2.7	-2.8	-2.3	-2.2	-2.7	-2.7	-2.7	-2.7	-2.5	-2.5
Democracy macx (+ Dem, 2019)	(0.2)	(0.5)	(0.4)	(0.5)	(0.4)	(0.5)	(0.4)	(0.3)	(0.4)	(0.3)	(0.2)	(0.5)
	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
			^	endent Variab								
Democracy Index (V-Dem, 2000)	-0.5	-0.4	-0.5	-1.1	-0.2	-0.3	-0.0	-0.2	-0.5	-0.5	-0.4	-0.6
	(0.2) 0.02	(0.5) 0.51	(0.2) 0.01	(0.4) 0.00	(0.3) 0.40	(0.6) 0.63	(0.3) 0.93	(0.2) 0.36	(0.2) 0.01	(0.2) 0.00	(0.1) 0.00	(0.2) 0.00
	0.02	0.51		endent Variab							0.00	0.00
Democracy Index (V-Dem, 2019)	0.5	0.6	0.4	0.4	1.0	1.1	0.5	0.7	0.1	0.4	0.4	0.4
	(0.2)	(0.3)	(0.2)	(0.2)	(0.4)	(0.4)	(0.3)	(0.2)	(0.2)	(0.2)	(0.1)	(0.2)
	0.01	0.03	0.03	0.08	0.02	0.00	0.13	0.00	0.53	0.06	0.00	0.00
				Dependent V	ariable is I	Mean CO2	Emissions	per Capita	in 2001-20	)19		
Democracy Index (V-Dem, 2000)	-3.6	-4.9	-2.6	-2.8	-2.7	-2.5	-2.7	-2.5	-2.8	-2.6	-2.3	-2.5
	(0.9) 0.00	(4.0) 0.22	(0.5)	(0.6)	(0.5)	(1.2)	(0.5) 0.00	(0.5)	(0.5)	(0.3) 0.00	(0.2)	(0.3)
	0.00	0.22	0.00	0.00 Dependent V	0.00 ariable is l	0.03 Mean CO2		0.00 per Capita	0.00 in 2020-20		0.00	0.00
Democracy Index (V-Dem, 2019)	-1.5	-0.7	-1.2	-1.0	-1.3	-1.2	-1.2	-0.9	-1.4	-1.0	-1.2	-0.9
Democracy macx (v-Dem, 2019)	(0.4)	(0.3)	(0.2)	(0.2)	(0.3)	(0.3)	(0.2)	(0.2)	(0.2)	(0.1)	(0.2)	(0.1)
	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
			De	pendent Varia	able is Me	an Energy (	Consumpti	on per Cap	ta in 2001	-2019		
Democracy Index (V-Dem, 2000)	-3.2	-5.0	-2.4	-2.7	-2.5	-2.5	-2.0	-2.0	-2.7	-2.3	-2.3	-2.4
	(0.6)	(3.8)	(0.3)	(0.5)	(0.4)	(1.0)	(0.4)	(0.4)	(0.4)	(0.3)	(0.2)	(0.3)
	0.00	0.19	0.00 De	0.00 pendent Varia	0.00 able is Me	0.01	0.00	0.00	0.00 ta in 2020	0.00	0.00	0.00
Democracy Index (V-Dem, 2019)	-2.0	-1.6	-1.3	-1.4	-1.4	-1.2	-1.5	-1.4	-1.6	-1.3	-1.2	-1.3
Democracy fildex (v-Defil, 2019)	(0.5)	(0.4)	(0.1)	(0.2)	(0.2)	(0.2)	(0.4)	(0.2)	(0.2)	(0.1)	(0.1)	(0.1)
	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
IVs		mortality		ion density		origin		uage		minerals		IVs
Number of IVs	1	1	1	1	1	1	2	2	10	10	15	15
Panel B: Ordinary Least Square	es											
• •							_					
				pendent Vari			· ·					
	24.2	26.1	28.6	29.4	28.6	29.4	15.2	20.2	15.0	19.8	28.9	30.4
	24.2 (5.5)	26.1 (3.6)	28.6 (4.1)	29.4 (2.9)	28.6 (4.1)	29.4 (2.9)	15.2 (9.5)	20.2 (9.9)	15.0 (9.5)	19.8 (9.9)	28.9 (3.9)	30.4 (3.0)
Democracy Index (V-Dem, 2019)	(5.5)	(3.6)	28.6 (4.1) D	29.4 (2.9) ependent Var	28.6 (4.1) iable is M	29.4 (2.9) ean Life Sa	15.2 (9.5) tisfaction (	20.2 (9.9) Growth Rat	15.0 (9.5) e in 2010-2	19.8 (9.9) 2019	(3.9)	(3.0)
Democracy Index (V-Dem, 2019)	-0.4	(3.6)	28.6 (4.1) -0.4	29.4 (2.9) ependent Var -0.7	28.6 (4.1) iable is M -0.4	29.4 (2.9)	15.2 (9.5) tisfaction ( -0.2	20.2 (9.9) Growth Rat -0.6	15.0 (9.5) e in 2010-2 -0.2	19.8 (9.9) 2019 -0.6	(3.9)	(3.0)
Democracy Index (V-Dem, 2019)	(5.5)	(3.6)	28.6 (4.1) D -0.4 (0.2)	29.4 (2.9) ependent Var	28.6 (4.1) iable is M -0.4 (0.2)	29.4 (2.9) ean Life Sa -0.7 (0.1)	15.2 (9.5) tisfaction ( -0.2 (0.2)	20.2 (9.9) Growth Rat -0.6 (0.2)	15.0 (9.5) e in 2010-2 -0.2 (0.2)	19.8 (9.9) 2019 -0.6 (0.2)	(3.9)	(3.0)
Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000)	-0.4	(3.6)	28.6 (4.1) D -0.4 (0.2)	29.4 (2.9) ependent Var -0.7 (0.1)	28.6 (4.1) iable is M -0.4 (0.2)	29.4 (2.9) ean Life Sa -0.7 (0.1)	15.2 (9.5) tisfaction ( -0.2 (0.2)	20.2 (9.9) Growth Rat -0.6 (0.2)	15.0 (9.5) e in 2010-2 -0.2 (0.2)	19.8 (9.9) 2019 -0.6 (0.2)	(3.9)	(3.0)
Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000)	(5.5) -0.4 (0.2)	(3.6) -0.8 (0.1)	28.6 (4.1) D -0.4 (0.2) D -2.5 (0.2)	29.4 (2.9) ependent Var -0.7 (0.1) ependent Var -2.4 (0.4)	28.6 (4.1) iable is M -0.4 (0.2) iable is M -2.5 (0.2)	29.4 (2.9) ean Life Sa -0.7 (0.1) ean Life Sa -2.4 (0.4)	15.2 (9.5) tisfaction ( -0.2 (0.2) tisfaction ( -2.1 (0.3)	20.2 (9.9) Growth Rat -0.6 (0.2) Growth Rat -2.0 (0.4)	15.0 (9.5) e in 2010-2 (0.2) e in 2020-2 -2.1 (0.3)	19.8 (9.9) 2019 -0.6 (0.2) 2022 -2.0 (0.3)	(3.9) -0.5 (0.2)	(3.0) -0.7 (0.1)
Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000)	(5.5) -0.4 (0.2) -2.4	(3.6) -0.8 (0.1) -2.3 (0.4)	28.6 (4.1) D -0.4 (0.2) D -2.5 (0.2) Depe	29.4 (2.9) ependent Var -0.7 (0.1) ependent Var -2.4 (0.4) endent Variab	$ \begin{array}{r} 28.6 \\ (4.1) \\ iable is M \\ -0.4 \\ (0.2) \\ iable is M \\ -2.5 \\ (0.2) \\ ole is Mear \end{array} $	29.4 (2.9) ean Life Sa -0.7 (0.1) ean Life Sa -2.4 (0.4) n Top 1% Ir	15.2 (9.5) tisfaction ( -0.2 (0.2) tisfaction ( -2.1 (0.3) acome Shar	20.2 (9.9) Growth Rat -0.6 (0.2) Growth Rat -2.0 (0.4) re Growth I	$\begin{array}{c} 15.0 \\ (9.5) \\ e \text{ in } 2010-2 \\ \hline -0.2 \\ (0.2) \\ e \text{ in } 2020-2 \\ \hline -2.1 \\ (0.3) \\ \text{Rate in } 200 \end{array}$	19.8 (9.9) 2019 -0.6 (0.2) 2022 -2.0 (0.3) 01-2019	(3.9) -0.5 (0.2) -2.5	(3.0) -0.7 (0.1) -2.4 (0.5)
Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019)	(5.5) -0.4 (0.2) -2.4 (0.2) -0.4	(3.6) -0.8 (0.1) -2.3 (0.4) -0.4	28.6 (4.1) D -0.4 (0.2) D -2.5 (0.2) Depe -0.4	29.4 (2.9) ependent Var -0.7 (0.1) ependent Var -2.4 (0.4) endent Variab -0.4	28.6 (4.1) iable is M -0.4 (0.2) iable is M -2.5 (0.2) ole is Mear -0.4	29.4 (2.9) ean Life Sa -0.7 (0.1) ean Life Sa -2.4 (0.4) n Top 1% Ir -0.4	15.2 (9.5) tisfaction ( -0.2 (0.2) tisfaction ( -2.1 (0.3) acome Shar -0.2	20.2 (9.9) Growth Rat -0.6 (0.2) Growth Rat -2.0 (0.4) re Growth I -0.2	15.0 (9.5) e in 2010-2 (0.2) e in 2020-2 -2.1 (0.3) Rate in 200 -0.2	19.8 (9.9) 2019 -0.6 (0.2) 2022 -2.0 (0.3) 01-2019 -0.2	(3.9) -0.5 (0.2) -2.5 (0.2) -0.4	(3.0) -0.7 (0.1) -2.4 (0.5) -0.4
Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019)	(5.5) -0.4 (0.2) -2.4 (0.2)	(3.6) -0.8 (0.1) -2.3 (0.4)	28.6 (4.1) D -0.4 (0.2) D -2.5 (0.2) Depe -0.4 (0.2)	29.4 (2.9) ependent Var -0.7 (0.1) ependent Var -2.4 (0.4) endent Variab -0.4 (0.2)	28.6 (4.1) iable is M -0.4 (0.2) iable is M -2.5 (0.2) ole is Mear -0.4 (0.2)	29.4 (2.9) ean Life Sa -0.7 (0.1) ean Life Sa -2.4 (0.4) n Top 1% In -0.4 (0.2)	15.2 (9.5) tisfaction ( -0.2 (0.2) tisfaction ( -2.1 (0.3) acome Shar -0.2 (0.2)	20.2 (9.9) Growth Rat -0.6 (0.2) Growth Rat -2.0 (0.4) re Growth I -0.2 (0.2)	15.0 (9.5) e in 2010-2 (0.2) e in 2020-2 -2.1 (0.3) Rate in 200 -0.2 (0.2)	19.8 (9.9) 2019 -0.6 (0.2) 2022 -2.0 (0.3) 01-2019 -0.2 (0.2)	(3.9) -0.5 (0.2) -2.5 (0.2)	(3.0) -0.7 (0.1) -2.4 (0.5)
Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000)	(5.5) -0.4 (0.2) -2.4 (0.2) -0.4 (0.2)	(3.6) -0.8 (0.1) -2.3 (0.4) -0.4 (0.2)	28.6 (4.1) D -0.4 (0.2) D -2.5 (0.2) Depe -0.4 (0.2) Depe	29.4 (2.9) ependent Var -0.7 (0.1) ependent Var -2.4 (0.4) endent Variab -0.4 (0.2) endent Variab	28.6 (4.1) iable is M -0.4 (0.2) iable is M -2.5 (0.2) ole is Mean -0.4 (0.2) ole is Mean	29.4 (2.9) ean Life Sa -0.7 (0.1) ean Life Sa -2.4 (0.4) n Top 1% Ir -0.4 (0.2) n Top 1% Ir	15.2 (9.5) tisfaction ( -0.2 (0.2) tisfaction ( -2.1 (0.3) acome Shar -0.2 (0.2) acome Shar	20.2 (9.9) Growth Rat -0.6 (0.2) Growth Rat -2.0 (0.4) e Growth I -0.2 (0.2) e Growth I	15.0 (9.5) e in 2010-3 -0.2 (0.2) e in 2020-3 -2.1 (0.3) Rate in 200 -0.2 (0.2) Rate in 202	19.8 (9.9) 2019 -0.6 (0.2) 2022 -2.0 (0.3) 01-2019 -0.2 (0.2) 20-2022	(3.9) -0.5 (0.2) -2.5 (0.2) -0.4 (0.2)	(3.0) -0.7 (0.1) -2.4 (0.5) -0.4 (0.2)
Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000)	(5.5) -0.4 (0.2) -2.4 (0.2) -0.4 (0.2) 0.3	(3.6) -0.8 (0.1) -2.3 (0.4) -0.4 (0.2) 0.3	28.6 (4.1) D -0.4 (0.2) D -2.5 (0.2) Depe -0.4 (0.2) Depe 0.4	29.4 (2.9) ependent Var -0.7 (0.1) ependent Var -2.4 (0.4) endent Variab -0.4 (0.2) endent Variab 0.4	28.6 (4.1) iable is M -0.4 (0.2) iable is M -2.5 (0.2) ole is Mear -0.4 (0.2) ole is Mear 0.4	29.4 (2.9) ean Life Sa -0.7 (0.1) ean Life Sa -2.4 (0.4) n Top 1% Ir -0.4 (0.2) n Top 1% Ir 0.4	15.2 (9.5) tisfaction ( -0.2 (0.2) tisfaction ( -2.1 (0.3) acome Shau -0.2 (0.2) ticome Shau -0.2 (0.2) ticome Shau -0.2 (0.2)	20.2 (9.9) Growth Rat -0.6 (0.2) Growth Rat -2.0 (0.4) e Growth I -0.2 (0.2) re Growth I 0.5	15.0 (9.5) e in 2010- -0.2 (0.2) e in 2020- -2.1 (0.3) Rate in 200 -0.2 (0.2) Rate in 202 0.3	19.8 (9.9) 2019 -0.6 (0.2) 2022 -2.0 (0.3) 01-2019 -0.2 (0.2) 20-2022 0.5	(3.9) -0.5 (0.2) -2.5 (0.2) -0.4 (0.2) 0.4	(3.0) -0.7 (0.1) -2.4 (0.5) -0.4 (0.2) 0.4
Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000)	(5.5) -0.4 (0.2) -2.4 (0.2) -0.4 (0.2)	(3.6) -0.8 (0.1) -2.3 (0.4) -0.4 (0.2)	28.6 (4.1) D -0.4 (0.2) D Depe -0.4 (0.2) Depe 0.4 (0.1)	29.4 (2.9) ependent Var -0.7 (0.1) ependent Var (0.1) endent Variab -0.4 (0.2) endent Variab 0.4 (0.1)	28.6 (4.1) iable is M -0.4 (0.2) iable is M -2.5 (0.2) ole is Mear -0.4 (0.2) ole is Mear -0.4 (0.2)	29.4 (2.9) ean Life Sa -0.7 (0.1) ean Life Sa -2.4 (0.4) n Top 1% Ir -0.4 (0.2) n Top 1% Ir 0.4 (0.1)	15.2 (9.5) tisfaction ( -0.2 (0.2) tisfaction ( -2.1 (0.3) accome Shar -0.2 (0.2) accome Shar 0.3 (0.2)	20.2 (9.9) Growth Rat -0.6 (0.2) Growth Rat -2.0 (0.4) e Growth I -0.2 (0.2) re Growth I 0.5 (0.2)	15.0 (9.5) e in 2010-3 (0.2) e in 2020- -2.1 (0.3) Rate in 2020- -0.2 (0.2) Rate in 2020- 20.2 (0.2) Rate in 2020- 0.3 (0.2)	19.8 (9.9) 2019 -0.6 (0.2) 2022 -2.0 (0.3) 11-2019 -0.2 (0.2) 20-2022 0.5 (0.2)	(3.9) -0.5 (0.2) -2.5 (0.2) -0.4 (0.2)	(3.0) -0.7 (0.1) -2.4 (0.5) -0.4 (0.2)
Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019)	(5.5) -0.4 (0.2) -2.4 (0.2) -0.4 (0.2) 0.3 (0.2)	(3.6) -0.8 (0.1) -2.3 (0.4) -0.4 (0.2) 0.3 (0.2)	28.6 (4.1) D -0.4 (0.2) D Depe -0.4 (0.2) Depe 0.4 (0.1)	29.4 (2.9) ependent Var -0.7 (0.1) ependent Variab -0.4 (0.2) endent Variab 0.4 (0.1) Dependent Variab	28.6 (4.1) iable is M -0.4 (0.2) iable is M -2.5 (0.2) ole is Mear -0.4 (0.2) ole is Mear -0.4 (0.2) ole is Mear -0.4 (0.2)	29.4 (2.9) ean Life Sa -0.7 (0.1) ean Life Sa -2.4 (0.4) n Top 1% Ir -0.4 (0.2) n Top 1% Ir 0.4 (0.2) n Top 1% Ir 0.4 (0.1) Mean CO2	15.2 (9.5) (0.5) (0.2) (0.2) (0.2) (0.2) (0.2) (0.3) (0.2) Emissions	20.2 (9.9) Growth Rat -0.6 (0.2) Growth Rat -2.0 (0.4) e Growth I -0.2 (0.2) e Growth I 0.5 (0.2) per Capita	15.0 (9.5) e in 2010-2 (0.2) e in 2020- -2.1 (0.3) Rate in 2020- -0.2 (0.2) Rate in 2020- 0.3 (0.2) in 2001-20	19.8         (9.9)           2019         -0.6           -0.2         (0.2)           2022         -2.0           -0.3)         11-2019           -0.2         (0.2)           00-2022         0.5           (0.2)         119	(3.9) -0.5 (0.2) -2.5 (0.2) -0.4 (0.2) 0.4 (0.1)	(3.0) -0.7 (0.1) -2.4 (0.5) -0.4 (0.2) 0.4 (0.2)
Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019)	(5.5) -0.4 (0.2) -2.4 (0.2) -0.4 (0.2) 0.3	(3.6) -0.8 (0.1) -2.3 (0.4) -0.4 (0.2) 0.3	28.6 (4.1) D -0.4 (0.2) D Depe -0.4 (0.2) Depe 0.4 (0.1)	29.4 (2.9) ependent Var -0.7 (0.1) ependent Var (0.1) endent Variab -0.4 (0.2) endent Variab 0.4 (0.1)	28.6 (4.1) iable is M -0.4 (0.2) iable is M -2.5 (0.2) ole is Mear -0.4 (0.2) ole is Mear -0.4 (0.2)	29.4 (2.9) ean Life Sa -0.7 (0.1) ean Life Sa -2.4 (0.4) n Top 1% Ir -0.4 (0.2) n Top 1% Ir 0.4 (0.1)	15.2 (9.5) tisfaction ( -0.2 (0.2) tisfaction ( -2.1 (0.3) accome Shar -0.2 (0.2) accome Shar 0.3 (0.2)	20.2 (9.9) Growth Rat -0.6 (0.2) Growth Rat -2.0 (0.4) e Growth I -0.2 (0.2) re Growth I 0.5 (0.2)	15.0 (9.5) e in 2010-3 (0.2) e in 2020- -2.1 (0.3) Rate in 2020- -0.2 (0.2) Rate in 2020- 20.2 (0.2) Rate in 2020- 0.3 (0.2)	19.8 (9.9) 2019 -0.6 (0.2) 2022 -2.0 (0.3) 11-2019 -0.2 (0.2) 20-2022 0.5 (0.2)	(3.9) -0.5 (0.2) -2.5 (0.2) -0.4 (0.2) 0.4	(3.0) -0.7 (0.1) -2.4 (0.5) -0.4 (0.2) 0.4
Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019)	(5.5) -0.4 (0.2) -2.4 (0.2) -0.4 (0.2) 0.3 (0.2) -2.2	(3.6) -0.8 (0.1) -2.3 (0.4) -0.4 (0.2) -0.4 (0.2) -1.7	28.6 (4.1) D -0.4 (0.2) D -2.5 (0.2) Depe -0.4 (0.2) Depe 0.4 (0.1) -2.1 (0.2)	29.4 (2.9) ependent Var -0.7 (0.1) ependent Variat -0.4 (0.2) -0.4 (0.2) -0.4 (0.1) Dependent Variat -0.4 (0.1) -0.4 (0.1) -0.4 (0.1) -0.4 (0.1) -0.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.1) -0.4 (0.2) -0.4 (0.2) -0.4 (0.1) -0.4 (0.1) -0.4 (0.1) -0.4 (0.1) -0.4 (0.1) -0.4 (0.1) -0.4 (0.1) -0.4 (0.1) -0.4 (0.1) -0.4 (0.2) -0.4 (0.4) (0.4) -0.4 (0.4) (0.4	28.6 (4.1) iable is M -0.4 (0.2) iable is M -2.5 (0.2) ole is Mear -0.4 (0.2) ole is Mear 0.4 (0.1) ariable is 1 -2.1 (0.2)	29.4 (2.9) ean Life Sa -0.7 (0.1) ean Life Sa -2.4 (0.4) n Top 1% Ir -0.4 (0.2) n Top 1% Ir -0.4 (0.1) Mean CO2 -2.0 (0.4)	15.2 (9.5) (15,5) (15,6) (10,2) (10,2) (10,3) (10,3) (10,2	20.2 (9.9) Growth Rat -0.6 (0.2) Growth Rat -2.0 (0.4) re Growth Rat -2.0 (0.4) re Growth I -0.2 (0.2) re Growth I 0.5 (0.2) per Capita -1.5 (0.4)	15.0 (9.5) e in 2010-2 (0.2) e in 2020-2 -2.1 (0.3) Rate in 2002 -0.2 (0.2) Rate in 2002 -0.2 (0.2) Rate in 2002 -0.3 (0.2) in 2001-20 -1.8 (0.3)	$\begin{array}{c} 19.8\\ (9.9)\\ 2019\\ \hline -0.6\\ (0.2)\\ 2022\\ \hline -2.0\\ (0.3)\\ 11-2019\\ \hline -0.2\\ (0.2)\\ 0.2022\\ \hline 0.5\\ (0.2)\\ 119\\ \hline -1.5\\ (0.4)\\ \end{array}$	(3.9) -0.5 (0.2) -2.5 (0.2) -0.4 (0.2) 0.4 (0.1) -2.1	(3.0) -0.7 (0.1) -2.4 (0.5) -0.4 (0.2) 0.4 (0.2) -2.0
Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000)	(5.5) -0.4 (0.2) -2.4 (0.2) -0.4 (0.2) 0.3 (0.2) -2.2	(3.6) -0.8 (0.1) -2.3 (0.4) -0.4 (0.2) -0.4 (0.2) -1.7	28.6 (4.1) D -0.4 (0.2) D -2.5 (0.2) Depe -0.4 (0.2) Depe 0.4 (0.1) -2.1 (0.2)	29.4 (2.9) ependent Var -0.7 (0.1) ependent Var -2.4 (0.4) endent Variab -0.4 (0.1) Dependent V. -0.4 (0.1) Dependent V.	28.6 (4.1) iable is M -0.4 (0.2) iable is M -2.5 (0.2) ole is Mear -0.4 (0.2) ole is Mear 0.4 (0.1) ariable is 1 -2.1 (0.2)	29.4 (2.9) ean Life Sa -0.7 (0.1) ean Life Sa -2.4 (0.4) n Top 1% Ir -0.4 (0.2) n Top 1% Ir -0.4 (0.1) Mean CO2 -2.0 (0.4)	15.2 (9.5) (15,5) (15,6) (10,2) (10,2) (10,3) (10,3) (10,2	20.2 (9.9) Growth Rat -0.6 (0.2) Growth Rat -2.0 (0.4) re Growth Rat -2.0 (0.4) re Growth I -0.2 (0.2) re Growth I 0.5 (0.2) per Capita -1.5 (0.4)	15.0 (9.5) e in 2010-2 (0.2) e in 2020-2 -2.1 (0.3) Rate in 2002 -0.2 (0.2) Rate in 2002 -0.2 (0.2) Rate in 2002 -0.3 (0.2) in 2001-20 -1.8 (0.3)	$\begin{array}{c} 19.8\\ (9.9)\\ 2019\\ \hline -0.6\\ (0.2)\\ 2022\\ \hline -2.0\\ (0.3)\\ 11-2019\\ \hline -0.2\\ (0.2)\\ 0.2022\\ \hline 0.5\\ (0.2)\\ 119\\ \hline -1.5\\ (0.4)\\ \end{array}$	(3.9) -0.5 (0.2) -2.5 (0.2) -0.4 (0.2) 0.4 (0.1) -2.1	(3.0) -0.7 (0.1) -2.4 (0.5) -0.4 (0.2) 0.4 (0.2) -2.0
Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000)	(5.5) -0.4 (0.2) -2.4 (0.2) -0.4 (0.2) 0.3 (0.2) -2.2 (0.2)	(3.6) -0.8 (0.1) -2.3 (0.4) -0.4 (0.2) 0.3 (0.2) -1.7 (0.4)	28.6 (4.1) D -0.4 (0.2) D -2.5 (0.2) Depe -0.4 (0.1) Depe -0.4 (0.1) -2.1 (0.2) -2.1 (0.2) -2.1 (0.2) -2.1 (0.2) -2.1	29.4 (2.9) ependent Var -0.7 (0.1) ependent Var -2.4 (0.4) endent Variab -0.4 (0.2) endent Var 0.4 (0.1) Dependent V -2.0 (0.4) Dependent V -2.0 (0.4)	28.6 (4.1) iable is M -0.4 (0.2) iable is M -2.5 (0.2) ole is Mear -0.4 (0.2) ole is Mear -0.4 (0.1) ariable is 1 -2.1 (0.2) ariable is 1 -2.1 (0.2) ariable is 1	29.4 (2.9) ean Life Sa -0.7 (0.1) ean Life Sa -2.4 (0.4) n Top 1% Ir -0.4 (0.2) n Top 1% Ir -0.4 (0.2) n Top 1% Ir -0.4 (0.1) Mean CO2 -2.0 (0.4) Mean CO2 -0.8 (0.1)	15.2 (9.5) tisfaction ( -0.2 (0.2) tisfaction ( -2.1 (0.3) teome Shat -0.2 (0.2) Emissions -1.8 (0.3) Emissions -1.8 (0.3)	20.2 (9.9) Growth Rat -0.6 (0.2) Growth Rat -2.0 (0.4) e Growth I -0.2 (0.2) e Growth I 0.5 (0.2) per Capita -1.5 (0.4) per Capita -0.5 (0.4)	15.0 (9.5) e in 2010 (0.2) e in 2020 (0.2) e in 2020 (0.3) (0.3) (0.2) (0.2) (0.2) (0.3) (0.2) (0.3) (0.2) (0.3) (	$\begin{array}{c} 19.8\\ (9.9)\\ 2019\\ \hline -0.6\\ (0.2)\\ 2022\\ \hline -2.0\\ (0.3)\\ 11-2019\\ \hline -0.2\\ (0.2)\\ 10-2022\\ \hline 0.5\\ (0.2)\\ 119\\ \hline -1.5\\ (0.4)\\ 122\\ \hline -0.8\\ (0.1)\\ \end{array}$	(3.9) -0.5 (0.2) -2.5 (0.2) -0.4 (0.2) 0.4 (0.1) -2.1 (0.2)	(3.0) -0.7 (0.1) -2.4 (0.5) -0.4 (0.2) 0.4 (0.2) -2.0 (0.4)
Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000)	(5.5) -0.4 (0.2) -2.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.3 (0.2) -2.2 (0.2) -1.2 (0.1)	(3.6) -0.8 (0.1) -2.3 (0.4) -0.4 (0.2) 0.3 (0.2) -1.7 (0.4) -0.8 (0.1)	28.6 (4.1) D -0.4 (0.2) D -2.5 (0.2) Depe -0.4 (0.1) -2.1 (0.2) -2.1 (0.2) Depe -0.4 (0.1) Depe	29.4 (2.9) ependent Var -0.7 (0.1) ependent Var -2.4 (0.4) endent Variab -0.4 (0.1) Dependent Var -2.0 (0.4) Dependent V -2.0 (0.4) Dependent Var -2.0 (0.1) pendent Var	28.6 (4.1) iable is M -0.4 (0.2) iable is M -2.5 (0.2) ole is Mear -0.4 (0.1) ariable is 1 -2.1 (0.2) ariable is 1 -2.1 (0.2) ariable is 1	29.4 (2.9) ean Life Sa -0.7 (0.1) ean Life Sa -2.4 (0.4) n Top 1% Ir -0.4 (0.2) n Top 1% Ir -0.4 (0.2) n Top 1% Ir -0.4 (0.1) Mean CO2 -2.0 (0.4) Mean CO2 -0.8 (0.1) an Energy (0.1)	15.2 (9.5) tisfaction ( -0.2 (0.2) tisfaction ( -2.1 (0.3) acome Shau -0.2 (0.2) termissions -1.8 (0.3) Emissions -1.8 (0.3) Emissions -1.1 (0.1) Consumptio	20.2 (9.9) Growth Rat -0.6 (0.2) Growth Rat -2.0 (0.4) e Growth I -0.2 (0.2) e Growth I -0.2 (0.2) e Growth I -0.5 (0.2) per Capita -0.5 (0.4) per Capita -0.8 (0.1) on per Capita	15.0 (9.5) e in 2010-2 (0.2) e in 2020-2 (0.3) e in 2020-2 (0.3) Rate in 2020- (0.3) (0.2) Rate in 2020 (0.2) (0.2) (0.2) (0.2) (0.2) (0.3) (0.2) (0.3) (0.2) (0.3) (0.2) (0.3) (0.2) (0.3) (0.2) (0.3) (0.2) (0.2) (0.3) (0.2) (0.3) (0.2) (0.3) (0.2) (0.3) (0.2) (0.3) (0.2) (0.3) (0.3) (0.2) (0.3	$\begin{array}{c} 19.8\\ (9.9)\\ (2019\\ \hline -0.6\\ (0.2)\\ 2022\\ \hline -2.0\\ (0.3)\\ 11-2019\\ \hline -0.2\\ (0.3)\\ 11-2019\\ \hline -0.2\\ (0.3)\\ 10-2012\\ \hline -0.2\\ (0.4)\\ 122\\ \hline -0.8\\ (0.1)\\ -2019\\ \end{array}$	(3.9) -0.5 (0.2) -2.5 (0.2) -0.4 (0.2) 0.4 (0.1) -2.1 (0.2) -1.1 (0.2)	(3.0) $-0.7$ $(0.1)$ $-2.4$ $(0.5)$ $-0.4$ $(0.2)$ $0.4$ $(0.2)$ $-2.0$ $(0.4)$ $-0.9$ $(0.1)$
Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000)	(5.5) -0.4 (0.2) -2.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.2 (0.2) -1.2 (0.1) -2.2	(3.6) -0.8 (0.1) -2.3 (0.4) -0.4 (0.2) -0.4 (0.2) -0.7 (0.4) -0.8 (0.1) -1.9	28.6 (4.1) D -0.4 (0.2) D Depe -0.4 (0.2) Depe 0.4 (0.2) Depe 0.4 (0.1) D -2.1 (0.2) D -2.1	29.4 (2.9) ependent Vari -0.7 (0.1) ependent Variab -0.4 (0.2) endent Variab -0.4 (0.2) endent Variab -0.4 (0.1) Dependent V -2.0 (0.4) Dependent Variab -0.4 (0.1) Dependent Variab -0.4 (0.1) Dependent Variab -0.4 (0.1) -2.0 (0.1)	$\begin{array}{c} \hline 28.6 \\ (4.1) \\ iable is M \\ \hline -0.4 \\ (0.2) \\ iable is M \\ -2.5 \\ (0.2) \\ ole is Mear \\ \hline -0.4 \\ (0.2) \\ ole is Mear \\ \hline -0.4 \\ (0.2) \\ ole is Mear \\ \hline -0.4 \\ (0.1) \\ ariable is 1 \\ \hline -2.1 \\ \hline -2.1 \\ \hline \\ ariable is Mear \\ \hline -2.1 \\ \hline \end{array}$	29.4 (2.9) ean Life Sa -0.7 (0.1) ean Life Sa -2.4 (0.4) n Top 1% Ir -0.4 (0.2) n Top 1% Ir -0.4 (0.2) n Top 1% Ir -0.4 (0.1) Mean CO2 -2.0 (0.4)	15.2 (9.5) (9.5) (0.2) (0.2) (0.2) (0.2) (0.2) (0.2) (0.2) (0.2) (0.2) (0.2) (0.2) Emissions -1.8 (0.3) Emissions -1.8	20.2 (9.9) Growth Rat -0.6 (0.2) Growth Rat -2.0 (0.2) Growth Rat -2.0 (0.2) e Growth I -0.2 (0.2) e Growth I -0.2 (0.2) per Capita -1.5 (0.4) per Capita -0.8 (0.1) n per Capita -1.4	15.0         (9.5)           e in 2010-2         -0.2           (0.2)         e in 2020-2           -2.1         (0.3)           Rate in 2000         -0.2           (0.2)         Rate in 2000           -0.2         (0.2)           Rate in 2001         -0.2           (0.3)         (0.2)           in 20001-20         -1.8           (0.3)         in 2020-20           -1.1         (0.1)           -1.8         2001	$\begin{array}{c} 19.8 \\ (9.9) \\ 2019 \\ \hline \\ -0.6 \\ (0.2) \\ 2022 \\ -2.0 \\ (0.2) \\ 10-2019 \\ \hline \\ -0.2 \\ (0.2) \\ 10-2022 \\ \hline \\ 0.5 \\ (0.2) \\ 10-2022 \\ \hline \\ 0.5 \\ (0.2) \\ 10-2019 \\ \hline \\ -1.5 \\ (0.1) \\ -2019 \\ \hline \\ -1.4 \end{array}$	(3.9) -0.5 (0.2) -2.5 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.1) -2.1 (0.2) -2.1	(3.0) -0.7 (0.1) -2.4 (0.5) -0.4 (0.2) 0.4 (0.2) -2.0 (0.4) -2.0 (0.4) -2.1
Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000)	(5.5) -0.4 (0.2) -2.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.3 (0.2) -2.2 (0.2) -1.2 (0.1)	(3.6) -0.8 (0.1) -2.3 (0.4) -0.4 (0.2) 0.3 (0.2) -1.7 (0.4) -0.8 (0.1)	28.6 (4.1) D -0.4 (0.2) D Depe -0.4 (0.2) Depe 0.4 (0.1) -2.1 (0.2) -1.1 (0.1) De -2.1 (0.2)	29.4 (2.9) ependent Var (0.1) ependent Var -0.7 (0.1) ependent Variab -0.4 (0.2) endent Variab -0.4 (0.2) endent Variab -0.4 (0.1) Dependent V -2.0 (0.4) Dependent Variab -0.8 (0.1) pendent Variab -0.8 (0.1) -0.8 (0.1) -0.8 (0.1) -0.8 (0.1) -0.8 (0.1) -0.8 (0.1) -0.4 (0.1) -	28.6 (4.1) iable is M -0.4 (0.2) iable is M -2.5 (0.2) ble is Mear -0.4 (0.2) ble is Mear -0.4 (0.1) ariable is I -2.1 (0.1) ariable is M -1.1 (0.1) able is Mear -2.2	29.4 (2.9) ean Life Sa -0.7 (0.1) ean Life Sa -2.4 (0.1) a Top 1% Ir -0.4 (0.2) a Top 1% Ir -0.4 (0.1) Mean CO2 -2.0 (0.4) -2.0 (0.4)	15.2 (9.5) tisfaction ( -0.2 (0.2) tisfaction ( -2.1 (0.2) tisfaction ( -0.2 (0.2) tisfaction ( -0.3 (0.2) tisfaction ( -1.1 (0.1) tisfaction ( -1.1 (0.3) tisfaction ( -1.1 (0.3) tisfaction ( -1.1 (0.3) tisfaction ( -1.1) tisfaction ( -1.1) tis	20.2 (9.9) Growth Rat -0.6 (0.2) Growth Rat -2.0 (0.2) e Growth I -0.2 (0.2) re Growth I -0.2 (0.2) re Growth I -0.2 (0.2) per Capita -0.5 (0.1) on per Capita -0.8 (0.1)	15.0 (9.5) e in 2010-2 (0.2) e in 2020-2 -2.1 (0.2) e in 2020-2 (0.2) Rate in 2020 -0.2 (0.2) Rate in 2020 -0.2 (0.2) n 2001-2 (0.2) -1.8 (0.1) ta in 2020-2 -1.1 (0.1) ta in 2020-2 -1.1 (0.1) ta in 2020-2 -1.1 (0.3)	$\begin{array}{c} 19.8 \\ (9.9) \\ 2019 \\ \hline \\ -0.6 \\ (0.2) \\ 2022 \\ \hline \\ -2.0 \\ (0.2) \\ 2022 \\ \hline \\ -0.2 \\ (0.2) \\ 10.2019 \\ \hline \\ -0.2 \\ (0.2) \\ \hline \\ 0.5 \\ (0.2) \\ 119 \\ \hline \\ -1.5 \\ (0.1) \\ -2019 \\ \hline \\ -0.8 \\ (0.1) \\ -2019 \\ \hline \\ -1.4 \\ (0.5) \\ \hline \end{array}$	(3.9) -0.5 (0.2) -2.5 (0.2) -0.4 (0.2) 0.4 (0.1) -2.1 (0.2) -1.1 (0.2)	(3.0) $-0.7$ $(0.1)$ $-2.4$ $(0.5)$ $-0.4$ $(0.2)$ $0.4$ $(0.2)$ $-2.0$ $(0.4)$ $-0.9$ $(0.1)$
Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019)	(5.5) $(0.2)$ $(0.2)$ $(0.2)$ $(0.2)$ $(0.2)$ $(0.2)$ $(0.2)$ $(0.2)$ $(0.2)$ $(0.2)$ $(0.1)$ $(0.2$	(3.6) -0.8 (0.1) -2.3 (0.4) -0.4 (0.2) 0.3 (0.2) -1.7 (0.4) -0.8 (0.1) -1.9 (0.4)	28.6 (4.1) D -0.4 (0.2) Depe -0.4 (0.2) Depe 0.4 (0.1) -2.1 (0.2) -1.1 (0.2) -2.1 (0.2) Depe 0.4 (0.1) Depe 0.4 (0.1) Depe 0.4 (0.2) Depe 0.4 (0.4) Depe 0.4 (0.4) Depe 0.4 (0.4) Depe 0.4 (0.4) Depe 0.4 (0.4) Depe 0.4 (0.4) Depe 0.4 (0.4) Depe 0.4 (0.4) Depee 0.4 (0.4) (	29.4 (2.9) ependent Var -0.7 (0.1) ependent Var -2.4 (0.4) -0.4 (0.4) -0.4 (0.2) endent Variab -0.4 (0.1) Dependent V -2.0 (0.4) Dependent Var -2.0 (0.4) Dependent Var -2.0 (0.1) Dependent Var -0.0 (0.1) Dependent Var -0 (0.1) Dependent Var -0 (0.1) Dependent Var -0 (0.1) Dependent Var -0 (0.1) Dependent Var -0 (0.1) Dependent Var -0 (0.1) Dependent Var -0 (0.1) Dependent Var -0 Dependent -0 Dependent -0 Dependent -0 Dependent -0 Dependent -0 Dependent -0 Dependent -0 Dependent -0 Dependent -0 Dependent -0 Dep	$\begin{array}{c} 28.6 \\ (4.1) \\ iable is M \\ -0.4 \\ (0.2) \\ iable is M \\ -2.5 \\ (0.2) \\ ole is Mear \\ -0.4 \\ (0.2) \\ ole is Mear \\ -0.4 \\ (0.2) \\ ole is Mear \\ -2.1 \\ (0.2) \\ ariable is 1 \\ -2.1 \\ (0.2) \\ ariable is Mear \\ -2.1 \\ (0.1) \\ able is Mear \\ -2.1 \\ (0.2) \\ able is Mear \\ -2.1 \\ (0.2) \\ able is Mear \\ -2.1 \\ (0.2) \\ (0.$	29.4 (2.9) ean Life Sa -0.7 (0.1) ean Life Sa -2.4 (0.4) n Top 1% Ir -0.4 (0.2) n Top 1% Ir -0.4 (0.2) n Top 1% Ir -0.4 (0.1) Mean CO2 -2.0 (0.4) Mean CO2 -0.8 (0.1) an Energy ( -2.0) (0.4) an Energy ( -2.0) (0.4)	15.2 (9.5) tisfaction ( -0.2 (0.2) tisfaction ( -2.1 (0.3) teome Shat -0.2 (0.2) teome Shat -0.3 (0.2) teome Shat -1.8 (0.3) Emissions -1.8 (0.3) Consumpti -1.8 (0.3) Consumpti	20.2 (9.9) Growth Rat -0.6 (0.2) Growth Rat -2.0 (0.4) e Growth I -0.2 (0.2) e Growth I -0.5 (0.2) per Capita -1.5 (0.4) per Capita -0.4 (0.1) on per Cap -1.5 (0.5) on per Cap	15.0 (9.5) e in 2010-2 (0.2) e in 2020-2 (0.2) e in 2020-2 (0.3) Rate in 2020 (0.2) (0.2) (0.2) (0.2) (0.3) (0.3) in 2001-22 -1.8 (0.3) in 2001-2 (-1.1 (0.1) ta in 2001 -1.8 (0.3) ta in 2020	$\begin{array}{c} 19.8 \\ (9.9) \\ 2019 \\ \hline -0.6 \\ (0.2) \\ 2022 \\ \hline -2.0 \\ (0.3) \\ 11-2019 \\ \hline -0.2 \\ (0.3) \\ 11-2019 \\ \hline -0.2 \\ (0.2) \\ 10-2022 \\ \hline 0.5 \\ (0.2) \\ 100 \\ $	(3.9) -0.5 (0.2) -2.5 (0.2) -0.4 (0.2) 0.4 (0.1) -2.1 (0.2) -1.1 (0.2) -2.1 (0.2)	(3.0) -0.7 (0.1) -2.4 (0.5) -0.4 (0.2) 0.4 (0.2) -2.0 (0.4) -2.0 (0.4) -2.1 (0.4)
Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000)	(5.5) -0.4 (0.2) -2.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.2) -2.4 (0.2) -0.4 (0.2) -2.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.2) -2.2 (0.1) -2.2 (0.2) -1.2 (0.2) -1.2 (0.2) -1.2 (0.2) -1.2 (0.2) -1.2 (0.2) -1.2 (0.2) -1.2 (0.2) -1.3 -1.3	(3.6) -0.8 (0.1) -2.3 (0.4) -0.4 (0.2) 0.3 (0.2) -1.7 (0.4) -0.8 (0.1) -1.9 (0.4) -1.0	28.6 (4.1) D -0.4 (0.2) D Depe -0.4 (0.2) Depe 0.4 (0.1) -2.1 (0.2) -1.1 (0.1) De -2.1 (0.2) De -1.1	29.4 (2.9) ependent Var -0.7 (0.1) ependent Variab -2.4 (0.4) endent Variab -0.4 (0.1) Dependent Variab -2.0 (0.4) Dependent Variab -2.0 (0.4) Dependent Variab -2.0 (0.4) Dependent Variab -2.0 (0.4) Dependent Variab -2.0 (0.4) Dependent Variab -2.0 (0.4) Dependent Variab -2.0 (0.1) -2.0 (0.4) Dependent Variab -2.0 (0.4) (0.4) -2.0 (0.4) (0.4) (0.4) (0.4) (0.4)(	28.6 (4.1) itable is M -0.4 (0.2) itable is M -2.5 (0.2) ole is Mear -0.4 (0.1) ariable is I -0.4 (0.2) de is Mear -0.4 (0.1) ariable is I -2.1 (0.2) ariable is Mear -2.1 (0.2) able is Mear -2.1 (0.2) -2.1 (0.2) -2.1 (0.2) -2.1 (0.2) -2.1 (0.2) -2.1 (0.2) -2.1 (0.2) -2.1 (0.2) -2.1 (0.2) -2.1 (0.2) -2.1 (0.2) -2.1 (0.2) -2.1 (0.2) -2.1 (0.2) -2.1 (0.2) -2.1 -2.1 -2.1 (0.2) -2.1 -2.1 -2.1 -2.1 (0.2) -2.1 -2.1 -2.1 -2.1 (0.2) -2.1 -2.1 -2.1 -2.1 -2.1 -2.1 -2.1 -2.1	29.4 (2.9) ean Life Sa -0.7 (0.1) ean Life Sa -2.4 (0.4) n Top 1% Ir -0.4 (0.2) n Top 1% Ir -0.4 (0.2) n Top 1% Ir -0.4 (0.1) Mean CO2 -2.0 (0.4) Mean CO2 -0.8 (0.1) an Energy (0 -2.0 (0.1) -2.0 (0.1) -2.0 (0.2) -2.0 (0.1) -2.0 (0.2) -2.0 (0.2) -2.0 -2.0 (0.1) -2.0 (0.2) -2.0 -2.0 (0.2) -2.4 (0.2) -2.0 (0.2) -2.0 (0.2) -2.0 (0.2) -2.0 (0.2) -2.0 (0.4) -2.0 (0.4) -2.0 (0.4) -2.0 (0.4) -2.0 (0.4) -2.0 (0.4) -2.0 (0.4) -2.0 (0.4) -2.0 (0.4) -2.0 (0.4) -2.0 (0.4) -2.0 (0.4) -2.0 (0.4) -2.0 (0.4) -2.0 (0.4) -2.0 (0.7) (0.7) -2.0 (0.7) (0.7) (0.7) (0.7) (0.7) (	15.2 (9.5) tisfaction ( -0.2 (0.2) tisfaction ( -2.1 (0.3) acome Shan -0.2 (0.2) teome Shan -0.2 (0.2) Emissions -1.8 (0.3) Emissions -1.1 (0.1) Consumptio -1.3	20.2 (9.9) Growth Rat -0.6 (0.2) Growth Rat -2.0 (0.4) e Growth I -0.2 (0.2) e Growth I -0.2 (0.2) e Growth I -0.2 (0.2) e Growth I -0.5 (0.2) per Capita -1.5 (0.4) per Capita -0.8 (0.1) on per Cap -1.4 (0.5) on per Cap -1.1	15.0 (9.5) e in 2010-2 (0.2) e in 2020-2 (0.2) e in 2020-2 (0.2) (0.2) (0.2) (0.2) (0.2) (0.2) (0.2) (0.2) (0.2) (0.2) (0.2) (0.2) (0.2) (0.3) in 2020-2 (-1.1) (0.1) ta in 2020 -1.3	$\begin{array}{c} 19.8\\ (9.9)\\ 2019\\ \hline 0.6\\ (0.2)\\ 2022\\ \hline -2.0\\ (0.3)\\ 10-2019\\ \hline -0.2\\ (0.3)\\ 10-2019\\ \hline 0.2\\ (0.2)\\ 10-2022\\ \hline 0.5\\ (0.2)\\ 10-2022\\ \hline -0.8\\ (0.1)\\ -2019\\ \hline -1.4\\ (0.1)\\ -2019\\ \hline -1.4\\ (0.1)\\ -2022\\ \hline -1.1\\ \end{array}$	(3.9) -0.5 (0.2) -2.5 (0.2) -0.4 (0.2) -0.4 (0.2) -0.4 (0.1) -2.1 (0.2) -1.1 (0.2) -1.2	(3.0) -0.7 (0.1) -2.4 (0.5) -0.4 (0.2) -0.4 (0.2) -2.0 (0.4) -2.0 (0.4) -2.0 (0.4) -2.1 (0.4) -2.1 (0.4) -1.2
Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019) Democracy Index (V-Dem, 2000) Democracy Index (V-Dem, 2019)	(5.5) $(0.2)$ $(0.2)$ $(0.2)$ $(0.2)$ $(0.2)$ $(0.2)$ $(0.2)$ $(0.2)$ $(0.2)$ $(0.2)$ $(0.1)$ $(0.2$	(3.6) -0.8 (0.1) -2.3 (0.4) -0.4 (0.2) 0.3 (0.2) -1.7 (0.4) -0.8 (0.1) -1.9 (0.4)	28.6 (4.1) D -0.4 (0.2) Depe -0.4 (0.2) Depe 0.4 (0.1) -2.1 (0.2) -1.1 (0.2) -2.1 (0.2) Depe 0.4 (0.1) Depe 0.4 (0.1) Depe 0.4 (0.2) Depe 0.4 (0.4) Depe 0.4 (0.4) Depe 0.4 (0.4) Depe 0.4 (0.4) Depe 0.4 (0.4) Depe 0.4 (0.4) Depe 0.4 (0.4) Depe 0.4 (0.4) Depee 0.4 (0.4) (	29.4 (2.9) ependent Var -0.7 (0.1) ependent Var -2.4 (0.4) -0.4 (0.4) -0.4 (0.2) endent Variab -0.4 (0.1) Dependent V -2.0 (0.4) Dependent Var -2.0 (0.4) Dependent Var -2.0 (0.1) Dependent Var -0.0 (0.1) Dependent Var -0 (0.1) Dependent Var -0 (0.1) Dependent Var -0 (0.1) Dependent Var -0 (0.1) Dependent Var -0 (0.1) Dependent Var -0 (0.1) Dependent Var -0 (0.1) Dependent Var -0 Dependent -0 Dependent -0 Dependent -0 Dependent -0 Dependent -0 Dependent -0 Dependent -0 Dependent -0 Dependent -0 Dependent -0 Dep	$\begin{array}{c} 28.6 \\ (4.1) \\ iable is M \\ -0.4 \\ (0.2) \\ iable is M \\ -2.5 \\ (0.2) \\ ole is Mear \\ -0.4 \\ (0.2) \\ ole is Mear \\ -0.4 \\ (0.2) \\ ole is Mear \\ -2.1 \\ (0.2) \\ ariable is 1 \\ -2.1 \\ (0.2) \\ ariable is Mear \\ -2.1 \\ (0.1) \\ able is Mear \\ -2.1 \\ (0.2) \\ able is Mear \\ -2.1 \\ (0.2) \\ able is Mear \\ -2.1 \\ (0.2) \\ (0.$	29.4 (2.9) ean Life Sa -0.7 (0.1) ean Life Sa -2.4 (0.4) n Top 1% Ir -0.4 (0.2) n Top 1% Ir -0.4 (0.2) n Top 1% Ir -0.4 (0.1) Mean CO2 -2.0 (0.4) Mean CO2 -0.8 (0.1) an Energy ( -2.0) (0.4) an Energy ( -2.0) (0.4)	15.2 (9.5) tisfaction ( -0.2 (0.2) tisfaction ( -2.1 (0.3) teome Shat -0.2 (0.2) teome Shat -0.3 (0.2) teome Shat -1.8 (0.3) Emissions -1.8 (0.3) Consumpti -1.8 (0.3) Consumpti	20.2 (9.9) Growth Rat -0.6 (0.2) Growth Rat -2.0 (0.4) e Growth I -0.2 (0.2) e Growth I -0.5 (0.2) per Capita -1.5 (0.4) per Capita -0.4 (0.1) on per Cap -1.5 (0.5) on per Cap	15.0 (9.5) e in 2010-2 (0.2) e in 2020-2 (0.2) e in 2020-2 (0.3) Rate in 2020 (0.2) (0.2) (0.2) (0.2) (0.3) (0.3) in 2001-22 -1.8 (0.3) in 2001-2 (-1.1 (0.1) ta in 2001 -1.8 (0.3) ta in 2020	$\begin{array}{c} 19.8 \\ (9.9) \\ 2019 \\ \hline -0.6 \\ (0.2) \\ 2022 \\ \hline -2.0 \\ (0.3) \\ 11-2019 \\ \hline -0.2 \\ (0.3) \\ 11-2019 \\ \hline -0.2 \\ (0.2) \\ 10-2022 \\ \hline 0.5 \\ (0.2) \\ 100 \\ $	(3.9) -0.5 (0.2) -2.5 (0.2) -0.4 (0.2) 0.4 (0.1) -2.1 (0.2) -1.1 (0.2) -2.1 (0.2)	(3.0) $-0.7$ $(0.1)$ $-2.4$ $(0.5)$ $-0.4$ $(0.2)$ $0.4$ $(0.2)$ $-2.0$ $(0.4)$ $-0.9$ $(0.1)$ $-2.1$ $(0.4)$

#### Table A12: 2SLS Regression Estimates of Democracy's Effects on Additional Outcomes

*Notes:* Excess deaths refer to deaths occurring beyond the expected level, based on the typical, non-crisis mortality rate for the specific population in each country. Life Satisfaction values are based on answers to the main life evaluation questions asked in the poll. The poll asks respondents to think of a Satisfaction, with the best possible life for them being a 10 and the worst possible life being a 0. CO2 Emissions per Capita refers to the Carbon Dioxide emissions from fossil fuels and industry released into the atmosphere per person. Energy Consumption per Capita refers to the amount of primary energy consumed in kilowatt-hours per person. The Democracy Index (V-Dem) is normalized to have mean zero and standard deviation one in each year. For IVs, columns 1 and 2 use log European settler mortality, columns 3 and 4 use log population density in the 1500s, columns 9 and 10 use the ability to grow crops and mine minerals, and columns 11 and 12 use all the IVs together. Robust standard errors are in parentheses. The p-values are displayed as 0.00 if they are strictly smaller than the 0.005 threshold. The F-statistics are from the first-stage regressions of the IVs against the democracy index in 2019. Panel B reports the OLS estimates. Columns 1, 3, 5, 7, 9, and 11 have no controls, while columns 2, 4, 6, 8, 10, and 12 include the following baseline controls: absolute latitude, mean temperature, mean precipitation, population density, and median age. For outcomes in 2020-2022, we also control for diabetes prevalence. N refers to the number of countries for which IVs, controls, and outcomes are available. Variable definitions and data sources are in Appendix Table A1.

	(1)	( <b>2</b> )	(2)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
	. ,	(2)	(3)	(4)	(5)	(0)	(7)	(8)	(9)	(10)	(11)	(12)
Panel A: Two-Stage Least Squa	res											
			Depe	ndent Variab	le is Mear	1 Per-Cap	ita GDP C	browth R	ate in 200	1-2019		
Democracy Index (V-Dem, 2000)	-2.3	-4.9	-2.1	-3.4	-1.8	-2.3	-1.3	-1.5	-2.0	-2.0	-2.1	-2.8
	(0.4)	(3.4)	(0.5)	(0.7)	(0.6)	(1.4)	(0.7)	(0.6)	(0.5)	(0.6)	(0.4)	(0.3)
	0.0	0.2	0.0	0.0	0.0	0.1	0.0	0.0	0.0	0.0	0.0	0.0
			Depe	ndent Variab	le is Mear	n Per-Cap	ita GDP C	Growth R	ate in 202	0-2022		
Democracy Index (V-Dem, 2019)	-0.9	-1.0	-1.0	-1.2	-0.9	-0.9	-0.9	-0.8	-1.0	-1.0	-1.0	-1.1
	(0.2)	(0.2)	(0.2)	(0.1)	(0.2)	(0.1)	(0.2)	(0.2)	(0.2)	(0.2)	(0.1)	(0.1)
	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
		Deper	ndent Vari	able is Mean	Per-Capi	ta Nightti	me Light	Intensity	Growth F	Rate in 2001	-2013	
Democracy Index (V-Dem, 2000)	-2.5	-5.7	-1.6	-1.5	-2.4	-3.7	-1.8	-3.0	-2.0	-3.0	-1.9	-2.1
, , , , , , , , , , , , , , , , , , , ,	(0.6)	(4.1)	(0.4)	(0.6)	(0.5)	(1.2)	(0.9)	(1.0)	(0.5)	(0.6)	(0.3)	(0.4)
	0.0	0.2	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
IVs	settler	mortality	populat	ion density	legal	origin	lang	uage	crops &	t minerals	all	IVs
Number of IVs	1	1	1	1	1	1	2	2	10	10	15	15
Observations	81	81	86	86	90	90	130	130	136	136	71	71
Panel B: Ordinary Least Squar	es											
			Depe	ndent Variab	le is Mear	1 Per-Cap	ita GDP C	Growth R	ate in 200	1-2019		
Democracy Index (V-Dem, 2000)	-2.0	-2.1	-1.9	-2.3	-1.9	-2.3	-1.4	-1.3	-1.4	-1.3	-2.0	-2.3
	(0.4)	(0.5)	(0.5)	(0.5)	(0.5)	(0.5)	(0.6)	(0.7)	(0.6)	(0.7)	(0.4)	(0.5)
	()	(010)	( )	ndent Variab	( )	( )	( )	( )	( )	( )	()	(010)
Democracy Index (V-Dem, 2019)	-1.0	-1.0	-1.1	-1.1	-1.1	-1.1	-0.9	-0.9	-0.9	-0.9	-1.1	-1.1
Demoeraely maer († Dem, 2013)	(0.1)	(0.10)	(0.1)	(0.08)	(0.1)	(0.08)	(0.2)	(0.2)	(0.2)	(0.2)	(0.1)	(0.08)
	(0.1)			able is Mean								(0.00)
Democracy Index (V-Dem, 2000)	-1.9	-2.0	-1.8	-1.9	-1.8	-1.9	-1.2	-1.4	-1.2	-1.4	-1.8	-1.9
2000)	(0.3)	(0.4)	(0.3)	(0.4)	(0.3)	(0.4)	(0.5)	(0.6)	(0.5)	(0.6)	(0.3)	(0.5)
Baseline Controls	×	<b>√</b>	X	(011) V	X	<b>v</b>	X	(0.0) ✓	X	(0.0) V	×	(0.0)
N	81	81	86	86	90	90	130	130	136	136	71	71
	51	51	00	50	20	20	150	150	150	100	, 1	71

# Table A13: 2SLS Regression Estimates of Democracy's Effects on GDP per Capita Growth

*Notes:* Panel A reports the 2SLS estimates of democracy's effect on mean GDP per capita and nighttime light intensity per capita growth rates in 2001-2019 and 2020-2022, using five different IV strategies. The Democracy Index (V-Dem) is normalized to have mean zero and standard deviation one in each year. For IVs, columns 1 and 2 use log European settler mortality, columns 3 and 4 use log population density in the 1500s, columns 5 and 6 use British legal origin, columns 7 and 8 use the fraction speaking English and the fraction speaking European, columns 9 and 10 use the ability to grow crops and mine minerals, and columns 11 and 12 use all the IVs together. Robust standard errors are in parentheses. The p-values, presented under the standard errors, are displayed as 0.00 if they are strictly smaller than the 0.005 threshold. Panel B reports the OLS estimates. Columns 1, 3, 5, 7, 9, and 11 have no controls, while columns 2, 4, 6, 8, 10, and 12 have the following baseline controls: absolute latitude, mean temperature, mean precipitation, population density, and median age. For outcomes in 2020-22, we also control for diabetes prevalence. N refers to the number of countries for which IVs, controls, and outcomes are available. Variable definitions and data sources are in Appendix Table A1.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
Panel A			Ι	Dependent V	ariable is	Mean G	DP Grow	th Rate i	n 2001-20	019		
Democracy Index (V-Dem, 2000)	-2.5	-4.2	-2.2	-3.1	-1.8	-1.7	-1.0	-1.1	-2.2	-1.6	-2.1	-2.6
	(0.3)	(2.7)	(0.4)	(0.6)	(0.5)	(1.3)	(0.9)	(0.6)	(0.5)	(0.6)	(0.2)	(0.2)
Democracy Index (Polity, 2000)	-2.8	-3.6	-2.6	-3.4	-2.0	-1.5	-1.3	-1.4	-2.8	-2.0	-2.3	-2.5
	(0.4)	(1.3)	(0.5)	(0.8)	(0.4)	(1.2)	(1.1)	(0.6)	(0.5)	(0.5)	(0.2)	(0.3)
Democracy Index (Freedom House, 2003)	-2.1	-3.1	-2.2	-3.3	-1.7	-1.4	-1.0	-1.1	-2.0	-1.5	-2.1	-2.6
	(0.3)	(0.9)	(0.4)	(0.7)	(0.5)	(1.1)	(1.0)	(0.6)	(0.4)	(0.7)	(0.3)	(0.2)
Democracy Index (Economist Intelligence Unit, 2006)	-2.7	-3.3	-2.5	-4.2	-1.9	-1.7	-1.3	-1.4	-2.3	-1.6	-2.4	-2.9
	(0.4)	(1.0)	(0.5)	(1.1)	(0.6)	(1.5)	(1.2)	(0.8)	(0.6)	(0.8)	(0.4)	(0.5)
Panel B			Ι	Dependent V	ariable is	Mean G	DP Grow	th Rate i	n 2020-20	022		
Democracy Index (V-Dem, 2019)	-0.9	-0.8	-0.9	-1.0	-0.8	-0.7	-0.7	-0.6	-1.0	-0.8	-0.9	-1.0
	(0.1)	(0.2)	(0.08)	(0.09)	(0.1)	(0.1)	(0.3)	(0.2)	(0.2)	(0.2)	(0.06)	(0.07)
Democracy Index (Polity, 2018)	-1.3	-1.0	-1.2	-1.3	-0.9	-0.8	-1.0	-0.9	-1.3	-1.1	-1.0	-1.1
	(0.2)	(0.2)	(0.2)	(0.2)	(0.2)	(0.2)	(0.3)	(0.3)	(0.2)	(0.1)	(0.06)	(0.06)
Democracy Index (Freedom House, 2019)	-1.1	-1.0	-1.1	-1.2	-0.8	-0.8	-0.8	-0.7	-1.1	-0.9	-1.0	-1.1
	(0.1)	(0.2)	(0.1)	(0.2)	(0.2)	(0.2)	(0.3)	(0.3)	(0.2)	(0.2)	(0.07)	(0.09)
Democracy Index (Economist Intelligence Unit, 2019)	-1.1	-1.0	-1.1	-1.3	-0.8	-0.8	-0.8	-0.7	-1.1	-0.9	-1.0	-1.1
	(0.1)	(0.2)	(0.1)	(0.2)	(0.2)	(0.2)	(0.3)	(0.3)	(0.2)	(0.2)	(0.08)	(0.1)
Panel C			Dependent	Variable is	Mean Nig	ghttime L	ight Inter	nsity Gro	wth Rate	in 2001-201	3	
Democracy Index (V-Dem, 2000)	-2.8	-5.3	-1.8	-1.3	-2.5	-3.2	-1.6	-3.0	-2.3	-2.7	-2.0	-2.0
• · · ·	(0.7)	(3.8)	(0.4)	(0.6)	(0.6)	(1.0)	(1.1)	(1.0)	(0.5)	(0.5)	(0.2)	(0.4)
Democracy Index (Polity, 2000)	-3.1	-4.5	-2.1	-1.5	-2.7	-2.8	-2.2	-4.0	-2.9	-3.0	-2.2	-2.1
	(0.8)	(2.7)	(0.4)	(0.5)	(0.5)	(0.7)	(1.3)	(1.3)	(0.7)	(0.6)	(0.2)	(0.4)
Democracy Index (Freedom House, 2003)	-2.4	-4.0	-1.8	-1.5	-2.3	-2.7	-1.7	-3.3	-2.0	-2.7	-2.0	-2.2
	(0.4)	(1.9)	(0.4)	(0.6)	(0.3)	(0.5)	(1.1)	(1.0)	(0.4)	(0.6)	(0.2)	(0.3)
Democracy Index (Economist Intelligence Unit, 2006)	-3.1	-4.0	-2.1	-1.8	-2.6	-3.2	-2.1	-3.9	-2.1	-2.4	-2.2	-2.3
	(0.6)	(1.8)	(0.5)	(0.7)	(0.5)	(0.8)	(1.4)	(1.6)	(0.6)	(0.6)	(0.4)	(0.5)
IVs	settler	mortality	population	on density	legal	origin	lang	uage	crops &	t minerals	all	IVs
Baseline Controls	×	✓	X	1	x	1	X	-	X	✓	X	1
N	77	77	82	82	86	86	120	120	124	124	68	68

#### Table A14: 2SLS Regression with Alternative Democracy Indices

*Notes*: This table compares the results of 2SLS regressions on the mean GDP growth rate in 2001-2019 (Panel A), the mean GDP growth rate in 2020-2022 (Panel B), and the mean nightime light intensity growth rate in 2001-2013 (Panel C) using four different democracy indices by V-Dem, Polity, Freedom House, and the Economist Intelligence Unit. When data for the democracy index does not exist for the baseline year, we use the value from the closest year. We normalize all indices to have mean zero and standard deviation one. Columns 1, 3, 5, 7, 9, and 11 have no controls, while columns 2, 4, 6, 8, 10, and 12 have the following baseline controls: absolute latitude, mean temperature, mean precipitation, population density, and median age. For the mean GDP growth rate in 2020-2022, we also control for diabetes prevalence. For IVs, columns 1 and 2 use log European settler mortality, columns 3 and 4 use log population density in the 1500s, columns 5 and 6 use British legal origin, columns 7 and 8 use the fraction speaking English and the fraction speaking European, columns 9 and 10 use the ability to grow crops and mine minerals, and columns 11 and 12 use all the IVs together. The estimates in this table are slightly different from those in Table 1 because this table uses only observations for which all of the democracy indices are available. N refers to the number of countries for which IVs, controls, and outcomes are available. Robust standard errors are in parentheses. Variable definitions and data sources are in Appendix Table A1.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
Panel A: Two-Stage Least Squa	ares											
			De	pendent Vari	able is N	lean GD	P Growt	h Rate ii	n 2001-20	07		
Democracy Index (V-Dem, 2000)	-2.8	-5.5	-2.3	-3.2	-2.2	-2.6	-1.5	-1.7	-2.4	-2.1	-2.4	-2.9
-	(0.4)	(3.9)	(0.5)	(0.7)	(0.6)	(1.3)	(1.0)	(0.6)	(0.5)	(0.6)	(0.3)	(0.4)
	0.00	0.17	0.00	0.00	0.00	0.06	0.12	0.00	0.00	0.00	0.00	0.00
			De	pendent Vari	able is N	lean GD	P Growt	h Rate ii	n 2008-20	09		
Democracy Index (V-Dem, 2007)	-3.9	-3.9	-3.3	-3.7	-3.2	-2.9	-2.1	-2.0	-3.6	-2.9	-3.2	-3.8
-	(0.6)	(1.1)	(0.5)	(0.5)	(0.6)	(0.9)	(0.9)	(0.9)	(0.6)	(0.7)	(0.2)	(0.2)
	0.00	0.00	0.00	0.00	0.00	0.00	0.02	0.02	0.00	0.00	0.00	0.00
			De	pendent Vari	able is N	lean GD	P Growt	h Rate ii	n 2010-20	19		
Democracy Index (V-Dem, 2009)	-1.7	-1.6	-1.8	-2.3	-1.4	-1.2	-1.1	-0.9	-1.8	-1.3	-1.6	-1.9
• • •	(0.2)	(0.2)	(0.3)	(0.4)	(0.3)	(0.5)	(0.5)	(0.4)	(0.3)	(0.4)	(0.1)	(0.1)
	0.00	0.00	0.00	0.00	0.00	0.01	0.02	0.04	0.00	0.00	0.00	0.00
IVs	settler	mortality	populat	ion density	legal	origin	lang	uage	crops &	k minerals	all	IVs
Number of IVs	1	1	1	1	1	1	2	2	10	10	15	15
Observations	81	81	86	86	90	90	132	132	138	138	71	71
Panel B: Ordinary Least Squar	es											
			De	pendent Vari	able is N	Iean GD	P Growt	h Rate iı	n 2001-20	07		
Democracy Index (V-Dem, 2000)	-2.3	-2.2	-2.2	-2.4	-2.2	-2.4	-1.9	-1.5	-1.9	-1.5	-2.3	-2.4
•	(0.4)	(0.5)	(0.4)	(0.5)	(0.4)	(0.5)	(0.5)	(0.6)	(0.5)	(0.6)	(0.4)	(0.5)
			De	pendent Vari	able is N	lean GD	P Growt	h Rate ii	n 2008-20	09		
Democracy Index (V-Dem, 2007)	-2.9	-2.7	-2.9	-3.1	-2.9	-3.1	-2.4	-1.8	-2.4	-1.7	-3.0	-3.2
•	(0.3)	(0.5)	(0.3)	(0.4)	(0.3)	(0.4)	(0.7)	(0.9)	(0.7)	(0.9)	(0.3)	(0.4)
			De	pendent Vari	able is N	Iean GD	P Growt	h Rate ii	n 2010-20	19		
Democracy Index (V-Dem, 2009)	-1.6	-1.5	-1.5	-1.6	-1.5	-1.6	-1.4	-1.0	-1.3	-1.0	-1.6	-1.7
	(0.1)	(0.2)	(0.1)	(0.2)	(0.1)	(0.2)	(0.3)	(0.4)	(0.3)	(0.4)	(0.1)	(0.1)
Baseline Controls	X	1	X	1	X	1	X	1	×	1	X	1
Ν	81	81	86	86	90	90	132	132	138	138	71	71

Table A15: 2SLS Regression Estimates of Democracy's Effects Before, During, and After the Great Recession

*Notes:* Panel A reports the 2SLS estimates of democracy's effect on mean GDP growth rates in 2001-7, 2008-9, and 2010-19, using five different IV strategies. The Democracy Index (V-Dem) is normalized to have mean zero and standard deviation one in each year. For IVs, columns 1 and 2 use log European settler mortality, columns 3 and 4 use log population density in the 1500s, columns 5 and 6 use British legal origin, columns 7 and 8 use the fraction speaking English and the fraction speaking European, columns 9 and 10 use the ability to grow crops and mine minerals, and columns 11 and 12 use all the IVs together. Robust standard errors are in parentheses. The p-values are displayed as 0.00 if they are strictly smaller than the 0.005 threshold. Panel B reports the OLS estimates. Columns 1, 3, 5, 7, 9, and 11 have no controls, while columns 2, 4, 6, 8, 10, and 12 have the following baseline controls: absolute latitude, mean temperature, mean precipitation, population density, and median age. N refers to the number of countries for which IVs, controls, and outcomes are available. Variable definitions and data sources are in Appendix Table A1.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
Panel A				Dependent	Variable is	Mean GE	P Growth	n Rate in	2001-201	9		
Democracy Index (V-Dem, 2000)	-2.5	-4.1	-2.2	-3.1	-1.8	-1.8	-1.0	-1.1	-2.2	-1.6	-2.1	-2.6
	(0.3)	(2.7)	(0.4)	(0.6)	(0.5)	(1.3)	(0.9)	(0.6)	(0.5)	(0.6)	(0.2)	(0.2)
Include US & China?	1	<b>√</b>	<i>✓</i>	✓ 	1	1	1	1	1	1	<i>✓</i>	<i>_</i>
N	82	82	87	87	91	91	133	133	139	139	72	72
Democracy Index (V-Dem, 2000)	-2.6	4.0	-1.9	-3.9	3.9	-15.6	-0.4	-0.7	-1.6	-0.3	-1.4	-0.9
	(0.8)	(6.7)	(1.0)	(3.6)	(8.6)	(44.9)	(1.2)	(0.6)	(0.5)	(0.9)	(0.4)	(0.7)
Include US & China?	×	×	X	×	X	X	X	×	×	X	X	X
N	80	80	85	85	89	89	131	131	137	137	70	70
Panel B				Dependent	Variable is	s Mean GE	P Growth	n Rate in	2020-202	2		
Democracy Index (V-Dem, 2019)	-0.9	-0.8	-0.9	-1.0	-0.8	-0.7	-0.7	-0.6	-1.0	-0.8	-0.9	-1.0
	(0.1)	(0.2)	(0.08)	(0.09)	(0.1)	(0.1)	(0.3)	(0.2)	(0.2)	(0.2)	(0.06)	(0.07
Include US & China?	1	1	1	✓	1	1	1	1	1	1	1	1
N	82	82	87	87	91	91	133	133	139	139	72	72
Democracy Index (V-Dem, 2019)	-0.9	-0.6	-1.0	-1.7	9.3	17.6	-0.3	-0.3	-0.6	-0.3	-1.0	-1.5
	(0.3)	(1.4)	(0.3)	(1.0)	(52.7)	(74.9)	(0.5)	(0.4)	(0.4)	(0.6)	(0.3)	(0.7)
Include US & China?	×	×	X	×	×	X	X	×	×	×	X	X
N	80	80	85	85	89	89	131	131	137	137	70	70
Panel C			Depender	nt Variable is	Mean Nig	ghttime Li	ght Intens	ity Grow	th Rate in	2001-2013		
Democracy Index (V-Dem, 2000)	-2.8	-5.2	-1.8	-1.4	-2.5	-3.2	-1.6	-3.0	-2.3	-2.7	-2.0	-2.0
	(0.6)	(3.7)	(0.4)	(0.6)	(0.6)	(1.0)	(1.1)	(1.0)	(0.5)	(0.5)	(0.2)	(0.4)
Include US & China?	1	1	1	1	1	1	1	1	1	1	1	~
N	81	81	86	86	90	90	130	130	136	136	71	71
Democracy Index (V-Dem, 2000)	-3.5	20.1	-0.07	0.4	-1.5	0.6	-0.09	-2.0	-1.8	-0.4	-0.5	-0.5
	(1.3)	(136.3)	(0.5)	(0.7)	(3.6)	(2.6)	(2.4)	(1.3)	(0.8)	(1.3)	(0.5)	(0.6)
Include US & China?	×	×	x	×	X	X	×	×	×	X	X	×
N	79	79	84	84	88	88	128	128	134	134	69	69
IVs	settler	mortality	populati	on density	legal	origin	lang	uage	crops &	t minerals	all	IVs
Controls	×	✓	X	✓	x	-	x	-	x	1	X	1

## Table A16: 2SLS Regression Excluding the US and China

*Notes:* This table compares the results of 2SLS regressions under two sample definitions (include the US and China vs. exclude the US and China). The Democracy Index (V-Dem) is normalized to have mean zero and standard deviation one in each year. Columns 1, 3, 5, 7, 9, and 11 have no controls, while columns 2, 4, 6, 8, 10, and 12 have the following baseline controls: absolute latitude, mean temperature, mean precipitation, population density, and median age. For the mean GDP growth rate in 2020-2022, we also control for diabetes prevalence. For IVs, columns 1 and 2 use log European settler mortality, columns 3 and 4 use log population density in the 1500s, columns 5 and 6 use British legal origin, columns 7 and 8 use the fraction speaking English and the fraction speaking European, columns 9 and 10 use the ability to grow crops and mine minerals, and columns 11 and 12 use all the IVs together. N refers to the number of countries for which IVs, controls, and outcomes are available. Robust standard errors are in parentheses. Variable definitions and data sources are in Appendix Table A1.

IVs Baseline Controls	settler 1 X	nortality	populati X	on density	legal X	origin 1	lang X	uage	crops &	minerals	all X	IVs
N	80	80	82	82	86	86	125	125	130	130	70	70
	(0.6)	(0.6)	(0.4)	(0.4)	(0.6)	(0.6)	(1.1)	(1.1)	(0.5)	(0.5)	(0.2)	(0.2
Democracy Index (V-Dem, 2000)	-2.8	-2.8	-1.7	-1.7	-2.4	-2.4	-1.5	-1.5	-2.2	-2.2	-1.9	-1.9
			Dependent	Variable is	Mean Nig	ghttime L	ight Inter	sity Gro	wth Rate i	n 2001-201	3	
N	81	81	84	85	88	89	128	128	133	134	71	71
	(0.1)	(0.1)	(0.07)	(0.08)	(0.1)	(0.1)	(0.3)	(0.3)	(0.1)	(0.1)	(0.06)	(0.00
Democracy Index (V-Dem, 2019)	-0.9	-0.9	-0.9	-0.9	-0.8	-0.8	-0.7	-0.7	-1.0	-1.0	-0.9	-0.9
			I	Dependent V	ariable is	Mean G	DP Grow	th Rate in	n 2020-20	22		
N	80	81	81	84	85	84	123	122	128	129	69	71
	(0.3)	(0.3)	(0.4)	(0.4)	(0.4)	(0.4)	(1.2)	(1.0)	(0.4)	(0.4)	(0.3)	(0.3
Democracy Index (V-Dem, 2000)	-2.3	-2.3	-1.9	-1.8	-1.9	-1.9	0.3	-0.8	-1.9	-2.0	-2.0	-2.0
			Ι	Dependent V	ariable is	Mean G	DP Grow	th Rate in	n 2001-20	19		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12

#### Table A17: 2SLS Regression Excluding Outliers

*Notes:* This table shows the 2SLS regression estimates on the effect of democracy on the mean GDP growth rate in 2001-2019, the mean GDP growth rate in 2020-2022, and the mean nighttime light intensity growth rate in 2001-2013 excluding countries with a standardized residual above 1.96 or below -1.96. For each 2SLS regression, we run the baseline specification, calculate the fitted values, use the fitted values to calculate the residual in the second stage regression, standardize the residuals to have mean zero and variance one, and finally rerun the 2SLS regression with the sample limited to countries that have a standardized residual between -1.96 and 1.96. The Democracy Index (V-Dem) is normalized to have mean zero and standard deviation one in each year. Columns 1, 3, 5, 7, 9, and 11 have no controls, while columns 2, 4, 6, 8, 10, and 12 have the following baseline controls: absolute latitude, mean temperature, mean precipitation, population density, and median age. For the mean GDP growth rate in 2020-2022, we also control for diabetes prevalence. For IVs, columns 1 and 2 use log European settler mortality, columns 3 and 4 use log population density in the 1500s, columns 5 and 6 use British legal origin, columns 7 and 8 use the fraction speaking English and the fraction speaking European, columns 9 and 10 use the ability to grow crops and mine minerals, and columns 11 and 12 use all the IVs together. N refers to the number of countries for which IVs, controls, and outcomes are available. Robust standard errors are in parentheses. Variable definitions and data sources are in Appendix Table A1.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
			De	ependent Va	iable is l	Mean GDI	P Growtl	n Rate in	2001-201	19		
Democracy Index (V-Dem, 2000)	-2.1	-3.9	-2.2	-3.1	-0.4	-20.0	-2.0	-2.7	-1.7	-1.1	-2.1	-2.1
	(0.5)	(9.0)	(0.5)	(1.3)	(2.3)	(73.0)	(0.6)	(0.8)	(0.6)	(0.5)	(0.3)	(0.4
			De	ependent Va	iable is l	Mean GDI	P Growth	n Rate in	2020-202	22		
Democracy Index (V-Dem, 2019)	-0.8	0.2	-1.1	-1.2	-0.7	6.0	-0.7	-0.8	-0.9	-0.7	-1.0	-1.1
	(0.3)	(0.7)	(0.2)	(0.3)	(0.4)	(18.1)	(0.3)	(0.4)	(0.3)	(0.3)	(0.1)	(0.3
		De	ependent V	/ariable is M	ean Nigh	ttime Lig	ht Intens	ity Grow	th Rate in	n 2001-2013	3	
Democracy Index (V-Dem, 2000)	-2.0	-6.9	-1.1	-0.6	-2.6	-10.0	-1.9	-2.9	-0.6	-1.3	-1.6	-1.1
-	(0.5)	(12.0)	(0.5)	(0.5)	(1.5)	(41.3)	(0.7)	(0.8)	(0.7)	(0.5)	(0.4)	(0.5)
IVs	settler	mortality	populat	ion density	legal	origin	lang	uage	crops &	t minerals	all	IVs
Baseline Controls	X	1	x	1	x	1	X	1	x	1	X	1
N	77	77	83	83	87	87	121	121	127	127	69	69

#### Table A18: 2SLS Regression Excluding G7 Countries

*Notes:* This table shows the 2SLS regression estimates of democracy's effect on the mean GDP growth rate in 2001-2019, the mean GDP growth rate in 2020-2022, and the mean nighttime light intensity growth rate in 2001-2013 that exclude G7 countries (Canada, France, Germany, Italy, Japan, the United Kingdom, and the United States) from the sample. The Democracy Index (V-Dem) is normalized to have mean zero and standard deviation one in each year. Columns 1, 3, 5, 7, 9, and 11 have no controls, while columns 2, 4, 6, 8, 10, and 12 have the following baseline controls: absolute latitude, mean temperature, mean precipitation, population density, and median age. For the mean GDP growth rate in 2020-2022, we also control for diabetes prevalence. For IVs, columns 1 and 2 use log European settler mortality, columns 3 and 4 use log population density in the 1500s, columns 5 and 6 use British legal origin, columns 7 and 8 use the fraction speaking English and the fraction speaking European, columns 9 and 10 use the ability to grow crops and mine minerals, and columns 11 and 12 use all the IVs together. N refers to the number of countries for which IVs, controls, and outcomes are available. Robust standard errors are in parentheses. Variable definitions and data sources are in Appendix Table A1.

Baseline Controls N	× 81	✓ 81	× 85	✓ 85	<b>×</b> 89	✓ 89	<b>×</b> 128	✓ 128	<b>×</b> 134	✓ 134	<b>x</b> 71	✓ 71
IVs		mortality		on density		origin	•	uage	-	& minerals		IVs
	(1.1)	(9.1)	(1.1)	(1.9)	(10.0)	(28.5)	(0.6)	(1.2)	(0.9)	(2.1)	(0.8)	(2.3)
Democracy Index (Weighting: None)	-2.9	-1.5	-2.5	-2.4	-13.2	-21.3	-3.1	-4.1	-2.9	-3.2	-2.9	-3.8
	(0.8)	(1.6)	(0.8)	(1.2)	(1.2)	(1.0)	(0.5)	(0.9)	(0.6)	(0.9)	(0.5)	(1.4)
Democracy Index (Weighting: Population)	-1.7	-1.5	-1.5	-3.0	-2.9	-3.4	-2.2	-3.2	-1.9	-1.6	-2.1	-3.1
	(0.6)	(3.7)	(0.4)	(0.6)	(0.6)	(1.0)	(1.1)	(1.0)	(0.5)	(0.5)	(0.2)	(0.4)
Democracy Index (Weighting: GDP)	-2.8	-5.2	-1.8	-1.3	-2.5	-3.2	-1.6	-3.0	-2.3	-2.7	-2.0	-2.0
Panel C		D	ependent V	/ariable is N	Iean Nigł	nttime Lig	t Inten	sity Gro	wth Rate	in 2001-20	13	
	(1.5)	(56.0)	(1.5)	(2.8)	(4.8)	(4.4)	(1.3)	(2.0)	(0.5)	(1.3)	(1.5)	(2.5)
Democracy Index (Weighting: None)	-0.2	18.3	0.7	1.8	3.1	3.5	1.1	2.3	-0.6	0.4	0.8	2.2
	(0.4)	(0.4)	(0.3)	(0.3)	(0.7)	(0.5)	(0.2)	(0.2)	(0.4)	(0.3)	(0.10)	(0.2)
Democracy Index (Weighting: Population)	-0.8	-0.5	-1.3	-1.2	-0.4	-0.4	-0.9	-0.6	-1.6	-1.1	-1.1	-0.9
(	(0.1)	(0.2)	(0.08)	(0.09)	(0.1)	(0.1)	(0.3)	(0.2)	(0.2)	(0.2)	(0.06)	(0.07)
Democracy Index (Weighting: GDP)	-0.9	-0.8	-0.9	-1.0	-0.8	-0.7	-0.7	-0.6	-1.0	-0.8	-0.9	-1.0
Panel B			De	ependent Va	riable is l	Mean GD	P Growt	h Rate i	n 2020-2	022		
	(0.4)	(4.0)	(0.3)	(0.5)	(2.1)	(5.8)	(0.2)	(0.4)	(0.3)	(0.6)	(0.3)	(0.5)
Democracy Index (Weighting: None)	-1.0	-0.2	-1.2	-1.3	-2.2	-3.8	-1.1	-1.1	-1.2	-0.4	-1.3	-1.5
	(0.6)	(0.8)	(1.7)	(1.3)	(0.9)	(0.8)	(0.6)	(0.6)	(0.8)	(0.6)	(0.4)	(0.4)
Democracy Index (Weighting: Population)	-1.7	-2.0	-3.4	-4.0	-1.1	-1.2	-2.7	-2.5	-2.9	-1.5	-1.6	-1.9
Democracy index (weighting, ODI)	(0.3)	(2.7)	(0.4)	(0.6)	(0.5)	(1.3)	(0.9)	(0.6)	(0.5)	(0.6)	(0.2)	(0.2)
Democracy Index (Weighting: GDP)	-2.5	-4.1	-2.2	-3.1	-1.8	-1.8	-1.0	-1.1	-2.2	-1.6	-2.1	-2.6
Panel A	(1)	(-)		ependent Va				. /		~ /	(11)	(1-)
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)

Table A19: 2SLS Regression with Alternative Weightings

*Notes:* This table compares the results of 2SLS regressions on the mean GDP growth rate in 2001-2019 (Panel A), the mean GDP growth rate in 2020-2022 (Panel B), and the mean nighttime light intensity growth rate in 2001-2013 (Panel C) with weighting of observations by baseline GDP, weighting by baseline population, and no weighting. The Democracy Index (V-Dem) is normalized to have mean zero and standard deviation one in each year. Columns 1, 3, 5, 7, 9, and 11 have no controls, while columns 2, 4, 6, 8, 10, and 12 have the following baseline controls: absolute latitude, mean temperature, mean precipitation, population density, and median age. For the mean GDP growth rate in 2020-2022, we also control for diabetes prevalence. For IVs, columns 1 and 2 use log European settler mortality, columns 3 and 4 use log population density in the 1500s, columns 5 and 6 use British legal origin, columns 7 and 8 use the fraction speaking English and the fraction speaking European, columns 9 and 10 use the ability to grow crops and mine minerals, and columns 11 and 12 use all the IVs together. N refers to the number of countries for which IVs, controls, and outcomes are available. Robust standard errors are in parentheses. Variable definitions and data sources are in Appendix Table A1.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
Panel A: No Control for Baseline GDP												
				Dependent V	/ariable is	s Mean GI	OP Grow	h Rate ir	2001-201	19		
Democracy Index (V-Dem, 2000)	-2.5	-4.2	-2.2	-3.1	-1.8	-1.8	-1.0	-1.1	-2.2	-1.6	-2.1	-2.6
	(0.3)	(2.7)	(0.4)	(0.6)	(0.5)	(1.3)	(1.0)	(0.6)	(0.5)	(0.6)	(0.2)	(0.2)
				Dependent V	Variable is	s Mean GI	OP Grow	h Rate ir	2020-202	22		
Democracy Index (V-Dem, 2019)	-0.9	-0.8	-0.9	-1.0	-0.8	-0.7	-0.7	-0.6	-1.0	-0.8	-0.9	-1.0
	(0.1)	(0.2)	(0.08)	(0.09)	(0.2)	(0.1)	(0.3)	(0.2)	(0.2)	(0.2)	(0.06)	(0.07)
Panel B: Control for Baseline GDP Per Cap	ita											
-				Dependent V	/ariable is	s Mean GI	OP Grow	h Rate ir	2001-201	19		
Democracy Index (V-Dem, 2000)	-3.1	1.9	-2.1	-3.0	1.4	19.1	-1.6	-0.8	-2.1	-1.1	-2.1	-2.1
	(0.8)	(6.5)	(0.6)	(0.9)	(5.1)	(86.0)	(0.6)	(0.5)	(0.8)	(0.6)	(0.3)	(0.3)
				Dependent V	/ariable is	s Mean GI	OP Grow	h Rate ir	2020-202	22		
Democracy Index (V-Dem, 2019)	-1.3	-1.3	-1.2	-1.5	-0.8	8.9	-1.2	-1.0	-1.5	-1.6	-1.1	-1.3
	(0.2)	(5.7)	(0.1)	(0.2)	(0.4)	(35.8)	(0.3)	(0.4)	(0.2)	(0.4)	(0.10)	(0.2)
Panel C: Control for Baseline Total GDP												
				Dependent V	/ariable is	s Mean GI	OP Grow	h Rate ir	2001-201	19		
Democracy Index (V-Dem, 2000)	-2.6	-7.1	-2.2	-3.1	-1.3	-0.3	-1.2	-0.9	-2.4	-1.6	-2.1	-2.5
-	(0.3)	(12.7)	(0.5)	(0.6)	(1.2)	(2.6)	(1.1)	(0.6)	(0.5)	(0.8)	(0.2)	(0.3)
				Dependent V	/ariable is	s Mean GI	OP Grow	h Rate ir	2020-202	22		
Democracy Index (V-Dem, 2019)	-1.0	-0.9	-1.0	-1.1	-0.9	-0.8	-0.8	-0.9	-1.0	-1.0	-1.0	-1.0
- · · · · · · · · · · · · · · · · · · ·	(0.07)	(0.2)	(0.06)	(0.09)	(0.1)	(0.2)	(0.2)	(0.2)	(0.1)	(0.1)	(0.04)	(0.08)
IVs	settler r	nortality	populati	on density	legal	origin	lang	uage	crops &	z minerals	all	IVs
Baseline Controls Other Than Baseline GDP	X	1	X	✓	×	1	×	<u>َر</u>	x	1	×	1
Ν	81	81	86	86	90	90	131	131	137	137	71	71

#### Table A20: Democracy's Effect on Economic Growth With Control for Baseline GDP

*Notes:* This table compares the 2SLS regression estimates of democracy's effect on the mean GDP growth rate in 2001-2019 and the mean GDP growth rate in 2020-2022 without controls for baseline GDP (Panel A), with additional controls for baseline GDP per capita (Panel B), and with additional controls for baseline total GDP (Panel C). The Democracy Index (V-Dem) is normalized to have mean zero and standard deviation one in each year. Columns 2, 4, 6, 8, 10, and 12 also have the following baseline controls: absolute latitude, mean temperature, mean precipitation, population density, and median age. For the mean GDP growth rate in 2020-2022, we also control for diabetes prevalence. For IVs, columns 1 and 2 use log European settler mortality, columns 3 and 4 use log population density in the 1500s, columns 5 and 6 use British legal origin, columns 7 and 8 use the fraction speaking European, columns 9 and 10 use the ability to grow crops and mine minerals, and columns 11 and 12 use all the IVs together. The sample sizes are slightly different from those in Table 1 because this table uses only observations for which all GDP per capita and total GDP data are available. N refers to the number of countries for which IVs, controls, and outcomes are available. Robust standard errors are in parentheses. Variable definitions and data sources are in Appendix Table A1.

	Table	e A21:	23L3 K	kegressioi	n with	Contin	ient Co	ontrois				
	(1)			(4)	(5)	(6)	(7)	(0)	(0)	(10)	(11)	(10)
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
			I	Dependent Va	riable is	Mean GD	P Growtl	n Rate in	2001-2019	9		
Democracy Index (V-Dem, 2000)	-0.9	-0.7	-0.8	-1.6	-0.8	-0.8	1.5	-6.4	-2.8	-1.8	-1.1	-0.8
	(0.9)	(3.1)	(0.8)	(0.5)	(0.5)	(0.5)	(1.5)	(6.4)	(0.5)	(0.6)	(0.4)	(0.3)
			Ι	Dependent Va	riable is	Mean GD	P Growtl	n Rate in	2020-2022	2		
Democracy Index (V-Dem, 2019)	-0.4	-0.3	-0.9	-1.0	-0.4	-0.02	0.7	-6.6	-1.4	-0.5	-0.9	-0.7
	(0.7)	(0.6)	(0.9)	(0.6)	(0.4)	(0.4)	(1.8)	(3.9)	(0.2)	(0.4)	(0.2)	(0.3)
		I	Dependent	Variable is N	Iean Nig	httime Lig	ght Intens	ity Grow	th Rate in	2001-2013		
Democracy Index (V-Dem, 2000)	-1.5	-3.9	0.6	-0.1	-1.6	-2.0	-8.7	5.1	-2.8	-1.7	-1.7	-0.9
	(0.9)	(3.3)	(1.2)	(0.7)	(0.6)	(1.0)	(9.8)	(6.2)	(0.5)	(1.3)	(0.4)	(0.4)
IVs	settler	mortality	populat	ion density	legal	origin	lang	uage	crops &	minerals	all	IVs
Baseline Controls	X	1	X	1	x	-	×	-	x	1	×	1
N	81	81	85	85	89	89	128	128	134	134	71	71

Table A21: 2SLS Regression with Continent Controls

*Notes:* This table shows the 2SLS regression estimates of democracy's effect on the mean GDP growth rate in 2001-2019, the mean GDP growth rate in 2020-2022, and the mean nighttime light intensity growth rate in 2001-2013 that adds dummy variables for each continent (Africa, Asia, Europe, North America, Oceania, and South America) as controls. The Democracy Index (V-Dem) is normalized to have mean zero and standard deviation one in each year. Columns 1, 3, 5, 7, 9, and 11 only control for continents, while columns 2, 4, 6, 8, 10, and 12 also have the following baseline controls: absolute latitude, mean temperature, mean precipitation, population density, and median age. For the mean GDP growth rate in 2020-2022, we also control for diabetes prevalence. For IVs, columns 1 and 2 use log European settler mortality, columns 3 and 4 use log population density in the 1500s, columns 5 and 6 use British legal origin, columns 7 and 8 use the fraction speaking English and the fraction speaking European, columns 9 and 10 use the ability to grow crops and mine minerals, and columns 11 and 12 use all the IVs together. N refers to the number of countries for which IVs, controls, and outcomes are available. Robust standard errors are in parentheses. Variable definitions and data sources are in Appendix Table A1.



## Figure A3: Correlation Between Democracy and Economic Growth by Decade

*Notes:* Panels (a)-(d) show the relationship between democracy and the mean GDP growth rates in four periods: 1981-1990, 1991-2000, 2001-2010, and 2011-2019. The Democracy Index (V-Dem) is normalized to have mean zero and standard deviation one in each year. The size of each circle (country) is proportional to its baseline GDP. The colors depend on the level of the democracy index (warmer colors for democracy and darker colors for autocracies). The line is the fitted line from a univariate OLS regression of the outcome against the democracy index that weights observations by baseline GDP. The shaded area corresponds to the 95% confidence interval. Variable definitions and data sources are in Appendix Table A1.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
Panel A: No Control for Baseline GDP									. ,			. ,
				Dependent	Variable is	Mean GDI	P Growth 1	Rate in 19	981-1990			
Democracy Index (V-Dem, 1980)	-0.6	-0.7	-1.1	151.6	-0.09	0.09	-0.7	-0.6	-0.8	-0.2	-0.2	-0.10
	(0.5)	(1.3)	(0.7)	(2814.7)	(0.6)	(0.8)	(0.8)	(0.9)	(0.5)	(0.8)	(0.5)	(0.8)
				Dependent	Variable is	Mean GDI	P Growth 1	Rate in 19	991-2000			
Democracy Index (V-Dem, 1990)	-1.6	-66.2	-1.2	-5.1	-0.6	-3.5	0.3	0.4	-0.9	0.007	-1.0	-1.9
	(0.5)	(554.8)	(0.5)	(2.6)	(0.8)	(5.6)	(1.2)	(0.9)	(0.5)	(1.0)	(0.6)	(0.7)
				Dependent								• •
Democracy Index (V-Dem, 2000)	-3.2	-5.7	-2.6	-3.2	-2.5	-3.0	-1.5	-1.7	-2.8	-2.4	-2.5	-3.0
	(0.5)	(4.1)	(0.5)	(0.7) Dependent	(0.6) Variabla ia	(1.5) Maan GDI	(1.1) P Growth 1	(0.7)	(0.5)	(0.6)	(0.3)	(0.4)
Democratic Index (V Dem. 2010)	-1.7	-1.5	-1.8	-2.1	-1.2		-1.2		-1.9	-1.5	-1.6	-1.7
Democracy Index (V-Dem, 2010)	-1./	-1.5 (0.1)	-1.8 (0.3)	-2.1 (0.3)	-1.2 (0.4)	-1.1 (0.5)	-1.2 (0.5)	-1.1 (0.3)	-1.9 (0.3)	-1.5 (0.2)	-1.6 (0.09)	-1.7 (0.1)
	. ,	(0.1)	(0.3)	(0.3)	(0.4)	(0.5)	(0.5)	(0.3)	(0.3)	(0.2)	(0.09)	(0.1)
Panel B: Control for Baseline GDP Per Cap	ita											
				Dependent	Variable is	Mean GDI		Rate in 19	981-1990			
Democracy Index (V-Dem, 1980)	-0.1	0.6	6.4	4.4	1.0	1.0	-0.2	-0.1	-0.05	0.2	0.9	0.9
	(1.2)	(1.0)	(6.5)	(3.1)	(0.6)	(0.4)	(1.2)	(0.8)	(0.6)	(0.6)	(0.3)	(0.3)
				Dependent								
Democracy Index (V-Dem, 1990)	-9.0	3.0	-3.0	-4.6	2.9	16.2	-0.07	0.5	-0.2	0.0006	-1.7	-1.6
	(7.0)	(5.0)	(1.4)	(2.5) Dependent	(17.2) Variable ia	(97.0)	(1.1) D Crowth 1	(0.7) Poto in 20	(0.6)	(0.8)	(0.8)	(0.5)
		2.0	2.0	•								1.0
Democracy Index (V-Dem, 2000)	-3.8 (1.1)	2.0 (5.6)	-2.0 (0.7)	-2.7 (0.8)	0.1 (4.4)	36.7 (368.2)	-2.0 (0.7)	-1.1 (0.4)	-2.5 (0.8)	-1.7 (0.6)	-2.2 (0.5)	-1.9 (0.3)
	(1.1)	(3.0)	(0.7)	Dependent						(0.0)	(0.5)	(0.3)
Democracy Index (V-Dem, 2010)	-2.2	-1.9	-2.6	-4.3	0.9	25.0	-1.4	-0.9	-2.2	-1.6	-1.7	-1.8
Democracy matex (V-Dem, 2010)	(1.2)	(2.7)	(0.7)	(1.6)	(2.8)	(115.5)	(0.6)	(0.4)	(0.5)	(0.4)	(0.3)	(0.6)
	(112)	(2.7)	(0.7)	(110)	(2:0)	(11010)	(0.0)	(0.1)	(0.0)	(011)	(0.5)	(0.0)
Panel C: Control for Baseline Total GDP				Dependent	Variable is	Moon CDI	Crowth	Poto in 10	081 1000			
Democratic Index (V Demo	0.0	0.0	2.4	-14.9			-4.0	-2.9	-0.7	0.4	0.4	0.2
Democracy Index (V-Dem, 1980)	-0.8 (0.5)	-0.8 (1.3)	-3.4 (2.9)	-14.9 (22.0)	-0.1 (0.9)	0.03 (0.9)	-4.0 (4.1)	-2.9	-0.7	-0.4 (1.1)	-0.4 (0.7)	-0.2 (0.8)
	(0.5)	(1.5)	(2.9)	Dependent						(1.1)	(0.7)	(0.8)
Democracy Index (V-Dem, 1990)	-1.8	-114.9	-1.7	-5.1	-0.7	-3.2	-0.6	0.5	-1.2	-0.03	-1.5	-2.1
Democracy macx (V-Dem, 1990)	(0.5)	(1645.4)	(0.7)	(2.2)	(1.3)	(6.0)	(1.1)	(0.9)	(0.5)	(1.1)	(0.6)	(0.7)
	(0.5)	(1015.1)	(0.7)	Dependent	· · ·	· · ·	· · ·	· · ·	· · ·	(1.1)	(0.0)	(0.7)
Democracy Index (V-Dem, 2000)	-3.2	-9.4	-2.4	-3.2	-2.1	-1.4	-1.5	-1.0	-2.9	-2.3	-2.4	-2.7
Democracy mach (+ Dem, 2000)	(0.5)	(17.8)	(0.6)	(0.7)	(1.2)	(2.9)	(1.3)	(0.7)	(0.5)	(0.8)	(0.4)	(0.4)
	()	<pre></pre>	····/	Dependent						····	··· /	<u></u>
Democracy Index (V-Dem, 2010)	-1.7	-1.8	-2.0	-2.3	-1.1	-0.9	-1.4	-1.3	-2.0	-2.0	-1.7	-2.0
, ,	(0.2)	(0.3)	(0.3)	(0.4)	(0.6)	(0.7)	(0.5)	(0.5)	(0.3)	(0.2)	(0.1)	(0.2)
IVs	settler	r mortality	popula	ation density	legal	origin	lang	uage	crops &	minerals	all	IVs
Baseline Controls Other Than Baseline GDP	X	✓	X	✓	X	√	X	~ ~	X	1	X	1

# Table A22: Democracy's Effect on Economic Growth by Decade

*Notes*: This table shows the 2SLS regression estimates of democracy's effect on mean GDP growth rates in 1981-1990, 1991-2000, 2001-2010, and 2011-2019. Panel A does not control for baseline GDP. Panel B controls for baseline GDP per capita. Panel C controls for baseline total GDP. The Democracy Index (V-Dem) is normalized to have mean zero and standard deviation one in each year. Columns 2, 4, 6, 8, 10, and 12 also have the following controls; columns 3 and 4 use log population density in the 1500s, columns 5 and 6 use British legal origin, columns 7 and 8 use the fraction speaking European, columns 9 and 10 use the ability to grow crops and mine minerals, and columns 11 and 12 use all the IVs together. The sample size is slightly different from that in Table 1 because this table uses only observations for which all GDP per capita and total GDP growth rate data are available. N refers to the number of countries for which IVs, controls, and outcomes are available. Robust standard errors are in parentheses. Variable definitions and data sources are in Appendix Table A1.

## Figure A4: Correlation Between Democracy Quadratic and Economic Growth by Decade

(b) 1991-2000

(a) 1981-1990

-2.5

Mean GDP Growth Rate in 1981-1990 0'5 2'2 Mean GDP Growth Rate in 1991-2000 10.0 China 0 7.5 akistar India 0 Japan 5.0 Bangladesh India 0 United States United State Bangladesh Pakistan United Kingdom United Kingdom 2.5 2.5 Brazi Ge German ligeria Russia Japan Nigeria 0.0 0.0 0 Italy Russia 0 -2.5 -2.5 -1 Democracy Index (V-Dem) in 1980 Democracy Index (V-Dem) in 1990 (d) 2011-2019 (c) 2001-2010 Mean GDP Growth Rate in 2001-2010 Mean GDP Growth Rate in 2011-2019 10.0 10.0 China Nigeria 7.5 7.5 • Bangladesh China India India 0 5.0 Indonesi 5.0 Pakistan Indones Brazil Germany Nice United States 2.5 2.5 United France 0 France United Kingdon Japan Japan German 2 Braz 0.0 0.0 Italy Itals . -2.5 -2.5 -1 Democracy Index (V-Dem) in 2000 Democracy Index (V-Dem) in 2010 (e) 1993-2000 (f) 2001-2013 Mean Night-time Light Intensity Growth Rate in 2001-2013 Mean Night-time Light Intensity Growth Rate in 1993-2000 10.0 10.0 7.5 Brazil 7.5 China India 00 Banglad Ru India 5.0 5.0 Brazil France Pakistan Germany Indonesi Bangladest d States France 2.5 2.5 Italy Ge Ital Nigeria Pakistan Japan United States o Japan Russia United Kingdo United Kingdom 0.0 0.0 Nigeria

*Notes:* Panels (a)-(d) show the relationship between democracy and the mean GDP growth rates in four periods: 1981-1990, 1991-2000, 2001-2010, and 2011-2019. Panels (e)-(f) display the relationship between democracy and the mean nighttime light intensity growth rates in 1993-2000 and 2001-2013. The Democracy Index (V-Dem) is normalized to have mean zero and standard deviation one in each year. The size of each circle (country) is proportional to its baseline GDP. The colors depend on the level of the democracy index (warmer colors for democracy and darker colors for autocracies). The line is the fitted line from an OLS regression of the outcome against the democracy index and its square, without controls, that weights observations by baseline GDP. The shaded area corresponds to the 95% confidence interval. Variable definitions and data sources are in Appendix Table A1.

Democracy Index (V-Dem) in 1990

-2.5

-1

Ó

Democracy Index (V-Dem) in 2000

## Figure A5: Correlation Between Democracy Quadratic and Residualized Economic Growth by Decade

(a) 1981-1990

(b) 1991-2000



*Notes:* Panels (a)-(d) show the relationship between democracy and the residualized mean GDP growth rates in four periods: 1981-1990, 1991-2000, 2001-2010, and 2011-2019. Panels (e)-(f) display the relationship between democracy and the residualized mean nighttime light intensity growth rates in 1993-2000 and 2001-2013. Using a similar approach to Barro and Sala-i-Martin (2004), who residualize growth rates against covariates listed in Table 12.3 of their text, we residualize growth rates against our baseline controls: absolute latitude, mean temperature, mean precipitation, populationdensity, and median age. The Democracy Index (V-Dem) is normalized to have mean zero and standard deviation one in each year. The size of each circle (country) is proportional to its baseline GDP. The colors depend on the level of the democracy index (warmer colors for democracy and darker colors for autocracies). The line is the fitted line from an OLS regression of the outcome against the democracy index and its square, without controls, that weights observations by baseline GDP. The shaded area corresponds to the 95% confidence interval. Variable definitions and data sources are in Appendix Table A1.



Figure A6: Correlation Between Democracy Change and Outcomes

(a) Mean GDP Growth Rate in 2001-2019

(b) Mean Night-time Light Intensity Growth Rate in 2001-2013

(c) Mean Democracy Annual Change in 2001-2019



*Notes:* Panels (a) and (b) show the correlation between the mean annual change in democracy and outcomes (the mean GDP growth rate in 2001-19 and mean night-time light intensity growth rate in 2001-13). Panel (c) displays the correlation between the baseline democracy index and mean democracy index annual change in 2001-2019. Democracy change is scaled to [-10, 10]. The size of each circle (country) is proportional to its baseline GDP. The colors represent the level of democracy change, with warmer colors indicating higher levels of democracy change. The line represents the OLS regression fitted line without controls, weighted by countries' baseline GDP. The shaded area corresponds to the 95% confidence interval. Variable definitions and data sources are in Appendix Table A1.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
Panel A: Two-Stage Least Squares				_								
				Dep	endent Variab	le is Mean G	DP Growth R	ate in 2001-2	2019			
Mean Democracy Annual Change in 2001-2019	-216.8	-126.2	606.8	-340.2	182.1	119.2	35.4	91.5	-50.9	-13.5	7.2	-11.4
	(274.1)	(158.7)	(2124.0)	(668.0)	(237.7)	(374.3)	(80.1)	(89.2)	(40.7)	(16.0)	(14.6)	(8.9)
	0.43	0.43	0.78	0.61	0.44	0.75	0.66	0.31	0.21	0.40	0.62	0.20
F-Statistic (First stage)	0.9	0.8	0.1	0.2 Dependent Var	0.6 riabla is Maar	0.2 Nighttima I	0.7	0.9 Growth Pate	3.9 in 2001-201	0.8	21.8	10.5
	161.0	54.0				0	0 7				10.0	17.0
Mean Democracy Annual Change in 2001-2013	-161.3	-54.0	-382.8	167.9	-230.6	-87.5	-52.7	-130.2	-26.4	-38.5	-19.8	-17.0
	(120.8)	(25.9)	(694.4)	(324.5)	(180.2)	(43.1)	(39.6)	(53.8)	(22.4)	(21.0)	(15.4)	(12.8)
	0.18	0.04	0.58	0.60	0.20	0.04	0.18	0.02	0.24	0.07	0.20	0.19
F-Statistic (First stage)	1.9	7.1	0.3	0.4	1.6	4.1	7.4	3.7	12.5	3.8	108262.3	598.7
					endent Variab							
Democracy Annual Change in 2000	28.5	16.4	39.5	-62.3	11.6	8.8	-3.7	-18.5	7.5	4.9	2.8	-0.8
	(32.9)	(16.8)	(49.3)	(96.2)	(11.5)	(14.9)	(6.5)	(18.4)	(4.8)	(3.0)	(2.1)	(1.8)
	0.39	0.33	0.42	0.52	0.31	0.55	0.57	0.32	0.12	0.10	0.17	0.65
F-Statistic (First stage)	0.9	1.6	0.8	0.4	4.4	1.4	0.8	1.3	3.6	2.1	20.4	5.6
			I	Dependent Va	riable is Mear	n Nighttime L	ight Intensity	Growth Rate	e in 2001-201	3		
Democracy Annual Change in 2000	31.9	12.9	32.7	21.9	16.1	25.5	20.9	-58.1	8.9	7.0	4.3	1.4
	(37.1)	(10.7)	(39.1)	(36.4)	(11.7)	(27.3)	(17.2)	(71.0)	(6.0)	(5.7)	(2.6)	(1.9)
	0.39	0.23	0.40	0.55	0.17	0.35	0.22	0.41	0.14	0.22	0.10	0.47
F-Statistic (First stage)	0.9	3.2	0.8	0.5	4.5	1.0	0.9	0.4	4.6	2.2	11.7	3.7
				Depende	nt Variable is	Mean Demo	cracy Annual	Change in 20	01-2019			
Democracy Index (V-Dem, 2000)	0.01	0.03	-0.00	0.01	-0.01	-0.01	-0.01	-0.01	0.01	-0.01	-0.01	-0.01
	(0.02)	(0.05)	(0.01)	(0.02)	(0.01)	(0.04)	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)
	0.45	0.54	0.77	0.64	0.38	0.68	0.46	0.30	0.46	0.29	0.05	0.42
F-Statistic (First stage)	5.6	0.7	14.3	12.2	5.9	2.3	2.7	7.8	4.2	1.6	84.1	12.3
IVs	settler n	nortality	populatio	n density	legal	origin	lang	uage	crops &	minerals	all I	Vs
Number of IVs	1	1	1	1	1	1	2	2	10	10	15	15
Panel B: Ordinary Least Squares												
				Dep	endent Variab	le is Mean G	DP Growth R	ate in 2001-2	2019			
Mean Democracy Annual Change in 2001-2019	-3.3	-4.5	-0.10	-8.0	-0.07	-7.9	-9.2	-4.5	-8.6	-4.4	-0.1	-8.3
	(8.2)	(5.9)	(9.7)	(7.6)	(9.7)	(7.5)	(6.3)	(5.1)	(6.2)	(5.0)	(10.3)	(8.1)
			I	Dependent Va	riable is Mear	n Nighttime L	ight Intensity	Growth Rate	e in 2001-201	3		
Mean Democracy Annual Change in 2001-2013	-11.5	-10.4	-10.5	-7.0	-10.5	-7.0	-6.8	-10.9	-5.5	-9.6	-13.1	-9.8
, ,	(8.4)	(7.8)	(8.4)	(7.1)	(8.4)	(7.1)	(8.0)	(6.6)	(7.2)	(5.8)	(10.2)	(8.5)
		. ,	. ,	Dep	endent Variab	le is Mean G	DP Growth R	ate in 2001-2	2019	. ,	, ,	
Democracy Annual Change in 2000	0.2	-1.0	0.4	-1.1	0.4	-1.1	0.4	-0.6	0.4	-0.6	0.4	-1.1
	(1.3)	(1.3)	(1.2)	(1.4)	(1.2)	(1.4)	(1.0)	(1.0)	(1.0)	(1.0)	(1.2)	(1.5)
				Dependent Va								
Democracy Annual Change in 2000	0.4	-1.3	1.2	-0.6	1.1	-0.6	0.09	-1.5	0.07	-1.7	1.2	-0.4
Beinoeraey Finnaa enange in 2000	(1.3)	(1.5)	(1.1)	(1.4)	(1.1)	(1.4)	(1.2)	(1.5)	(1.2)	(1.5)	(1.1)	(1.4)
	(1.5)	(1.5)	()		nt Variable is					(1.5)	()	(1.1)
Democracy Index (V-Dem, 2000)	-0.01	-0.02	-0.02	-0.02	-0.02	-0.02	-0.007	-0.02	-0.007	-0.02	-0.02	-0.02
Democracy fluex (V-Deffi, 2000)	-0.01 (0.006)	-0.02 (0.01)	-0.02 (0.007)	-0.02 (0.01)	-0.02 (0.007)	-0.02 (0.01)	-0.007 (0.004)	-0.02 (0.008)	-0.007	-0.02 (0.008)	-0.02 (0.007)	-0.02 (0.01)
Baseline Controls	(0.006) ×	(0.01)	(0.007) ×	(0.01)	(0.007) ×	(0.01)	(0.004) ×	(0.008)	(0.004) X	(0.008)	(0.007) X	(0.01)
N	81	81	86	86	90	<b>v</b> 90	132	132	138	138	71	71
19	01	01			90	90	132	132	130	138	/1	/1

# Table A23: 2SLS Regression Estimates of Democracy Change's Effects

Notes: Panel A reports the 2SLS estimates of the effects of mean annual democracy change on the mean GDP growth rate in 2001-2019 and mean nighttime light intensity growth rate in 2001-2013. The panel also presents the 2SLS estimates of the effect of democracy on mean democracy annual change in 2001-2019. Democracy change is scaled to [-10, 10]. Democracy Index (V-Dem, 2000) is normalized to have mean zero and standard deviation one. For IVs, columns 1 and 2 use log European settler mortality, columns 3 and 4 use log population density in the 1500s, columns 5 and 6 use British legal origin, columns 7 and 8 use the fraction speaking European, columns 9 and 10 use the ability to grow crops and mine minerals, and columns 11 and 12 use all the IVs together. Robust standard errors are in parentheses. The p-values, presented under the standard errors, are displayed as 0.00 if they are strictly smaller than the 0.005 threshold. The high values of F-Statistics for the Mean Democracy Annual Change in 2001-2013 in Panel A (columns 11 and 12) are likely due to near-perfect multicollinearity among the instrumental variables. Columns 1, 3, 5, 7, 9, and 11 have no controls, while columns 2, 4, 6, 8, 10, and 12 include the following baseline controls: absolute latitude, mean temperature, mean precipitation, population density, and median age. N refers to the number of countries for which IVs, controls, and outcomes are available. Panel B reports the OLS estimates. Variable definitions and data sources are in Appendix Table A1.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
Panel (a): 1960-2020												
Democracy	1.347	1.048	1.159	1.219	1.272	1.102	1.123	0.822	-0.025	0.405	0.576	1.050
Democracy	(0.356)	(0.314)	(0.289)	(0.306)	(0.504)	(0.462)	(0.415)	(0.414)	(1.432)	(1.317)	(1.336)	(1.648)
Log GDP Lag Controls	(0.550) 1 lag	$2 \log s$	4  lags	(0.500) 4 lags	(0.304) 1 lag	$2 \log s$	4  lags	4  lags	(1.432) 1 lag	$2 \log s$	(1.550) 14 lags	4 lags
p-value lags 5 to 8	1 lug	2 1455	+ 1ug3	[ 0.406]	i iug	2 1455	4 1455	[ 0.358]	1 lug	2 1455	14 1455	[ 0.012]
Long-run effect of democracy	45.877	30.526	28.632	25.073	25.717	19.970	18.445	11.623	128.034	-3616.231	442.944	107.518
Long-run enteet of democracy	(15.010)	(10.100)	(7.871)	(7.371)	(12.066)	(9.523)	(7.676)	(6.500)	(136.473)	(91321.937)	(960.540)	(71.352)
Effect of democracy after 25 years	24.094	20.559	22.789	22.350	18.484	16.600	16.798	11.313	26.351	37.062	47.493	45.360
Effect of democracy after 25 years	(6.560)	(6.268)	(5.789)	(5.860)	(7.872)	(7.469)	(6.688)	(6.117)	(9.913)	(11.468)	(11.046)	(13.486)
Persistence of GDP process	0.971	0.966	0.960	0.951	0.951	0.945	0.939	0.929	0.991	1.000	0.997	0.987
ensistence of GD1 process	(0.005)	(0.005)	(0.006)	(0.007)	(0.008)	(0.007)	(0.008)	(0.010)	(0.009)	(0.009)	(0.007)	(0.008)
AR2 test p-value	(0.005)	(0.005)	(0.000)	(0.007)	0.14	0.00	0.95	0.78	(0.00))	(0.00)	(0.007)	(0.000)
Observations	8499	8335	7997	7287	8267	8103	7765	7067	11041	11041	11041	10797
Countries in sample	181	181	181	180	181	181	181	177	11041	11041	11041	10777
F												
Panel (b): 1980-2020												
Democracy	1.906	1.567	1.689	1.828	1.960	1.748	1.722	1.532	-0.025	0.405	0.576	1.050
	(0.478)	(0.439)	(0.422)	(0.524)	(0.818)	(0.767)	(0.683)	(0.726)	(1.432)	(1.317)	(1.336)	(1.648)
Log GDP Lag Controls	1 lag	2 lags	4 lags	4 lags	1 lag	2 lags	4 lags	4 lags	1 lag	2 lags	4 lags	4 lags
p-value lags 5 to 8	-	-	-	[ 0.060]	-	-	-	[ 0.015]	-	-	-	[ 0.032]
Long-run effect of democracy	46.341	31.888	27.035	21.341	28.085	22.087	19.168	12.711	1.596	-18.678	-47.760	20.876
	(13.652)	(9.655)	(7.148)	(6.618)	(15.210)	(11.503)	(8.630)	(6.547)	(90.860)	(64.375)	(138.601)	(30.194)
Effect of democracy after 25 years	30.122	25.607	25.366	22.108	23.484	20.410	18.935	12.942	-0.760	14.511	20.437	14.211
	(7.835)	(7.330)	(6.425)	(6.707)	(11.502)	(10.157)	(8.390)	(6.732)	(43.478)	(46.778)	(46.424)	(20.864)
Persistence of GDP process	0.959	0.951	0.938	0.914	0.930	0.921	0.910	0.879	1.016	1.022	1.012	0.950
	(0.006)	(0.006)	(0.008)	(0.012)	(0.014)	(0.012)	(0.012)	(0.015)	(0.015)	(0.014)	(0.015)	(0.033)
AR2 test p-value	. ,				0.99	0.00	0.47	0.41	. ,			
Observations	6558	6371	5995	5237	6326	6139	5763	5017	7421	7421	7421	7175
Countries in sample	181	181	181	180	181	181	181	177				

Table A24: Effect of Democratization on (Log) GDP per Capita

				Т	able A24 – C	Continued						
Panel (c): 1995-2020												
Democracy	0.196	0.382	0.450	0.350	1.817	2.010	1.781	1.301	-0.025	0.405	0.576	0.798
Democracy	(0.694)	(0.680)	(0.638)	(0.822)	(1.356)	(1.373)	(1.295)	(1.491)	(1.432)	(1.317)	(1.336)	(1.613)
Log GDP Lag Controls	1 lag	2 lags	4 lags	4 lags	1 lag	2 lags	4 lags	4 lags	1 lag	2 lags	4 lags	4 lags
p-value lags 5 to 8	e	6	U	[ 0.399]	e	e	U	[ 0.095]	U	U	0	[ 0.022]
Long-run effect of democracy	3.569	6.357	6.112	3.471	18.814	20.435	16.316	9.414	1.596	-18.678	-47.760	15.683
-	(12.674)	(11.349)	(8.595)	(8.198)	(15.909)	(15.721)	(13.067)	(11.230)	(90.862)	(64.375)	(138.601)	(29.506)
Effect of democracy after 25 years	2.698	5.271	5.615	3.397	17.330	19.395	16.134	9.524	-0.760	14.511	20.437	10.622
	(9.572)	(9.395)	(7.914)	(7.996)	(14.201)	(14.610)	(12.874)	(11.384)	(43.480)	(46.778)	(46.424)	(20.422)
Persistence of GDP process	0.945	0.940	0.926	0.899	0.903	0.902	0.891	0.862	1.016	1.022	1.012	0.949
-	(0.010)	(0.010)	(0.010)	(0.017)	(0.020)	(0.018)	(0.017)	(0.020)	(0.015)	(0.014)	(0.015)	(0.033)
AR2 test p-value					0.35	0.00	0.33	0.93				
Observations	4309	4118	3736	2974	4077	3886	3504	2754	4706	4706	4706	4550
Countries in sample	181	181	181	180	181	181	181	177				
Panel (d): 2001-2020												
Democracy	0.197	0.210	-0.241	0.337	1.120	1.577	0.240	-0.717	0.140	1.150	0.528	0.913
	(0.785)	(0.807)	(0.753)	(0.923)	(1.657)	(1.587)	(1.682)	(1.613)	(1.421)	(1.289)	(1.315)	(1.603)
Log GDP Lag Controls	1 lag	2 lags	4 lags	4 lags	1 lag	2 lags	4 lags	4 lags	1 lag	2 lags	4 lags	4 lags
p-value lags 5 to 8	e	e	e	[ 0.020]	e	e	e	[ 0.003]	e	C	C	[ 0.014]
Long-run effect of democracy	2.565	2.498	-2.139	1.391	9.982	12.494	1.529	-2.386	-7.612	-113.345	-41.301	16.057
2	(10.203)	(9.606)	(6.656)	(3.842)	(15.933)	(13.722)	(10.803)	(5.264)	(78.965)	(202.775)	(126.872)	(26.431)
Effect of democracy after 25 years	2.217	2.268	-2.079	1.387	9.473	12.188	1.526	-2.387	4.412	39.893	16.735	11.560
	(8.823)	(8.723)	(6.469)	(3.828)	(14.924)	(13.286)	(10.781)	(5.265)	(44.214)	(38.934)	(44.537)	(19.359)
Persistence of GDP process	0.923	0.916	0.887	0.758	0.888	0.874	0.843	0.700	1.018	1.011	1.012	0.943
-	(0.007)	(0.009)	(0.011)	(0.025)	(0.021)	(0.019)	(0.018)	(0.029)	(0.015)	(0.014)	(0.014)	(0.030)
AR2 test p-value					0.32	0.02	0.22	0.19				
Observations	3270	3079	2697	1936	3038	2847	2465	1718	3620	3620	3620	3500
Countries in sample	181	181	181	180	181	181	181	170				

*Notes:* These tables present estimates of the effect of democratization on log GDP per capita for the four periods. The reported coefficient on democracy is multiplied by 100. Cols. 1–4 present results from the within estimator. Cols. 5–8 present results from Arellano and Bond's (1991) GMM estimator. The AR2 row reports the p-value for a test of serial correlation in the residuals of the GDP series. Cols. 9–12 present results from the HHK (Hahn et al., 2001) estimator. In all specifications, we control for a full set of country and year fixed effects. Cols. 4, 8, and 12 include eight lags of GDP per capita as controls, but we report only the p-value of a test for joint significance of lags 5–8. Standard errors robust against heteroskedasticity and serial correlation at the country level are reported in parentheses. See Acemoglu et al. (2019) for the details of the method.

# Panel (a): 1960-2020

=

Panel (a): 1960-2020					(	Covariates Included	1		
	(1)	(2)	GDP in 1960 Quintiles x Year Effects (3)	Soviet Dummies (4) A. 2SLS Esti	Regional Trends (5) mates with F	Regional GDP and Trade (6) Fixed Effects	Regional Unrest GDP and Trade (7)	Spatial Lag of GDP (8)	Spatial Lags of GDP and Democracy (9)
Democracy	3.012 (0.792)	3.098 (0.803)	1.125 (0.689)	3.132 (0.943)	1.697 (0.885)	4.042 (0.917)	3.689 (0.910)	3.055 (0.770)	3.021 (1.312)
Long-run effect of democracy	(0.792) 76.913 (25.172)	(0.803) 79.243 (25.684)	35.226 (23.846)	(0.943) 80.800 (28.928)	36.788 (20.657)	85.877 (24.084)	(0.910) 78.077 (23.069)	(0.770) 77.170 (24.356)	72.586 (35.601)
Effect of democracy after 25 years	59.974 (17.252)	61.723 (17.559)	25.618 (16.538)	62.522 (20.138)	(20.057) 32.051 (17.703)	72.245 (18.073)	65.669 (17.549)	60.578 (16.759)	54.747 (24.387)
Observations Countries in sample	7963 180	7960 180	5496 148	7960 180	6309 174	7960 180	7960 180	7769 177	7582 176
Democracy	1.722	1.685	1.726	1.688	HHK Estimat	1.663	1.116	1.613	1.559
Long-run effect of democracy	(0.584) 1898.011	(0.542) 714.539	(0.597) 55.563	(0.553) 821.087	(0.469) 8.984	(0.445) 58.275	(0.382) 38.181	(0.478) 247.376	(0.453) 217.684
Effect of democracy after 25 years	(17105.111) 56.487 (17.895)	(2193.722) 54.610 (16.149)	(23.248) 37.806 (13.383)	(2962.032) 54.686 (16.583)	(9.467) 7.753 (7.987)	(25.498) 36.758 (10.360)	(17.762) 24.354 (8.483)	(328.407) 48.708 (15.428)	(243.348) 46.258 (14.001)
Observations	11041	11041	9089	11041	10675	11041	11041	10858	10858
Panel (b): 1980-2020				A. 2SLS Esti	mates with F	Fixed Effects			
Democracy	10.589 (1.879)	10.783 (1.832)	8.393 (2.699)	14.012 (3.294)	5.294 (4.935)	9.957 (1.785)	8.962 (1.719)	10.080 (1.670)	11.231 (5.380)
Long-run effect of democracy	182.369 (43.396)	186.057 (43.536)	140.635 (56.457)	239.517 (68.835)	74.107 (79.319)	147.775 (35.129)	133.987 (31.746)	165.314 (36.974)	269.017 (242.820)
Effect of democracy after 25 years	165.078 (34.313)	168.257 (34.111)	128.381 (47.621)	215.716 (55.709)	72.918 (76.764)	138.666 (29.875)	125.338 (27.226)	152.242 (30.130)	240.240 (185.726)
Observations Countries in sample	5966 180	5966 180	5034 148	5966 180 B. I	4315 174 HHK Estimat	5966 180 tes	5966 180	5860 177	5672 176
Democracy	2.052	2.050	0.633	2.262	0.425	1.018	0.949	1.997	1.470
Long-run effect of democracy	(0.856) -497.306 (1322.104)	(0.791) 2264.355 (22204.012)	(0.905) 72.719 (126.368)	(0.797) -1481.879 (9158.911)	(0.798) 4.290 (8.256)	(0.638) 18.761 (12.022)	(0.516) 20.072 (11.822)	(0.733) 399.691 (959.507)	(0.717) 161.145 (224.586)
Effect of democracy after 25 years	(1322.104) 75.544 (29.748)	(22204.012) 70.004 (24.535)	(120.308) 18.557 (26.329)	(9158.911) 80.151 (25.864)	(8.256) 4.255 (8.172)	(12.022) 16.214 (9.981)	(11.822) 16.366 (8.903)	(959.507) 62.097 (25.795)	(224.586) 41.471 (21.608)
Observations	7421	7421	7421	7421	7175	7421	7421	7298	7298

#### Panel (c): 1995-2020

Taller (c): 1995-2020						Covariates Include	d		
	(1)	(2)	GDP in 1960 Quintiles x Year Effects (3)	Soviet Dummies (4) A. 2SLS F	Regional Trends (5) Estimates wit	Regional GDP and Trade (6) th Fixed Effects	Regional Unrest GDP and Trade (7)	Spatial Lag of GDP (8)	Spatial Lags of GDP and Democracy (9)
Democracy	18.494 (11.535)	9.862 (6.857)	-1.099 (11.964)	9.862 (6.857)	-2.345 (13.126)	3.421 (5.657)	0.907 (6.520)	10.825 (6.891)	6.161 (6.715)
Long-run effect of democracy	(11.333) 262.797 (184.018)	(0.837) 136.101 (101.868)	(11.904) -13.201 (142.086)	(0.837) 136.101 (101.868)	(13.120) -15.725 (86.867)	(5.657) 39.196 (66.865)	(0.320) 10.121 (73.385)	(0.891) 141.110 (96.650)	(8.471) (80.073)
Effect of democracy after 25 years	(184.018) 240.109 (162.783)	(101.808) 125.009 (91.538)	(142.080) -12.504 (134.822)	(101.808) 125.009 (91.538)	(80.807) -15.721 (86.848)	(00.803) 37.440 (63.537)	9.713 (70.330)	(90.030) 132.040 (88.935)	65.710 (75.906)
Observations Countries in sample	3714 180	3714 180	3132 148	3714 180	2063 174 B. HHK Esti	3714 180	3714 180	3639 177	3451 176
Democracy	0.296 (1.594)	0.179 (1.443)	-0.179 (1.308)	0.241 (1.439)	-0.032 (1.965)	-2.072 (1.136)	-0.949 (0.886)	-0.224 (1.509)	-0.011 (1.284)
Long-run effect of democracy	(1.5)+) -15.499 (86.364)	(1.445) -18.478 (153.651)	60.061 (497.578)	(1.437) -25.402 (158.773)	(1.903) -0.157 (9.772)	(1.130) -41.487 (31.882)	-21.645 (24.729)	(1.50) 10.700 (73.982)	0.396 (44.440)
Effect of democracy after 25 years	14.462 (77.318)	(155.051) 7.013 (56.525)	(497.578) -6.024 (44.291)	9.416 (56.074)	(9.772) (-0.157) (9.752)	-33.478 (21.606)	(24.72)) -16.455 (16.757)	-9.559 (64.033)	-0.555 (62.194)
Observations	4706	4706	4706	4706	4550	4706	4706	4628	4628
Panel (d): 2001-2020				A. 2SLS I	Estimates wit	h Fixed Effects			
Democracy	16.419 (10.892)	4.468 (6.555)	-4.187 (11.708)	4.468 (6.555)	-18.309 (22.670)	3.773 (6.924)	4.438 (6.934)	3.012 (5.658)	0.790 (6.931)
Long-run effect of democracy	151.853 (108.065)	39.663 (59.414)	-34.289 (92.147)	39.663 (59.414)	-71.669 (85.042)	31.484 (59.391)	37.022 (59.879)	26.263 (50.152)	5.467 (47.954)
Effect of democracy after 25 years	147.009 (103.574)	38.576 (57.644)	-33.654 (90.765)	38.576 (57.644)	-71.653 (85.032)	30.833 (58.018)	36.264 (58.468)	25.644 (48.908)	5.424 (47.567)
Observations Countries in sample	2681 180	2681 180	2255 148	2681 180	1030 174 B. HHK Esti	2681 180 mates	2681 180	2612 177	2424 176
Democracy	0.254 (2.198)	1.045 (1.689)	-1.032 (1.549)	1.113 (1.702)	-0.302 (3.047)	-2.794 (1.504)	-1.876 (1.286)	0.900 (1.940)	-0.609 (1.550)
Long-run effect of democracy	(2.198) -26.698 (254.597)	(1.089) -108.637 (248.661)	168.169 (641.086)	-82.885 (157.804)	(3.047) -1.516 (15.325)	-248.555 (673.599)	(1.280) -62.557 (70.136)	(1.940) -32.777 (67.863)	30.511 (93.297)
Effect of democracy after 25 years	(234.397) 9.523 (81.435)	(248.001) 38.217 (63.510)	-31.762 (48.189)	43.179 (68.032)	(15.325) -1.510 (15.263)	(073.399) -74.759 (50.397)	(70.130) -37.516 (28.385)	(07.803) 39.697 (90.685)	(53.257) -23.744 (58.016)
Observations	3620	3620	3620	3620	3500	3620	3620	3560	3560

Notes: These tables present IV estimates of the effect of democratization on log GDP per capita for the four periods. The reported coefficient of democracy is multiplied by 100. Within each period, Panel A presents 2SLS estimates instrumenting democracy with up to four lags of regional democracy waves and the p-value of a Hansen overidentification test. Panel B presents the corresponding first-stage estimates and the excluded (excl.)-instruments F statistic. Panel C presents results using the HHK (Hahn et al. 2001) estimator instrumenting democracy with up to four lags of regional democracy with up to four lags of regional democracy waves (except for col. 1, where we use only one lag). In all specifications we control for a full set of country and year fixed effects and four lags of GDP per capita. In addition, we control for the covariates specified in each column label and described in the text. Standard errors robust against heteroskedasticity and serial correlation at the country level are in parentheses. See Acemoglu et al. (2019) for the details of the method.



Figure A7: Dynamic Panel Model Estimates of the Effects of Democratization on the Log of GDP Per Capita

*Notes:* These figures plot the estimated change in the log of GDP per capita caused by a permanent transition to democracy. The effects are obtained by forward iteration of the estimated process for GDP. A 95 percent confidence interval obtained with the delta method is presented in dotted lines. Time (in years) relative to the year of democratization runs on the horizontal axis. See Acemoglu et al. (2019) for the details of the method.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
Panel A: Severity		Dep	pendent Va	riable is Cor	itainment l	Health Ind	ex at 10th	Covid-19	Case (unit:	std. deviat	ion)	
Democracy Index (V-Dem, 2019)	-0.4 (0.07)	-0.3 (0.07)	-0.4 (0.05)	-0.4 (0.03)	-0.4 (0.06)	-0.3 (0.04)	-0.4 (0.08)	-0.3 (0.05)	-0.3 (0.06)	-0.4 (0.04)	-0.4 (0.04)	-0.4 (0.02)
Panel B: Coverage		De	ependent Va	ariable is Co	verage of	Containme	ent Measur	es at 10th	Covid-19	Case (unit:	%)	
Democracy Index (V-Dem, 2019)	-8.1 (2.3)	-7.1 (1.9)	-10.2 (1.5)	-9.2 (0.5)	-9.4 (1.7)	-7.8 (1.3)	-9.3 (2.4)	-8.8 (1.1)	-8.2 (1.3)	-9.0 (0.7)	-9.7 (1.2)	-9.5 (0.5)
Panel C: Speed		Depend	ent Variabl	e is Days Be	etween 10t	h Covid-1	9 Case and	Any Con	tainment N	leasure (un	it: days)	
Democracy Index (V-Dem, 2019)	0.7 (3.5)	-5.6 (2.1)	-3.1 (2.2)	-4.3 (1.3)	-3.4 (2.7)	-4.5 (1.7)	-1.2 (2.7)	-1.7 (1.8)	-0.8 (2.3)	-5.3 (1.4)	-3.9 (1.7)	-2.8 (0.7)
IVs	settler n	nortality	populati	on density	legal	origin	lang	uage	crops &	minerals	all	IVs
Baseline Controls N	<b>x</b> 80	✓ 80	<b>×</b> 84	✓ 84	<b>X</b> 88	✓ 88	<b>x</b> 129	✓ 129	<b>x</b> 132	✓ 132	<b>×</b> 70	✓ 70

# Table A26: Potential Policy Mechanisms Behind Democracy's Effect in 2020

*Notes:* This table reports the 2SLS estimates of democracy's effect on potential policy mechanisms behind democracy's negative impact in 2020, using five different IV strategies. Panel A reports the 2SLS estimates of democracy's effect on the containment health index at the 10th confirmed case of Covid-19. It is normalized to have standard deviation one. Panel B estimates the effect on the coverage of containment measures at the 10th confirmed case of Covid-19. Panel C estimates the effect on the number of days between the 10th confirmed case of Covid-19 and the introduction of any containment measure. The Democracy Index (V-Dem) is normalized to have mean zero and standard deviation one in each year. Columns 1, 3, 5, 7, 9, and 11 have no controls, while columns 2, 4, 6, 8, 10, and 12 have the following baseline controls: absolute latitude, mean temperature, mean precipitation, population density, median age, and diabetes prevalence. For IVs, columns 1 and 2 use log European settler mortality, columns 3 and 4 use log population density in the 1500s, columns 5 and 6 use British legal origin, columns 7 and 8 use the fraction speaking English and the fraction speaking European, columns 9 and 10 use the ability to grow crops and available. Robust standard errors are in parentheses. Variable definitions and data sources are in Appendix Table A1.

# Data Sets (accessed March 2025)

- Acemoglu, D. and S. Johnson (2005). Replication Data for: Unbundling Institutions. https://economics.mit.edu/people/faculty/ daron-acemoglu/data-archive.
- Acemoglu, D., S. Johnson, and J. A. Robinson (2001). Replication Data for: The Colonial Origins of Comparative Development: An Empirical Investigation. https://economics.mit.edu/people/faculty/daron-acemoglu/data-archive.
- Acemoglu, D., S. Naidu, P. Restrepo, and J. A. Robinson (2019). Replication Data for: Democracy Does Cause Growth. https: //economics.mit.edu/people/faculty/daron-acemoglu/data-archive.
- Banks, Arthur S., W. K. A. (2024). Cross-National Time-Series Data (CNTS). https://www.cntsdata.com/.
- Blavatnik School of Government at the University of Oxford (2021). Oxford Covid-19 Government Response Tracker. https://covidtracker.bsg.ox.ac.uk.
- Center for Systemic Peace (2018). Polity5 Annual Time Series, 1946-2018. https://www.systemicpeace.org/inscrdata.html.
- Coppedge, M., J. Gerring, C. H. Knutsen, S. I. Lindberg, J. Teorell, N. Alizada, D. Altman, M. Bernhard, M. F. Agnes Cornell, L. Gastaldi, H. Gjerløw, A. Glynn, A. Hicken, G. Hindle, N. Ilchenko, J. Krusell, A. Luhrmann, S. F. Maerz, K. L. Marquardt, K. McMann, V. Mechkova, J. Medzihorsky, P. Paxton, D. Pemstein, J. Pernes, J. Von Römer, B. Seim, R. Sigman, S.-E. Skaaning, J. Staton, A. Sundström, E.-t. Tzelgov, Y.-t. Wang, T. Wig, S. Wilson, and D. Ziblatt (2021). V-Dem Dataset v11.1 Varieties of Democracy Project. https://v-dem.net/data/dataset-archive/.
- Easterly, W. and R. Levine (2003). Replication Data for: Tropics, Germs, and Crops: How Endowments Influence Economic Development. https://web.archive.org/web/20201106074603/http://faculty.haas.berkeley.edu/ross\_levine/Papers/ 2003\_JME\_Tropics\_Data.xls.

Economist Intelligence Unit (2021). Democracy Index 2020. https://www.eiu.com/n/campaigns/democracy-index-2020/.

- Feenstra, R. C., R. Inklaar, and M. P. Timmer (2015). Data for: The Next Generation of the Penn World Table. https://www.aeaweb.org/articles?id=10.1257/aer.20130954.
- Food and Agriculture Association of the United Nations (2020). FAO Global Statistical Yearbook. http://www.fao.org/faostat/en/#data.
- Fraser Institute (2021). Economic Freedom of the World: 2021 Annual Report. https://www.fraserinstitute.org/sites/default/ files/economic-freedom-of-the-world-2021-master-index-data-for-researchers-iso.xlsx.
- Freedom House (2020). Freedom in the World 2020: A Leaderless Struggle for Democracy. https://freedomhouse.org/report/ freedom-world/2020/leaderless-struggle-democracy.
- Global Carbon Budget (2024). Per capita CO2 emissions GCB. https://ourworldindata.org/grapher/ co-emissions-per-capita.
- Google Dataset Publishing Language (2021). countries.csv. https://developers.google.com/public-data/docs/canonical/ countries\_csv.
- Hall, R. E. and C. I. Jones (1999). Replication Data for: Why Do Some Countries Produce So Much More Output Per Worker Than Others? https://web.stanford.edu/~chadj/HallJones400.asc.
- Helliwell, J., R. Layard, J. D. Sachs, J.-E. De Neve, L. B. Aknin, and S. Wang (2023). World Happiness Report 2023. https: //worldhappiness.report/ed/2023/#appendices-and-data.

International Diabetes Federation (2019). IDF Diabetes Atlas 2019. https://www.diabetesatlas.org/data/en/.

LaPorta, R., F. L. de Silanes, and A. Shleifer (2008). Replication Data for: The Economic Consequences of Legal Origins. https://scholar.harvard.edu/shleifer/publications/economic-consequences-legal-origins.

Lührmann et al. (2020). V-Party Dataset v1. https://v-dem.net/data/dataset-archive/.

Martinez, L. R. (2022). Replication Data for: How Much Should We Trust the Dictator's GDP Growth Estimates? https://sites.google.com/site/lrmartineza.

Mechkova et al. (2021). Digital Society Project Dataset v3. https://digitalsocietyproject.org/data-version-3/.

The Economist (2024). Excess Deaths Model. https://ourworldindata.org/grapher/ excess-deaths-cumulative-per-100k-economist.

UNESCO Institute for Statistics (2021). Data for Sustainable Development Goals. http://uis.unesco.org.

- United Nations Department of Economic and Social Affairs, Population Division (2019). World Population Prospects 2019, Online Edition. Rev. 1. https://population.un.org/wpp/.
- United Nations, World Population Prospects (2024a). Median Age, Medium Projection United Nations WPP. https://ourworldindata.org/grapher/median-age.
- United Nations, World Population Prospects (2024b). Population, Total United Nations WPP. https://ourworldindata.org/ explorers/population-and-demography?tab=map.
- UNU-WIDER Government Revenue Dataset (2023). Tax Revenues as a Share of GDP UNU-WIDER. https://ourworldindata.org/grapher/tax-revenues-as-a-share-of-gdp-unu-wider.
- US Energy Information Administration (2023). Primary Energy Consumption per Capita. https://ourworldindata.org/grapher/ per-capita-energy-use.

World Bank (2021a). Climate Change Knowledge Portal. https://climateknowledgeportal.worldbank.org/download-data.

- World Bank (2021b). Entrepreneurship Survey and Database. https://www.doingbusiness.org/en/data/exploretopics/ entrepreneurship.
- World Bank (2021c). World Development Indicators. https://databank.worldbank.org/source/ world-development-indicators.
- World Bank (2025a). Foreign Direct Investment, Net Inflows as Share of GDP. https://ourworldindata.org/grapher/ foreign-direct-investment-net-inflows-as-share-of-gdp.

World Bank (2025b). Services, value added (current US\$). https://data.worldbank.org/indicator/NV.SRV.TOTL.CD.

- World Bureau of Metal Statistics (2019a). Copper Workbook February 2019. https://www.lseg.com/en/data-analytics/ financial-data/commodities-data/metals-data/world-bureau-of-metal-statistics-wbms.
- World Bureau of Metal Statistics (2019b). Silver Workbook February 2019. https://www.lseg.com/en/data-analytics/ financial-data/commodities-data/metals-data/world-bureau-of-metal-statistics-wbms.
- World Inequality Database (2023). World Inequality Database. https://wid.world/news-article/ new-get-wid-world-inequality-data-stata-ssc-package.